# TEX-AN 2021 Communications Technology Services Agreement

Version 2.0

**DIR Contract No DIR-TELE-CTSA-007** 

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

## **Table of Contents**

1	Introduction	1	
1.1 1.2 1.3 1.4	Provision, Performance, and Management by Successful Respondent		
1.5	Construction of Agreement	3	
	1.5.1 Severability	3	
	No Implied Authority	∠	
2	CTSA Documents		
2.1 2.2 2.3	Definition	5	
3	Term	5	
3.1 3.2	Initial Term		
4 Cus	Relationship among DIR, Successful Respondent, other TEX-AN 2021 Contract Hold tomers	-	
4.2 4.3 4.4	DIR as Successful Respondent's Customer  Direct Sales Transaction Service Customers for Successful Respondent  Customers as Authorized End Users of Successful Respondent's Services  Use of Third Parties  DIR (and Customer) Right of Use		
	<ul><li>4.5.1 Successful Respondent Cooperation</li><li>4.5.2 Notice by Successful Respondent</li></ul>		
4.6 4.7	Covenant of Cooperation		
5	Awarded Services	9	
5.1 5.2 5.3	Description Rates, Prices, and Telecommunications Fees and Surcharges Authorized Changes to Rates to DIR and Prices	9	
6	Successful Respondent Personnel Management	10	
6.1 6.2 6.3	Qualifications, Retention, and Replacement of Successful Respondent Employees	10	
	6.3.1 General		

	6.3.3	Cooperation with Other TEX-AN 2021 Contract Holders	12
	6.3.4	Cooperation with Customers	12
	6.3.5	Successful Respondent's Subcontractor's Compliance	12
6.4	Conduc	ct of and Responsibility for Successful Respondent Employees	12
6.5		sibility for Subcontractors and Third Party Providers	
6.6		y of Premises, Equipment, Data, and Personnel	
6.7		ound and/or Criminal History Investigation	
7	Notice	S	14
7.1		ry of Notice	
7.2		e of Designee	
7.3 7.4	_	ation of Problems	
7.5		ation of Defaults	
7.6		of Adverse Impact	
		•	
8		al Terms and Conditions	
8.1	_	tion of Authority	
8.2		iver of Sovereign Immunity	
8.3	Force N	Majeure	17
	8.3.1	General	1
	8.3.2	Duration and Notification	
	8.3.3	Substitute Services; Termination	
	8.3.4	Payment Obligation	
	8.3.5	Allocation of Resources	18
8.4	Compe	titive Pricing	18
8.5	Publici	ty	18
8.6	Assign	ment	19
	8.6.1	Assignment by Successful Respondent	19
	8.6.2	Assignment by DIR	
	8.6.3	Amendment	19
	8.6.4	Assumption	
	8.6.5	Impermissible Assignment	20
8.7	RFO E	rrors and/or Omissions	20
8.8	Abando	onment or Default	20
8.9	Time is	s of the Essence	20
		f Performance	
8.11	Educat	ion Department General Administrative Regulations (EDGAR)	20
9	Amend	lments, Modifications, and Procedures	20
0 1	Genera	1 Amendments and Modifications	20
9.2		ments and Modifications Resulting from Changes in Law or Contract and as Mutually Ag	
7.2		micros and modifications resulting from changes in Edw of Contract and as mutually rig	-
9.3		ed Compliance with Amendment and Modification Procedures	
9.4	•	cations Resulting from Imposition of Remedies	
9.5		ments to Exhibit B, Attachment B-1 Initiated by Either Party	
		marking	
9.7	All Oth	ner Changes	2

9.8 N	Mutual Covenant of Cooperation	22
10 11	Availability of Appropriated Funds and Statutory Authority Governing Law and Regulations	
11.1 (	Governing Law and Venue	22
	Successful Respondent Responsibility for Compliance with Laws and Regulations	
	Compliance with Immigration Laws	
	Equal Opportunity Compliance	
11.5 H	Historically Underutilized Business (HUB) Program	24
12	Performance Measurement	24
13	Audit and Financial Compliance	24
13.1 (	CTSA Record Retention and Audit	24
	Access to Records, Books, and Documents	
	Audits by Governmental Authorities	
	Audits of Services, Deliverables, and Inspections	
	Response/Compliance with Audit or Inspection Findings	26
	Audit of Telecommunications Fees and Surcharges, Rates to DIR, Prices, and Assessments	
(	Conducted Pursuant to Section 13.3 of the CTSA	27
14	Terms and Conditions of Successful Respondent Billing and DIR and Customer Payn	
	Monthly Consolidated Invoice to DIR	
	nvoices for Direct Sales Transaction to Customers	
	Limitations on the Right to Bill	
	Felecommunications Fees, Surcharges, and Taxes         No Rights of Set-off	
	Expenses	
	Dispute Handling and Adjustments	
	Billing Adjustments	
	Back-billing and POP Moves	
14.10	Liability for Taxes, Insurance, and Indemnification	31
14.11	1	
14.12		
14.13		
14.14 14.15	Payments for Services.	
14.13	1 0	
	14.15.1 DIR Administrative Fee	
1	14.15.2 Direct Sales Transaction Service Reporting	33
15	Disclosure and Confidentiality of Information	
	Confidentiality	
	Disclosure or Loss of DIR's and Customers' Confidential Information	
	Public Information	
	Unauthorized Acts	
	Legal Action	
16	Liability	
	Property Damage	
	Risk of Loss	
16.3 I	Limitation of Liability	38

17	Reme	dies and Disputes	39
		nding and Expectations	
17.2	Tailored I	Remedies	39
	17.2.1	Understanding of the Parties	39
	17.2.2	Notice and Opportunity to Cure for Deficiencies	
17.3	Terminati	on of CTSA	44
	17.3.1	Termination by Mutual Agreement of the Parties.	44
	17.3.2	Termination in the Best Interest of the State	
	17.3.3	Termination for Cause	44
17.4	DIR Righ	ts Upon Successful Respondent's Bankruptcy	40
	17.4.1	General Rights	40
	17.4.2	DIR Rights in Event of Bankruptcy Rejection	
17.5	Effective	Date of Termination	4′
		of Termination Effective Date	
	•	and Other Provisions at CTSA Termination	
		ent of CTSA in the Event of Remedies	
		rlement Assistance	
17.10		ssful Respondent Responsibility for Associated Costs	
17.1	•	e Resolution	
	17.11.1	General Agreement of the Parties	
	17.11.2	Duty to Negotiate in Good Faith	
	17.11.3	Claims for Breach of CTSA	
17.12		ty of Successful Respondent	
18	Assura	ances and Certifications	50
18.1	Conflicts	of Interest	50
	18.1.1	Representation	50
	18.1.2	General Duty Regarding Conflicts of Interest	50
19	Repre	sentations, Warranties, and Covenants	52
19.1	Authoriza	ıtion	5
19.2	Ability to	Perform	5
		l Respondent Certifications	
		ship and Performance	
	_	ce with Agreement	
		ent and Misappropriationional Conflicts of Interest	
	Ü		
	19.7.1	Definition	
	19.7.2 19.7.3	Warranty	
	19.7.3 19.7.4	Continuing Duty to Disclose	
	19.7.4	Flow Down Obligation	
		onnel Recruitment Prohibition	
		back Provisionback Provision	

19.10	Debt or Back Taxes Owed to the State	55
19.11	Outstanding Debts and Judgments	55
19.12	Antitrust	50
19.13	Agency Executive	5
19.14	Deceptive Trade Practices; Unfair Business Practices	5
19.15		
19.16		
19.17	Preservation of Records	5
20	Intellectual Property	57
20.1 I	nfringement Indemnification	5
	Exceptions	
20.3 F	Rights in Data	58
20.4 (	Ownership of Pre-existing Materials	58
20.5 7	Third party Commercial Software	59
20.6 F	Pre-existing Materials for Custom Software Deliverables Related to Services	59
20.7 (	Ownership of Work Product by DIR or Customer	59
20.8 I	License to Intellectual Property	59
21	Transition Plan for Expiration of the CTSA	60
21.1 S	Scope	60
	General Statement for Transition	
	Agreement for Transition Assistance Services	
	Fransition Assistance Services	
21.5 (	Covenant of Cooperation	64
	Other Successful Respondent Agreements with Customers	
22	Miscellaneous	65
22.1 E	Entire Agreement	6:
	Counterparts	
	Authority to Execute	
	oit A. Definitions	
	ntroductionDefinitions and Acronyms	······· 4
Exhil	oit B. Descriptions, Rates to DIR, Prices, and Related Telecommunications Fees and	******* 4
	narges for Awarded Services	
	ntroduction	
	Contractual Principles	
3 N	Methodology for Updating Associated Attachments	
3.1 F	Recommendations	2
3.2 F	Review and Input	2
3.3 A	Approval	3
3.4 A	Appeal	3
<b>4</b> A	Attachments to Exhibit B	
	bit C. Performance Management	
	ntroduction	
	Performance Management Contractual Principles	
2.1 (	Credit Assessments	,
	~~ ~ ~~ ~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~ ~~	4

Vendor Contract No.\_\_\_\_\_

	Service Level Agreements (SLAs)	
	Service Instance Performance	
	Chronic Outage Reports	
3	Methodology for Updating Associated Attachment C-1	
3.1	Recommendations	
3.2	Review and Input	
3.3	Approval	
3.4	Appeal	4
4	Documents Referenced in Exhibit C Performance Management	4
4.1	Attachments to Exhibit C	4
Exh	ibit D. Plans	2
1	Introduction	
2	Requirements	
3	Methodology for Updating Associated Plans	2
3.1	Recommendations	
3.2	Review and Input	
3.3 3.4	Approval	
	••	
4	Documents Referenced in Exhibit D Plans	
4.1	Attachments to Exhibit D.	
4.2	Exhibit D Plans	3
	4.2.1 Attachments D-1 through D-7 to be determined by the Parties and posted to the contract landing page on the DIR website	3
Exh	ibit E. Insurance	2
1	Introduction	
2	Insurance Contractual Principles	2
3	Methodology for Updating Associated Exhibit E Attachments and Plans	
4	Documents Referenced in Exhibit E Insurance	3
4.1		
4.2	Attachment E-1. Successful Respondent's Certificate of Insurance	3
	tibit F. Form of TEX-AN 2021 Customer Services Agreement	
1	Introduction	
<b>2 3</b>	Contractual Principles  Methodology for Updating Associated Exhibit Attachment	
	2 0	
3.1	Recommendations	
4	Documents Referenced in Exhibit F Form of TEX-AN 2021 Customer Services Agreement	2

Vendor Contract No.\_\_\_\_\_

## Table of Exhibits

Exhibit A.	Definitions	<b>A.1</b>
ExhibitB.	Descriptions, Rates to DIR, Prices, and Related Telecommunications	
	Fees and Surcharges for Awarded Services	<b>B.1</b>
Exhibit C.	Performance Management	<b>C.1</b>
Exhibit D.	Plans	<b>D.1</b>
Exhibit E.	Insurance	<b>E.1</b>
Exhibit F.	Form of TEX-AN 2021 Customer Services Agreement	<b>F.1</b>

Table of Contents vii

Vendor	Contract No.	

1

## TEX-AN 2021 Communications Technology Services Agreement

This TEX-AN 2021 Communications Technology Services Agreement (CTSA) is entered into by and between the State of Texas, acting by and through the Texas Department of Information Resources (DIR), with its principal place of business at 300 West 15th Street, Suite 1300, Austin, Texas 78701, and Hughes Network Systems, LLC (Successful Respondent), with a principal place of business at 11717 Exploration Lane, Germantown, Maryland 20876.

#### **Preamble**

WHEREAS, in accordance with the laws of the State, DIR issued the Request for Offer (RFO) for TEX-AN Communications 2021 (TEX-AN 2021) on the Texas Comptroller of Public Accounts' (CPAs) Electronic State Business Daily (ESBD) Website, Request for Offer (RFO) number DIR–CPO-TMP-552 (the RFO);

WHEREAS, in response to the RFO, Successful Respondent submitted Successful Respondent's Response, dated March 23, 2021, as revised, amended and supplemented thereafter (the Response)

WHEREAS, based on the Response, DIR and Successful Respondent have engaged in extensive negotiations, discussions, and due diligence that have culminated in the formation of the contractual relationship described in this TEX-AN 2021 Communications Technology Services Agreement (hereinafter CTSA or Contract); and

NOW THEREFORE, in consideration of the mutual promises and covenants contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, DIR and Successful Respondent (collectively, the "Parties" and each, a "Party") hereby agree as follows:

#### 1 Introduction

#### 1.1 Provision, Performance, and Management by Successful Respondent

DIR desires that certain communications technology Services procured in accordance with State law (pursuant to Chapter 2170, Texas Government Code) be provided, performed, and managed by Successful Respondent as and to the extent described in this CTSA. Successful Respondent has carefully reviewed DIR's requirements, has performed all due diligence it deems necessary, has received all information requested and required of DIR, and desires to deploy, perform, and manage such communications technology Services for DIR, the Customers, and, as required, other TEX-AN NG and TEX-AN 2021 Contract Holders. If Successful Respondent later discovers such due diligence was insufficient or lacking certain information, Successful Respondent shall assume any associated costs or expenses.

## 1.2 Successful Respondent's Experience and Qualifications

The Successful Respondent represents and warrants that it is an established provider of communications technology Services as awarded under this CTSA and has the skills, qualifications, expertise, financial

resources, and experience necessary to provide the communications technology Services, plans, reports, and other deliverables described in this CTSA.

#### 1.3 Definitions

- a) The terms defined in this Agreement include the plural as well as the singular and the derivatives of such terms. Unless otherwise expressly stated, the words "herein," "hereof," and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular article, section, subsection, exhibit, attachment, or other subdivision. Article, section, subsection, exhibit, and attachment references refer to articles, sections and subsections of, and exhibits and attachments to, this Agreement. The words "include" and "including" shall not be construed as terms of limitation. Unless otherwise modified, the words "day," "month," and "year" mean, respectively, calendar day, calendar month, and calendar year. The word "notice" and "notification" and their derivatives mean notice or notification in writing. References in this Agreement to any law shall be to such law in changed or amended form or to a newly adopted law replacing a prior law. All references to this Agreement shall include the exhibits and attachments to this Agreement unless otherwise provided. Other terms used in this Agreement are defined in the context in which they are used and have the meanings there indicated.
- b) Except as otherwise expressly provided in this CTSA, all capitalized terms (including derivatives of such terms) used in this CTSA shall have the meanings ascribed to them in Exhibit A Definitions; provided however, that any such defined terms, if not capitalized herein, shall have the same meaning as the defined term unless the context or industry usage require a different meaning. Words having well-known technical or trade meanings shall be accorded such meaning, unless expressly defined otherwise herein. Uncapitalized terms or phrases are to be given their usual meaning.
- c) Global drafting conventions.
  - 1) The terms "include," "includes," and "including" are terms of inclusion, and where used in this CTSA, are deemed to be followed by the words "without limitation."
  - 2) Any references to "sections," "exhibits," or "attachments" are deemed to be references to sections, exhibits, or attachments to this CTSA.
  - 3) Any references to agreements, contracts, statutes, or administrative rules or regulations in this CTSA are deemed references to these documents as amended, modified, or supplemented from time to time during the term of this CTSA.

#### 1.4 Inducements

- a) In executing this CTSA, DIR relies on Successful Respondent's representations, warranties, and covenants regarding the following:
  - 1) Successful Respondent, including its agents and Subcontractors, regularly provide the types of Services described in the RFO to other public or private entities;
  - 2) Successful Respondent, including its agents and Subcontractors, have the skills, qualifications, expertise, financial resources, and experience necessary to perform the Services described in the RFO, Successful Respondent's Response, and this CTSA in an efficient and cost-effective manner with a high degree of quality and responsiveness, and have performed similar Services for other public or private entities;

Vendor Contract No.	
---------------------	--

- 3) Successful Respondent has thoroughly reviewed, analyzed, and understood the RFO, has timely raised all questions or objections to the RFO, and has had the opportunity to review and fully understand the DIR's current TEX-AN NG program, operating environment for the Services, this CTSA, and the needs and requirements of DIR, the Customers, and the State during the Term of the CTSA;
- 4) Successful Respondent has had the opportunity to review and understand the State's stated objectives in entering into this CTSA and, based on such review and understanding, Successful Respondent has the ability and capability to perform the Services in accordance with the terms and conditions of this CTSA;
- Successful Respondent also has reviewed and understands all of the risks associated with the TEX-AN 2021 Program as described in the RFO and the CTSA, including the risk of nonappropriation of funds;
- 6) Successful Respondent shall at all times be capable of and legally authorized to provide the Services; and
- 7) The Rates to DIR, Prices, and any discounts offered hereunder shall be true and correct.
- b) Accordingly, on the basis of these representations, warranties, and covenants, DIR desires to engage Successful Respondent to perform the Services under the terms and conditions set forth in this CTSA, and Successful Respondent covenants to perform the Services under the terms and conditions set forth in this CTSA.

## 1.5 Construction of Agreement

#### 1.5.1 Severability

If any provision of this CTSA (or any portion thereof) or the application of any such provision (or portion thereof) to any person, entity, or circumstance is held to be illegal, invalid, or otherwise unenforceable in any respect by a final judgment, order of the State Office of Administrative Hearings, or court of competent jurisdiction, such provision shall be deemed to be void and unenforceable. Notwithstanding the preceding sentence, the remaining provisions of this CTSA, if capable of substantial performance, shall remain in full force and effect. By entering into this CTSA, DIR makes no representations or warranties regarding the enforceability of the terms of this CTSA and DIR does not waive any applicable law that conflicts with the terms of this CTSA.

#### 1.5.2 Survival of Terms

Notwithstanding the expiration or and termination of this CTSA:

- a) any provision of this CTSA that contemplates performance or observance subsequent to any termination or expiration of this CTSA shall survive any termination or expiration of this CTSA and continue in full force and effect:
- b) all provisions of this Agreement shall survive the expiration or termination of this Agreement to the fullest extent necessary to give the Parties the full benefit of the bargain expressed herein
- c) any provisions that the Parties have expressly agreed will survive any termination or expiration shall survive the expiration or any termination of this Agreement;
- d) any terms that remain to be performed or by their nature would be intended to be applicable
  following any such termination or expiration, including the Administrative Fee, any and all
  payment obligations invoiced prior to the termination or expiration hereof, and indemnification
  rights and obligations;

- e) the limitation on use and disclosure of DIR or Customer Data shall survive the expiration or any termination of this Agreement and shall be perpetual;
- f) each party's confidentiality obligations under this CTSA shall continue for any period required by applicable law or until such time as the Confidential Information is publicly known and made generally available through no action or inaction of the receiving Party;
- g) all applicable services, warranties, or service agreements that were entered into between Successful Respondent and a Customer under the terms and conditions of the CTSA shall survive the expiration or termination of the CTSA; and
- h) all Purchase Orders issued and accepted by Successful Respondent shall survive expiration or termination of the CTSA for the term of the Purchase Order unless the Customer terminates the Purchase Order sooner. However, regardless of the term of the Purchase Order, no Purchase Order shall survive the expiration or termination of the CTSA for more than two (2) years, unless Customer makes an express finding and justification for the longer term. The finding and justification must either be included in the Purchase Order, or referenced in it prior to the expiration or termination of the CTSA and maintained in Customer's procurement record.
- i) The Successful Respondent agrees that DIR may require continued performance, not including termination assistance, beyond the initial or any renewal CTSA term, of any of the within described Services at the rates specified in the CTSA. This option may be exercised more than once, but the total extension of performance hereunder shall not exceed four (4) calendar months. Such extension of Services shall be subject to the requirements of the CTSA, with the sole and limited exception that the original date of termination shall be extended pursuant to this provision.

### 1.5.3 Headings

The article and section headings and the Table of Contents in this CTSA are for reference and convenience only and may not be considered in the interpretation of this CTSA.

#### **1.6** No Implied Authority

- a) The authority delegated to Successful Respondent by DIR is limited to the terms of this CTSA. DIR is the State agency designated by the Texas Legislature to administer the TEX-AN 2021 Program, and no other agency of the State grants Successful Respondent any authority related to this TEX-AN 2021 Program. Successful Respondent may not rely upon implied authority, and specifically is not delegated authority under this CTSA to:
  - 1) make public policy;
  - 2) promulgate, amend, or disregard administrative regulations or program policy decisions made by DIR for administration of the TEX-AN 2021 Program; or
  - 3) unilaterally communicate or negotiate with any Customer or the Texas Legislature on behalf of DIR regarding the TEX-AN 2021 Program.
- b) Successful Respondent may not incur any debts, obligations, expenses, or liabilities of any kind on behalf of the State or DIR.

## 1.7 Legal Authority

a) DIR is authorized to enter into this CTSA under Chapter 2170, Texas Government Code. Successful Respondent is authorized to enter into this CTSA pursuant to the authorization of its governing board or controlling owner or officer.

Vendor Contract No.	
---------------------	--

b) The person or persons signing and executing this CTSA on behalf of the Parties, or representing themselves as signing and executing CTSA on behalf of the Parties, warrant and guarantee that he, she, or they have been duly authorized to execute this CTSA and to validly and legally bind the Parties to all of its terms, performances, and provisions.

#### 2 CTSA DOCUMENTS

#### 2.1 Definition

This CTSA includes each of the exhibits and attachments attached to this CTSA, which are hereby incorporated into and deemed part of this CTSA, and unless otherwise expressly stated all references to this CTSA shall include the exhibits and attachments. A listing of the exhibits and attachments is included in the Table of Contents.

## **2.2** Compliance with Procurement Laws

This CTSA is the result of compliance with applicable procurement laws of the State. DIR issued a solicitation on the CPA's ESBD, RFO DIR—CPO-TMP-552, on January 4, 2021, for TEX-AN 2021 Communications Technology Services. Successful Respondent responded, and DIR conducted its evaluation. DIR determined that Successful Respondent should be invited into negotiations. DIR and Successful Respondent have engaged in extensive negotiations, discussions, and due diligence that have culminated in the formation of the contractual relationship described in this CTSA. Upon execution of all CTSAs arising from the RFO, a notice of award for RFO DIR-CPO-TMP-552 shall be posted by DIR on the ESBD.

#### 2.3 Order of Precedence

In the event of any conflict or contradiction between or among the various documents compromising the CTSA, the documents will control in the following order of precedence:

- a) Articles 1 through 22 of this CTSA, and all amendments thereto,
- b) Exhibit A Definitions to the CTSA, and all amendments thereto.
- c) All other Exhibits to the CTSA, and all amendments thereto,
- d) Attachments to the CTSA, and all amendments thereto,
- e) Appendices to the CTSA, and all amendments thereto;
- f) The RFO Number DIR-CPO-TMP-552, as amended and clarified by DIR official revisions or addenda, and incorporated by reference herein.
- g) Successful Respondent's final Response dated September 2, 2021, and incorporated by reference herein.

#### 3 TERM

#### 3.1 Initial Term

The Initial Term of this CTSA shall commence on the Effective Date, date of last signature, and continue for four (4) years from the Effective Date, unless this CTSA is terminated earlier as provided herein, in

which case the Initial Term of this CTSA shall end at 11:59:59 p.m., Central Time, on the effective date of such termination (the "Initial Term").

## 3.2 Extension(s)

By giving written notice to Successful Respondent no less than thirty (30) days prior to the expiration date of the Initial Term or the then-current Renewal Term, DIR, in its sole and absolute discretion, shall have the right to extend the Term of the CTSA for up to four (4) years from the expiration of the Initial Term, in any combination of months or years as necessary to complete the purpose of this procurement. DIR, in its sole and absolute discretion, may extend the CTSA more than once up to a total of an additional four-year period (each a "Renewal Term"). The exact period(s) of the extension(s) shall be specified in the DIR notice of extension. The total period of time during which the CTSA is in effect is the Term. No termination charges or penalties shall be applicable to any termination before or after the end of the Initial Term.

## 4 RELATIONSHIP AMONG DIR, SUCCESSFUL RESPONDENT, OTHER TEX-AN 2021 CONTRACT HOLDERS, AND CUSTOMERS

## 4.1 DIR as Successful Respondent's Customer

Pursuant to its authority found in Chapter 2170, Texas Government Code, DIR is the single State agency mandated to procure telecommunications services on behalf of Customers. In all interactions under the CTSA, DIR is in the role of Successful Respondent's customer, with the exception of Local Services, Small Office/Home Office (SO/HO) Internet Connectivity Services, wireless voice and Data Services, Technology Based Conferencing Services, and telecommunications managed Services (hereinafter collectively "Direct Sales Transaction Services"). Except as noted below, DIR will issue all Purchase Orders to Successful Respondent on behalf of itself and the Customers. Except as noted below, DIR shall act as the Successful Respondent's billing agent for all Services ordered and consumed by the Customers and shall pay net proceeds from such billings to Successful Respondent.

## 4.2 Direct Sales Transaction Service Customers for Successful Respondent

- a) If Successful Respondent is awarded any Direct Sales Transaction Services, then Successful Respondent shall receive Purchase Orders for such Direct Sales Transaction Services from the Direct Sales Transaction Service Customer(s). Successful Respondent shall bill Direct Sales Transaction Service Customers for the Services at the Prices allowed by this CTSA (or Statement of Work (SOW) as applicable) that includes the Administrative Fee. Successful Respondent shall remit the associated Administrative Fee to DIR on a monthly basis. Successful Respondent shall also be responsible for sales reports and other administrative duties associated with providing these Services to the Direct Sales Transaction Service Customers.
- b) Successful Respondent agrees to process all Direct Sales Transaction Service Customer orders for Services available under the CTSA through the CTSA. Successful Respondent shall not offer Direct Sales Transaction Services to Customers outside of the CTSA or for rates higher than negotiated herein. Successful Respondent agrees that all discounts offered to DIR are equal to or exceed any discount offered by Successful Respondent to any similarly situated Customer.

Vendor	Contract No.	
vendor	COMPACE INC.	

## 4.3 Customers as Authorized End Users of Successful Respondent's Services

As specified in Section 2170.004, Texas Government Code, certain Texas governmental and nongovernmental entities are authorized to use the telecommunications services for which DIR contracts in this CTSA. For DIR's purposes, these entities are Customers. From the Successful Respondent's perspective, these entities may be referred to as "Authorized End Users," whether as organizations or persons employed by such organizations. All references herein to "Customer" include "Authorized End Users." When ordering certain Services from Successful Respondent, non-state agency Customers may enter into a certain TEX-AN 2021 Customer Services Agreement (CSA), with DIR, which sets forth the terms and conditions for the deployment, acceptance, and delivery of such certain Services by Successful Respondent to such Customer. Under each CSA, DIR shall be the billing agent on behalf of Successful Respondent to receive payment for the Services from the Customer. The minimally acceptable terms and conditions of the CSA are attached as Exhibit F. Customers operating under the CSA will ensure funds are committed for the payment of services to DIR where DIR is acting as Successful Respondent's billing agent. In addition, Successful Respondent understands and agrees that if Successful Respondent responds to certain Customer pricing requests or Statements of Work, then, in order to contract with the Customer, Successful Respondent may be required to comply with additional terms and conditions or certifications that an individual Customer may require due to state and federal law (e.g., privacy and security requirements).

#### 4.4 Use of Third Parties

An award from the RFO is not exclusive to the Successful Respondent. Successful Respondent acknowledges that DIR intends to award or has awarded multiple CTSAs for Services out of the RFO to other Successful Respondents (TEX-AN 2021 Contract Holders). Successful Respondent further acknowledges that some of the Services provided by Successful Respondent hereunder may also be provided by other TEX-AN 2021 Contract Holders. Finally, Successful Respondent acknowledges that Successful Respondent may have a dependency on other TEX-AN 2021 Contract Holders to serve a particular Customer in a particular area. Therefore, Successful Respondent warrants and covenants, for itself and its Subcontractors, it shall not discriminate against another TEX-AN 2021 Contract Holder in the deployment and delivery of any Service in any manner to DIR and/or any Customer, or otherwise cause any disruption to Services as deployed and delivered to DIR and/or Customers by any other TEX-AN 2021 Contract Holder.

## 4.5 DIR (and Customer) Right of Use

- a) Notwithstanding anything to the contrary contained in this CTSA, Successful Respondent acknowledges and agrees that: (i) this is not a requirements CTSA and DIR and Customers shall not be required to obtain their requirements for any of the Services from Successful Respondent; and (ii) Successful Respondent is not the exclusive provider to DIR or the Customers of any of the Services and DIR and the Customers may at any time themselves and/or through third parties (each, a "DIR Contractor") provide and/or obtain any services (including services to supplement, replace, or render unnecessary the Services).
- b) Nothing in this Agreement shall be construed or interpreted as limiting DIR's right or ability to add or delete Customers, or DIR's or any Customer's right or ability during the Term to change requirements, move parts of Services in and out of scope, or to increase or decrease their demand for Services. To the extent DIR or a Customer obtains from Successful Respondent, or provides to itself, any of the Services, the charges, if any, shall be adjusted downward in accordance with **Exhibit B Pricing and Volumes**, to the extent applicable, or equitably adjusted downward in

Vendor Contract No.	
---------------------	--

proportion to the portion of the Services that Successful Respondent shall not be providing. Both Successful Respondent and Customer shall approve such adjustments in writing.

c) DIR may elect to solicit and receive bids from third parties to perform any new services. If DIR elects to use third parties to perform new services: (i) such new services shall not be deemed "Services" under the provisions of this Agreement; and (ii) Successful Respondent shall cooperate with such third parties as provided above.

#### 4.5.1 Successful Respondent Cooperation

Successful Respondent acknowledges that its provision of the Services shall require significant cooperation with third parties, and Successful Respondent shall fully cooperate and work in good faith with third parties as described in this CTSA and to the extent otherwise requested by DIR.

## 4.5.2 Notice by Successful Respondent

Without limiting its obligations under this CTSA, Successful Respondent shall expeditiously notify DIR when it becomes aware that an act or omission of DIR or Customer personnel or a DIR Contractor shall cause, or has caused, a problem or delay in providing the Services, and shall work with DIR, the Customers, and the DIR Contractor to prevent or circumvent such problem or delay. Successful Respondent shall cooperate with DIR, the Customers, and DIR Contractors to resolve differences and conflicts arising between the Services and other activities undertaken by DIR, the Customers and DIR Contractors.

## 4.6 Covenant of Cooperation

Successful Respondent hereby agrees and covenants that it shall fully cooperate with other TEX-AN 2021 Contract Holders in fulfilling the terms of this CTSA and/or CSA with DIR and/or Customers. Successful Respondent is required to cooperate to the fullest extent possible to assist DIR in communications and negotiations with all Customers as directed by DIR. Successful Respondent acknowledges that this covenant is material to DIR and that failure to maintain such cooperation may be grounds for termination for cause. Each Customer shall cooperate with its selected Contract Holders and DIR in fulfilling the terms of each CSA to which they are a party.

## **4.7** Special Covenant to Cooperate with DIR on Internal Business Process Improvements

All TEX-AN 2021 Contract Holders shall interface with DIR's internal service delivery system to support DIR's service delivery processes, including the BMC Helix ITSM platform version 20.08. Network and any subsequent versions to network demarcation points are determined on a case by case basis depending upon type of Service. Successful Respondent shall provide timely, complete, and commercially reasonable cooperation in the implementation and use of the DIR adopted systems at no additional cost to DIR or the Customers. As of the Effective Date, the BMC Helix ITSM platform includes service request management (Service ordering), asset management, performance and service level management, incident management (help desk/trouble tickets), change management, and knowledge management. Other systems in development to support Tex-AN-2021 include DIR enterprise billing and customer relationship management. As of the Effective Date, DIR and Successful Respondent have agreed to the following formats and Data to include detailed data for quotes, orders, trouble tickets, billing, Service Level Agreements, and network monitoring and formats to include email, delimited flat files, and web services XML as documented in **Appendix B Master Vendor Reporting Guide**. DIR will provide reasonable written notice to Successful Respondent as new systems become available for Successful Respondent's use. If Successful Respondent is unable to implement and use the DIR adopted internal

Vendor Contract No.	
---------------------	--

systems at no additional cost to DIR or Customers, DIR reserves the right to terminate the CTSA. Successful Respondent shall promptly notify DIR if it encounters an unforeseen challenge to implementing any DIR adopted internal systems. DIR agrees to negotiate in good faith to resolve the Successful Respondent's challenge, reserving the right to terminate the CTSA if no resolution is apparent within thirty (30) days of notification of the challenge.

### 5 AWARDED SERVICES

## **5.1** Description

A description of the Services awarded to Successful Respondent under this CTSA are found in Exhibit B to the CTSA. This CTSA is an indefinite quantity contract. DIR has not made and does not make any representations or warranties about the amount or type of Services that Successful Respondent may sell to DIR or the Customers as a result of executing this CTSA.

## **5.2** Rates, Prices, and Telecommunications Fees and Surcharges

- a) The rates for the Services as set forth in Attachments to Exhibit B to the CTSA are the Rates to DIR, with the exception of the Prices for which Successful Respondent shall directly bill the Customers at the Prices set forth in Attachments to Exhibit B, or as negotiated through the SOW for Telecommunications Managed Services. For all other Services in use, Successful Respondent shall present a Monthly Consolidated Invoice to DIR based on the Rates to DIR. In its role as billing agent for Successful Respondent, DIR shall bill Customers the Rates to DIR, plus an additional percentage as the Administrative Fee, as authorized by Chapter 2170, Texas Government Code. Successful Respondent acknowledges and agrees that DIR shall retain its Administrative Fee from the amounts it collects from Customers.
- b) If the Successful Respondent is awarded Local Services, SO/HO Internet Connectivity Services, wireless Services, and/or Technology Based Conferencing Services, Attachments to Exhibit B also contains the Prices for these Services, which include the Successful Respondent's rate plus the Administrative Fee. Successful Respondent shall bill the Direct Sales Transaction Service Customers for the Price as defined in Exhibit B and shall remit the Administrative Fee to DIR on a monthly basis. Successful Respondent shall provide DIR with monthly Direct Sales Transaction Service Reports.
- c) If the Successful Respondent is awarded Telecommunication Managed Services, Successful Respondent shall work directly with Customers to negotiate an SOW that will include pricing for the Services, which will include the Successful Respondent's rate plus the Administrative Fee. Successful Respondent shall bill the Direct Sales Transaction Service Customer for the Price as defined in the SOW and shall remit the Administrative Fee to DIR on a monthly basis. Successful Respondent shall provide DIR with monthly Direct Sales Transaction Service Reports.
- d) Attachments to Exhibit B also lists for each awarded Service, all applicable telecommunications fees and/or surcharges that are required to be assessed by Successful Respondent against Customers in accordance with federal and/or state law. DIR shall include the listed telecommunications fees and surcharges on bills to Customers and Customers shall be responsible for paying such telecommunications fees and surcharges in full. If applicable to the Direct Sales Transaction Services, Successful Respondent shall include the telecommunications fees and surcharges on its bills to the Direct Sales Transaction Service Customers, and the Direct Sales Transaction Service Customers shall be responsible for paying such telecommunications fees and surcharges in full. No telecommunications fees and surcharges that are not set forth in

9

Vendor Contract No.

Attachments to Exhibit B may appear on bills to DIR or Customers. Notwithstanding the foregoing, if federal or state laws and regulations that impose additional fees and/or surcharges become effective after the Effective Date of this CTSA then such new fees and/or surcharges may be assessed by Successful Respondent against DIR Customers; however, such charges shall be implemented through the contract amendment process as outlined in this CTSA. Additionally, Successful Respondent reserves the right to update existing fees and/or surcharges as such updates are imposed by issuing government entity with notice to DIR and incorporation into the CTSA by compliance with Article 9 of this CTSA.. DIR reserves the right to review telecommunications taxes, fees, and surcharges for applicability of payments by public entities and reserves the right to exempt Customers pursuant to federal, state, and/or local exemptions.

## **5.3** Authorized Changes to Rates to DIR and Prices

Successful Respondent may propose Attachments to Exhibit B changes to lower the Rates to DIR and/or the Prices. DIR shall have reasonable time to review and determine, in its sole and absolute discretion, if the change to the rates to DIR is a best value for the State. If DIR approves the Rates to DIR or Price change, such change shall be implemented using the Electronic Administrative Update (EAU) process defined in CTSA Section 9.5 to amend the CTSA and document the new Rates to DIR and/or Price. The Parties agree to make new Rates to DIR or Prices available to Customers as quickly as possible. Successful Respondent initiated Point of Presence (POP) location changes, or any other changes that have an adverse effect (price increase) on Rates to DIR or Prices must be authorized by an amendment to the CTSA before being billed. Changes that lower pricing will be administered via the EAU process.

## 6 Successful Respondent Personnel Management

## 6.1 Qualifications, Retention, and Replacement of Successful Respondent Employees

Successful Respondent agrees to maintain the organizational and administrative capacity and capabilities to carry out all Successful Respondent duties and responsibilities, including providing and supporting the Services, under this CTSA. The personnel Successful Respondent assigns to perform the duties and responsibilities under this CTSA will be properly trained and qualified for the functions they are to perform. Notwithstanding transfer or turnover of its personnel, or of its agents' or Subcontractors' personnel, Successful Respondent remains obligated to perform all duties and responsibilities, including providing and supporting the Services, without degradation and in accordance with the terms of this CTSA.

## **6.2** Responsibility for Successful Respondent Personnel

a) Under no circumstances will Successful Respondent Personnel, be considered or deemed employees of DIR or the State, but will be considered Successful Respondent's employees, agents, or Subcontractors for all purposes. Successful Respondent, not the State, DIR, or the Customers, has the right, power, authority, and duty to supervise and direct the activities of the Successful Respondent Personnel and to compensate such Successful Respondent Personnel for any work performed by them hereunder. Successful Respondent, and not the State, DIR, or the Customers, shall be responsible and therefore solely liable for all acts and omissions of Successful Respondent Personnel associated or related in any manner with this CTSA or the related products, equipment, and Services, including acts and omissions constituting negligence, gross negligence, willful misconduct, and/or fraud.

Vendor Contract No.	
---------------------	--

- b) Except as expressly provided in this CTSA, neither Successful Respondent nor any of Successful Respondent Personnel may act in any sense as agents or representatives of DIR or the State of Texas.
- c) Successful Respondent Personnel shall be paid exclusively by Successful Respondent for all Services performed. Successful Respondent is responsible for and must comply with all requirements and obligations related to such Successful Respondent Personnel under local, State, or federal law, including minimum wage, social security, unemployment insurance, State and federal income tax, and workers' compensation obligations.
- d) Successful Respondent assumes sole and full responsibility for its acts and the acts of Successful Respondent Personnel relating to the performance of this CTSA.
- e) Successful Respondent agrees that any claim on behalf of any person arising out of employment, alleged employment, termination of employment, failing to employ, agency, or Subcontracts (including, but not limited to, claims of discrimination against Successful Respondent, its officers, or its agents or its Subcontractors) are the sole responsibility of Successful Respondent and are not the responsibility of DIR or any Customer, and that Successful Respondent will indemnify and hold harmless the State from any and all such claims asserted against the State, DIR or any Customer. Successful Respondent understands that any person who alleges a claim arising out of employment, alleged employment, termination of employment, failing to employ, agency, or Subcontract by Successful Respondent (including, but not limited to, claims of discrimination against Successful Respondent, its officers, or its agents or tis Subcontractors) will not be entitled to any compensation, rights, or benefits from DIR (including, but not limited to, tenure rights, medical and hospital care, sick and annual/vacation leave, severance pay, or retirement benefits).

## 6.3 Cooperation with DIR, Other Contract Holders, and Customers

#### 6.3.1 General

- a) Successful Respondent shall perform the Services in a manner that shall not: (1) disrupt or have an unnecessary adverse impact on the activities or operations of DIR, the Customers, or another TEX-AN 2021 Contract Holder; (2) degrade the Services then being received by DIR or the Customers; or (3) disrupt or interfere with the ability of DIR or the Customers to obtain the full benefit of the Services.
- b) In performing the Services and using the DIR Facilities, Successful Respondent shall observe and comply with all DIR and Customer policies, rules, and regulations applicable at or to DIR Facilities or the provision of the Services which have been communicated to Successful Respondent or Successful Respondent Personnel in advance in writing (which may include email communications and notices of information contained on web sites or conspicuously posted at DIR Facilities to the extent consistent with the way in which DIR or the Customers disseminate such information to their own employees) or by such means as are generally used by DIR or DIR Customers to disseminate such information to its employees or contractors and those applicable to specific DIR Facilities (collectively, DIR Rules). The Parties acknowledge and agree that, as of the Effective Date, the Successful Respondent is fully informed as to the DIR Rules that have been communicated to it consistent with the foregoing. Successful Respondent shall be responsible for the promulgation and distribution of DIR Rules to Successful Respondent Personnel as and to the extent necessary and appropriate. Additions or modifications to the DIR Rules may be: (i) disclosed to Successful Respondent and Successful Respondent Personnel in writing (which may include email communications and notices of information contained on web

Vendor Contract No.	
---------------------	--

sites to the extent consistent with the way in which DIR or Customers disseminate such information to their own employees); (ii) conspicuously posted at a DIR Facility; (iii) electronically posted; or (iv) communicated to Successful Respondent or Successful Respondent Personnel by means generally used by DIR and Customers to disseminate such information to its employees or contractors (including oral communications in immediately applicable or emergency situations). Successful Respondent and Successful Respondent Personnel shall observe and comply with such additional or modified DIR Rules.

#### 6.3.2 DIR Cost Avoidance

As part of the performance measures reported to State leadership, DIR must provide the cost avoidance the State has achieved through the CTSA. Upon request by DIR, Successful Respondent shall provide DIR with a detailed report of a representative sample of products sold under the CTSA. The report shall contain: product part number, product description, list price and price to Customer under the CTSA. Additionally, Successful Respondent shall promptly provide DIR with all other CTSA performance related information that assists DIR in reporting any other performance measures for which it is responsible.

#### 6.3.3 Cooperation with Other TEX-AN 2021 Contract Holders

Successful Respondent agrees to reasonably cooperate with and work with the other TEX-AN 2021 Contract Holders, Subcontractors, and third party representatives as requested by DIR. To the extent permitted by DIR's financial and personnel resources, DIR agrees to reasonably cooperate with Successful Respondent.

#### 6.3.4 Cooperation with Customers

Successful Respondent must ensure that Successful Respondent employees, agents, and Subcontractors will cooperate with DIR and Customers at no charge to DIR or Customers for purposes relating to the administration of the TEX-AN 2021 Program including, but not limited to the following purposes:

- a) The investigation and prosecution of fraud, abuse, and waste in the TEX-AN 2021 Program;
- b) Audit, inspection, or other investigative purposes; and
- c) Testimony in judicial or quasi-judicial proceedings relating to the Services under this CTSA or other delivery of information to DIR or Customers' investigators or legal staff.

#### 6.3.5 Successful Respondent's Subcontractor's Compliance

In all cases where Successful Respondent is required to ensure its Subcontractors' compliance with a section of this CTSA (or must otherwise require compliance by its Subcontractor with the terms of this CTSA, or is responsible for or required to indemnify the State for any taxes, benefits, compliance with law or other responsibilities of its Subcontractors, other than the performance of Services), Successful Respondent shall:

- a) include the necessary language of the section of the CTSA in the applicable Subcontract; and
- b) in consultation with DIR regarding the relative prudence of continuing to use the Subcontractor for Services as opposed to potentially terminating the Subcontract for breach, Successful Respondent shall, using commercially reasonable efforts, exercise contractual remedies upon such Subcontractor's non-compliance.

#### 6.4 Conduct of and Responsibility for Successful Respondent Employees

a) While performing the Services, Successful Respondent's Personnel must:

Vendor Contract No.	
---------------------	--

- 1) comply with applicable State, DIR, and Customer policies, procedures, rules, and regulations;
- comply with requests of DIR or Customers regarding personal and professional conduct generally applicable to the Service locations, including, but not limited to, conforming to all physical security standards and requirements, as required given the Service location;
- 3) attend workplace training offered by DIR and/or the Customers at DIR's or Customer's request; and
- 4) otherwise conduct themselves in a businesslike and professional manner.
- b) If DIR determines, in its sole and absolute discretion, that a particular employee, agent, or Subcontractor is not conducting him/herself or itself in accordance with this Section 4.4, DIR may provide Successful Respondent with notice and documentation concerning such conduct. Upon receipt of such notice, Successful Respondent must promptly investigate the matter and take appropriate action that may include:
  - 1) Removing the employee, agent, or Subcontractor from Service delivery under this CTSA;
  - 2) Providing DIR with written notice of such removal; and
  - 3) Replacing the employee, agent, or Subcontractor with a similarly qualified individual acceptable to DIR, in its sole and absolute discretion, in the case of personnel and Subcontractors.
- c) Successful Respondent shall replace any employee, agent, or Subcontractor who, as determined by DIR, in its sole and absolute discretion, after consultation with Successful Respondent, is unable to work effectively with DIR staff or Customer staff or is not adequately performing assigned responsibilities. In such event, Successful Respondent shall provide replacement employee(s), agent(s), or Subcontractor(s) with equal or greater skills and qualifications as soon as reasonably practicable. Replacement of personnel shall be subject to DIR prior review and written approval, which approval will not be unreasonably withheld or delayed. The Parties will work together in the event of any such required replacement so as not to disrupt Service delivery under the CTSA. Other than in instances when DIR determines, in its sole and absolute discretion, that removal must be handled immediately to avoid imminent harm to the TEX-AN 2021 Program or safety to others, DIR and Successful Respondent will provide written notice and cooperate with the other Party in a manner intended to minimize staff turnover and the adverse impact any requested removal may have on the Services. Nothing in this CTSA will prevent Successful Respondent from replacing employees. If the Parties have not been able to resolve DIR's concerns within five (5) Business Days of DIR communicating its concerns, Successful Respondent shall not assign the individual to that position and shall propose to DIR the assignment of another individual of suitable ability and qualifications.
- d) Successful Respondent agrees that anyone used by Successful Respondent to fulfill the terms of this CTSA is an employee, agent, or Subcontractor of Successful Respondent and remains under Successful Respondent's sole direction and control.
- e) Successful Respondent agrees to be responsible for the following with respect to Successful Respondent Personnel:
  - Damages incurred by Successful Respondent Personnel in the performance of their duties under this CTSA; and
  - 2) Determination of the hours to be worked and the duties to be performed by Successful Respondent Personnel.

3) Successful Respondent agrees and will inform Successful Respondent Personnel that there is no right of action against DIR or any Customer for any duty owed by Successful Respondent pursuant to this CTSA. Successful Respondent expressly agrees that neither DIR nor any Customer assumes any liability for the actions of, or judgments rendered against, the Successful Respondent, its employees, agents, or Subcontractors. Successful Respondent agrees that it has no right to indemnification or contribution from DIR or any Customer for any judgments rendered against Successful Respondent or Successful Respondent Personnel. DIR's liability to the Successful Respondent Personnel, if any, will be governed by Chapter 101, Texas Civil Practice & Remedies Code.

## **6.5** Responsibility for Subcontractors and Third Party Providers

- a) Successful Respondent remains fully responsible for obligations, services, and functions performed by its Subcontractors and third party providers to the same extent as if such obligations, services, and functions were performed by Successful Respondent's employees or agents and, for purposes of this CTSA, such work will be deemed work performed by Successful Respondent. DIR reserves the right to require the replacement of any Subcontractor found by DIR, in its sole and absolute discretion, to be unacceptable in accordance with CTSA Section 6.4.
- b) Successful Respondent must not disclose Confidential Information of DIR, a Customer, or the State to a Subcontractor unless and until such Subcontractor has agreed in writing to protect the confidentiality of such Confidential Information in the manner required of Successful Respondent under this CTSA.
- c) Successful Respondent must identify any Subcontractor that is a newly-formed subsidiary or entity, whether or not an Affiliate of Successful Respondent, substantiate the proposed Subcontractor's ability to perform the subcontracted Services, and certify to DIR that no loss of Service will occur as a result of the performance of such Subcontractor and provide any other information DIR may require. The Successful Respondent hereby assumes responsibility for all contractual responsibilities whether or not the Successful Respondent performs them. The Successful Respondent agrees that it is the sole point of contact for DIR with regard to all matters under this CTSA, including any matters with respect to Successful Respondent's Subcontractors.
- d) Successful Respondent shall not change a Subcontractor prior to submitting for DIR's approval, in its sole and absolute discretion, a revised HUB Subcontracting Plan. Only upon DIR's approval of a revised HUB Subcontracting Plan may Successful Respondent proceed with replacement of a Subcontractor.
- e) All Subcontracts are required to be in writing and signed by the Successful Respondent and Subcontractor

## 6.6 Security of Premises, Equipment, Data, and Personnel

Successful Respondent may, from time to time during the performance of the CTSA, have access to the premises, Equipment, Software, and other property, including data, files, information, and/or materials (collectively referred to as "Data") belonging to DIR or Customers. Successful Respondent will use its best efforts to preserve the safety, security, and the integrity of the premises, software, Equipment, Data, and other property of DIR or Customer, in accordance with the instruction of DIR or Customers. Successful Respondent will be responsible for damage to DIR or Customer's premises, Software, Equipment, Data, other property, and its contents, when such damage is caused by Successful Respondent Personnel.

Vandor	Contract No.	
venaor	Contract No.	

## **6.7** Background and/or Criminal History Investigation

Prior to commencement of any Services, background checks (including national fingerprint record checks and drug testing) and/or criminal history investigation of certain of the Successful Respondent Personnel who will be involved in the provision of Services, may be performed by or at the request of DIR or certain Customers having legislative authority to require such investigations. Should any Successful Respondent Personnel who will be involved in the provision of Services to DIR under the CTSA, or to a Customer under a CSA, not be acceptable to DIR or Customer, in their sole and absolute discretion, as a result of the background and/or criminal history check, then DIR or Customer, in their sole and absolute discretion, may immediately require replacement of the Successful Respondent Personnel in question.

## 7 NOTICES

#### **7.1** Delivery of Notice

Any notice or other legal communication required or permitted to be made or given by either Party pursuant to this CTSA, including any demand, will be in writing and deemed to have been duly given:

- a) Three (3) Business Days after the date of mailing if sent by certified U.S. mail, postage prepaid, with return receipt requested; or
- b) When delivered if delivered personally or sent by express courier service.

#### 7.2 Notice

Any notice under this CTSA will be sufficient if delivered to the following persons or their successors:

If to Successful Respondent:

Hughes Network Systems, LLC 11717 Exploration Lane Germantown, Maryland 20876 David.tuscano@hughes.com

With a copy to:

Matthew Brown

Telephone: 301-428-2707 <u>Matthew.brown@hughes.com</u>

If to DIR:

Chief Procurement Officer

Department of Information Resources

300 W. 15th St., Suite 1300

Austin, Texas 78701 Phone: (512) 475-4700

With copies to:

General Counsel

Department of Information Resources 300 W. 15th Street, Suite 1300

Vendor Contract No.

Austin, Texas 78701 Phone: (512) 475-4700 Facsimile: (512) 475-4759 DIRLegal@dir.texas.gov

and

Texas Department of Information Resources 300 West 15th Street, Suite 1300 Austin, Texas 78701 Attention: Director of Contract Management

Phone: (512) 475-4700

## **7.3** Change of Designee

Either Party may change the above-referenced designees or address with five (5) days' prior written notice to the other Party.

#### 7.4 Notification of Problems

Successful Respondent shall provide to DIR written notice of any acts or omissions (whether by DIR, a Customer, Successful Respondent, or any third party), any failure to perform any of each such party's obligations under the CTSA or CSA and any other events that may affect Successful Respondent's performance of Successful Respondent's obligations under the CTSA. Successful Respondent shall provide such written notice promptly after Successful Respondent first knew or should have known of such acts, omissions, or Failures or other events. Such written notice shall describe in reasonable detail such acts, omissions, Failures, or other events and the manner in which the foregoing may affect Successful Respondent's performance.

#### **7.5** Notification of Defaults

DIR and Successful Respondent shall promptly inform the other Party in writing of any breach of, misuse, or fraud in connection with any third party contract, Equipment lease, or third party materials license used in connection with the Services of which it becomes aware and shall cooperate with the other Party to prevent or stay any such breach, misuse, or fraud.

## 7.6 Notice of Adverse Impact

If Successful Respondent becomes aware of any failure of Successful Respondent to comply with its obligations under this Agreement or any other situation (a) that has impacted or reasonably could impact the maintenance of DIR's or any Customer's financial integrity or internal controls, the accuracy of DIR's or any Customer's financial, accounting or other records and reports, or compliance with DIR Rules, DIR policies or procedures or applicable laws, or (b) that has had or reasonably could have any other material adverse impact on the Services in question or the impacted operations of DIR or the Customers, then, Successful Respondent shall immediately inform DIR in writing of such situation and the impact or expected impact and Successful Respondent and DIR shall meet to formulate an action plan to minimize or eliminate the impact of such situation.

Vendor	Contract No.	
vendor	COMPACE INC.	

#### 8 GENERAL TERMS AND CONDITIONS

## **8.1** Delegation of Authority

Whenever, by any provision of this CTSA, any right, power, or duty is imposed or conferred on DIR, the right, power, or duty so imposed or conferred is possessed and exercised by the DIR Executive Director unless any such right, power, or duty is specifically delegated to the duly appointed agents or employees of DIR. The DIR Executive Director will reduce any such delegation of authority to writing and provide a copy to Successful Respondent on request.

## **8.2** No Waiver of Sovereign Immunity

The Parties expressly agree that no provision of this CTSA shall be construed as or constitute a waiver by DIR or the State of any immunities from suit or from liability that DIR or the State have by operation of law.

## **8.3** Force Majeure

#### 8.3.1 General

No Party shall be liable for any default or delay in the performance of its obligations under this CTSA if and to the extent such default or delay is caused, directly or indirectly, by fire, flood, earthquake, elements of nature or acts of God, wars, riots, civil disorders, rebellions or revolutions, acts of terrorism, epidemics, or any other similar cause beyond the reasonable control of such Party except to the extent that the non-performing Party is at fault in failing to prevent or causing such default or delay, and provided that such default or delay cannot reasonably be circumvented by the non-performing Party through the use of alternate sources, workaround plans, or other means. A strike, lockout, or labor dispute involving Successful Respondent personnel shall not excuse Successful Respondent from its obligations hereunder.

### 8.3.2 Duration and Notification

In the event of a Force Majeure Event, the non-performing Party shall be excused from further performance or observance of the obligation(s) so affected for as long as such circumstances prevail and such Party continues to use commercially reasonable efforts to recommence performance or observance whenever and to whatever extent possible without delay. Any Party so prevented, hindered, or delayed in its performance shall, as quickly as practicable under the circumstances, notify the Party to whom performance is due by telephone (to be confirmed in writing within one (1) day of the inception of such delay) and describe at a reasonable level of detail the circumstances of the Force Majeure Event, the steps being taken to address such Force Majeure Event, and the expected duration of such Force Majeure Event.

#### 8.3.3 Substitute Services: Termination

a) If any Force Majeure Event described above has substantially prevented, hindered, or delayed, or is reasonably expected to substantially prevent, hinder, or delay the performance by Successful Respondent of Services necessary for the performance of critical DIR or Customer functions for longer than twenty-four (24) hours, Successful Respondent shall, unless and until otherwise directed by DIR, use commercially reasonable efforts to procure such Services from an alternate source at Successful Respondent's expense for so long as the delay in performance shall continue, up to the Charges actually paid to Successful Respondent for the Services with respect to the period of nonperformance. If Successful Respondent is unable to procure such substitute Services

Vendor Contract No.	
---------------------	--

- on an expedited basis or DIR or Customer elects to contract directly for such Services, DIR or Customer may procure such Services from an alternate source at DIR's or Customer's expense.
- b) In addition, if any Force Majeure Event described above substantially prevents, hinders, or delays the performance by Successful Respondent of Services necessary for the performance of critical DIR or Customer functions: (i) for more than seven (7) days, then DIR or Customer may, upon notice to Successful Respondent, terminate all or any portion of the Services so affected (including portions that are no longer required if the impacted portion is terminated) as of the termination date specified in the notice; or (ii) for more than fifteen (15) days, then DIR may, upon notice to Successful Respondent, terminate this Agreement, in whole or in part, as of the termination date specified in the notice. Successful Respondent shall not have the right to additional payments or increased charges as a result of any Force Majeure Event affecting Successful Respondent's ability to perform.

### 8.3.4 Payment Obligation

If Successful Respondent fails to provide Services in accordance with this CTSA due to the occurrence of a Force Majeure Event, all amounts payable to Successful Respondent hereunder shall be adjusted downward so that DIR or Customer is not required to pay any amounts for Services that DIR and/or any Customer are not receiving, whether from Successful Respondent or from an alternate source at Successful Respondent's expense pursuant to above.

#### 8.3.5 Allocation of Resources

Without limiting Successful Respondent's obligations under this Agreement, whenever a Force Majeure Event causes Successful Respondent to allocate limited resources between or among Successful Respondent's customers and Affiliates, DIR and Customers shall receive at least the same treatment as comparable Successful Respondent customers.

## **8.4** Competitive Pricing

- a) Successful Respondent shall only charge Rates to DIR and Prices that are no higher in the aggregate than the charges charged by Successful Respondent directly to substantially similar retail third party customers for the provision of the same or substantially similar Services on substantially similar terms and conditions.
- b) No later than each Contract Anniversary Date, Successful Respondent shall certify to DIR, in writing, that it has been in full compliance with Section 8.4 a) above at all times during the prior twelve (12) month period. Any failure of Successful Respondent to comply with this certification requirement or any certification filed by Successful Respondent that is not true shall constitute a Material Breach. If Successful Respondent has breached Section 3.4 (a) above, then Successful Respondent shall be obligated to award DIR and/or Customer, as appropriate, Credits on the next appropriate Invoice that are equal to the overcharges paid as a result of any such breach.
- c) DIR shall have the right, in its sole and absolute discretion, to conduct an audit of Successful Respondent's compliance with this Section 8.4 pursuant to the procedures set forth in CTSA, Section 13.6.

#### 8.5 Publicity

a) Except as provided in Subsection (b) below, Successful Respondent must not use the name of DIR, the State, or any other Customer, or refer to DIR or any such Customer(s) directly or indirectly in any media release, or public announcement, relating to this CTSA or its subject matter, including, in any promotional or marketing materials, customer lists, or business

presentations (other than proposals or reports submitted to DIR, a Customer, an administrative agency of the State, or the federal government).

b) Successful Respondent may publish, at its sole expense, any media release or public announcement, relating to this CTSA or its subject matter, including, in any promotional or marketing materials, customer lists, business presentations (other than proposals or reports submitted to DIR or a Customer, an administrative agency of the State, or a governmental agency or unit of another state or the federal government), or results of Successful Respondent performance under this CTSA with DIR's prior review and approval, which DIR may exercise at its sole and absolute discretion. Successful Respondent will provide DIR a copy of any such publication no less than five (5) Business Days prior to its intended public release unless otherwise agreed by the Parties. Successful Respondent will provide additional copies at the request of DIR. Approval of the annual Marketing Plan will constitute approval by DIR for Successful Respondent to publish all materials approved in connection with such Marketing Plan.

## 8.6 Assignment

## 8.6.1 Assignment by Successful Respondent

Successful Respondent will not assign all or any portion of its rights under or interests in this CTSA (including by operation of law) or delegate any of its duties without prior written consent of DIR, which consent may be withheld in DIR's sole discretion. Any written request for assignment or delegation must be accompanied by written acceptance of the assignment by the Assignee, in a form acceptable to DIR in its sole and absolute discretion. Except where otherwise specifically agreed in writing by DIR, DIR's acceptance of any assignment or delegation does not release Successful Respondent from its obligations pursuant to this CTSA. The requirements of this Section 8.6.1 shall not apply in connection with Successful Respondent's assignment to (i) any Affiliate, or (ii) any person acquiring all or substantially all the assets of Successful Respondent or all or substantially all the assets of an Affiliate to which this CTSA, or any of the interests, rights, and obligations of Successful Respondent hereunder, previously has been assigned. DIR's consent to any assignment under this CTSA shall not constitute DIR's consent to further assignment.

#### 8.6.2 Assignment by DIR

DIR may, without the approval of Successful Respondent, assign or transfer its rights or obligations under this CTSA, in whole or in part, to any other State agency as directed by the State Legislature or as otherwise required under law. Successful Respondent understands and agrees DIR may in one (1) or more transactions assign or transfer this CTSA. DIR will provide written notification of such assignment to Successful Respondent.

#### 8.6.3 Amendment

Each Party agrees to cooperate to amend the CTSA as necessary to maintain an accurate record of the contracting Parties.

### 8.6.4 Assumption

Each Party to whom an assignment or transfer is made (an "Assignee") must assume all or any part of Successful Respondent's or DIR's interests in this CTSA, the Services, and any documents executed with respect to this CTSA, including, without limitation, its obligation for all or any portion of the payments due hereunder.

Vendor	Contract No.	
vendor	COMBACE NO.	

#### 8.6.5 Impermissible Assignment

Any attempted assignment that does not comply with the terms of this section shall be null and void ab initio; provided, however, that if Successful Respondent assigns this CTSA in contravention of this section by operation of law, such assignment shall be voidable at the option of DIR.

#### 8.7 RFO Errors and/or Omissions

Successful Respondent will not take advantage of or exploit any errors and/or omissions in the RFO or the resulting CTSA. Successful Respondent must promptly notify DIR of any such errors and/or omissions that are discovered.

#### 8.8 Abandonment or Default

Abandon means to relinquish or renounce the CTSA for any reason other than for a Force Majeure Event. If the Successful Respondent abandons the CTSA, DIR reserves the right to cancel the CTSA without notice and either re-solicit and re-award the CTSA or take such further action or no action in the best interest of the State.

#### **8.9** Time is of the Essence

In consideration of the need to ensure uninterrupted and continuous Services, the Parties agree that time is of the essence in the performance of this CTSA.

#### **8.10** Place of Performance

Unless otherwise agreed to in writing, all Services performed by Successful Respondent, its Subcontractors, and agents under the CTSA must be performed in the contiguous United States.

#### **8.11** Education Department General Administrative Regulations (EDGAR)

The Education Department of General Administrative Regulations (EDGAR) are the federal regulations that govern all federal grants awarded by the U.S. Department of Education on or after December 26, 2014. EDGAR encourages the use of cooperative agreements for procurement or use of common or share goods and services in order to foster greater economy and efficiency. DIR uses an open market competitive procurement process to award contracts as required by State law, including Texas Government Code Chapters 2054 and 2157. If Successful Respondent provides evidence of its EDGAR compliance to DIR's satisfaction, DIR may identify Successful Respondent as certifying that all or a portion of Successful Respondent's listings are EDGAR eligible, and DIR may then permit Successful Respondent to so identify all or part of its offerings on Successful Respondent's DIR website. In such cases, upon request from eligible Customer(s), Successful Respondent must complete EDGAR certification affirmation forms to satisfy Customer requirement(s).

## 9 AMENDMENTS, MODIFICATIONS, AND PROCEDURES

#### 9.1 General Amendments and Modifications

The Parties developed appropriate procedures for amending or modifying the CTSA in this Section 9. Except for the amendments subject to the below subsections, this CTSA may be amended by mutual written agreement. Amendments may be made for changes in law, circumstances affecting performance of the Services, or as otherwise agreed to by the Parties.

Vendor Contract No.	
---------------------	--

## 9.2 Amendments and Modifications Resulting from Changes in Law or Contract and as Mutually Agreed

Except for changes subject to Section 9.1 above, this CTSA may be amended by mutual written agreement of the Parties if changes in federal or State laws, rules, regulations, policies, guidelines, or circumstances affect the performance of the Services, provisions need to be added or amended within the CTSA, or as otherwise agreed to in writing by the Parties.

## **9.3** Required Compliance with Amendment and Modification Procedures

No different or additional Services will be authorized or performed except pursuant to amendment or modification of this CTSA that is executed in compliance with the CTSA. No waiver of any term, covenant, or condition of this CTSA will be valid unless executed by both Parties. Successful Respondent shall not be entitled to payment for any Services that are not authorized by this CTSA or a properly executed amendment or modification to the CTSA.

## **9.4** Modifications Resulting from Imposition of Remedies

This CTSA may be modified under the terms of CTSA Article 16 Remedies and Disputes.

## 9.5 Adjustments to Exhibit B, Attachment B-1 Initiated by Either Party.

- a) The Successful Respondent may propose changes in the contents of **Exhibit B** for Services of this CTSA.
- b) The process for submitting, reviewing, and approving/disapproving and appealing proposed adjustments or recommendations for changes to any element contained in Exhibit B, or the Attachments, shall be in accordance with this section and Article 3 of Exhibit B to the CTSA. The Parties desire to develop and implement efficiency updates to the Electronic Administrative Update (EAU) fast-track system process for the continuous improvement of the Rates to DIR and Prices available to the Customers under the CTSA, and other non-substantive administrative updates, without reliance on a formal amendment process. An approved change to the contents of Attachments to Exhibit B, approved in accordance with this section and Article 3 of Exhibit B, shall be considered an approved amendment of the CTSA for all purposes.

#### 9.6 Benchmarking

DIR reserves the right to benchmark pricing of Services offered by Successful Respondent throughout the Term of the Contract, and to request price reductions to Successful Respondent Services that benchmarking may show to be outside industry standard pricing. Successful Respondent agrees to work in good faith with DIR to bring pricing into alignment in the best interest of the State.

## 9.7 All Other Changes

All other modifications to the CTSA must be accomplished through a formal written amendment executed by an authorized representative of DIR and Successful Respondent. DIR reserves the right to require periodic updates of the CTSA hereto, to maintain alignment with new and changed legislation, legal requirements, rules, and regulatory requirements relevant to the performance by Successful Respondent of this CTSA.

Vendor	Contract No.	
vendor	COMPACE INC.	

## **9.8** Mutual Covenant of Cooperation

For all such CTSA amendments, the Parties covenant to cooperate in the good faith and timely negotiation of such matters and to proceed to execution of necessary instruments without delay.

#### 10 AVAILABILITY OF APPROPRIATED FUNDS AND STATUTORY AUTHORITY

- a) This CTSA is expressly conditioned on the availability of State and local government appropriated funds. Successful Respondent will have no right of action against DIR in the event that DIR is unable to perform its obligations under this CTSA as a result of the suspension, termination, withdrawal, or failure of funding to DIR or lack of sufficient funding of DIR, or, by extension, failure or lack of funding from Customers, including non-appropriation or exercise of emergency budget execution authority for any Services, activities, or functions contained within the scope of this CTSA. DIR will negotiate in good faith with Successful Respondent to resolve any Successful Respondent claims for payment for Accepted Services that are pending at the time funds become unavailable. DIR will make reasonable efforts to provide written advance notice to Successful Respondent upon learning that funding for this CTSA may be changed or discontinued as set forth above.
- b) If DIR, any Customer(s), and/or the subject matter of this CTSA become subject to a legislative or regulatory change or the revocation of statutory or regulatory authority that would (1) render the continued provision of the Services impossible or unnecessary, (2) render this CTSA invalid, illegal, or otherwise unenforceable, (3) substantially decrease the amount and types of the Services, or (4) terminate the appropriations for this CTSA, then DIR may, upon notice to Successful Respondent, terminate this Agreement, in whole or in part, as of the termination date specified in the notice.

## 11 GOVERNING LAW AND REGULATIONS

#### **11.1** Governing Law and Venue

This CTSA and the rights and obligations of the Parties under this CTSA is governed by and construed in accordance with the laws of the State of Texas without regard to conflict of laws principles. Provided Successful Respondent first complies with the procedures set forth in Section 16.11 Dispute Resolution, proper venue for a claim arising from this CTSA will be in a State court in Travis County, Texas. Nothing in this CTSA shall be construed to waive the State's sovereign immunity.

## 11.2 Successful Respondent Responsibility for Compliance with Laws and Regulations

- a) Successful Respondent represents, warrants, and covenants that, with respect to the provision of the Services and the performance of any of Successful Respondent's other legal and contractual obligations hereunder, Successful Respondent is and shall be in compliance in all material respects with all applicable laws and shall remain in compliance with such laws during the Term, including, but not limited to, all State and federal telecommunications laws, tax laws, State and federal employment and immigration laws, State and federal regulatory requirements, licensing provisions, and identifying and procuring applicable permits, certificates, approvals, and inspections required under such laws.
- b) Successful Respondent is responsible for ensuring each of its Successful Respondent Personnel who are involved in the provision of Services under this CTSA are legally eligible to work in the

Vendor Contract No.	
---------------------	--

State and the United States, properly licensed, certified, and/or have proper permits to perform any activity related to the Services.

- c) If any charge of non-compliance by Successful Respondent with any such laws occurs or Successful Respondent is aware that it is not in compliance with such laws that could have a material adverse impact on the performance, receipt, or use of the Services, Successful Respondent shall promptly notify DIR of such charge or non-compliance, as applicable. Successful Respondent warrants that the Services comply with all applicable federal, State, county, and municipal laws, regulations, codes, ordinances, guidelines, and policies. Successful Respondent will indemnify DIR and any Customer from and against any losses, liability, claims, damages, penalties, costs, fees, or expenses arising from or in connection with Successful Respondent's (including Successful Respondent Personnel, third party providers, or Subcontractors) failure to comply with or violation of any federal, State, county, and/or municipal law, regulation, code, ordinance, or policy.
- d) At no additional charge, Successful Respondent shall provide DIR with data and reports in Successful Respondent's possession as reasonably necessary for DIR to comply with all laws applicable to the Services (including the Equipment, materials, and other resources it provides or has assumed operational responsibility hereunder).
- e) Successful Respondent represents, warrants, and covenants that the materials, Equipment, and systems owned, developed, implemented, provided, or used by Successful Respondent in providing the Services are in compliance with all applicable laws and shall remain in compliance with such laws during the Term.
- f) In the event of any changes in laws (including laws other than applicable to the provision of the Services and the performance of any of Successful Respondent's other legal and contractual obligations hereunder), Successful Respondent shall implement any necessary modifications to the Services, materials, Equipment, and systems prior to the deadline imposed by the regulatory or governmental body having jurisdiction for such requirement or change. Successful Respondent shall bear the costs associated with compliance with laws.

## 11.3 Compliance with Immigration Laws

Successful Respondent shall comply with the requirements related to federal immigration laws and regulations, including but not limited to the Immigration and Reform Act of 1986, the Illegal Immigration Reform and Immigrant Responsibility Act of 1996 ("IIRIRA"), and the Immigration Act of 1990 (8 U.S.C.1101, et seq.) regarding employment verification and retention of verification forms for any individual(s) hired on or after the effective date of the IIRIRA, who will perform any labor or services under this CTSA. Nothing herein is intended to exclude compliance by Successful Respondent with all other relevant federal immigration statutes and regulations promulgated pursuant thereto.

## **11.4** Equal Opportunity Compliance

Successful Respondent represents, warrants, and covenants that it shall abide by all applicable laws, regulations, and executive orders pertaining to equal employment opportunity, including federal or State laws and the laws of the state in which its primary place of business is located. In accordance with such laws, regulations, and executive orders, the Successful Respondent agrees that no person in the United States will, on the grounds of race, color, religion, national origin, sex, age, veteran status, or disability, be excluded from employment with or participation in, be denied the benefits of, or be otherwise subjected to discrimination under any Services or activity performed by Successful Respondent under this CTSA. If Successful Respondent is found by DIR, in its sole and absolute discretion, any regulator with

Vendor Contract No.	
---------------------	--

jurisdiction over Successful Respondent's Services, or a court of competent jurisdiction to not be in compliance with these requirements during the Term of this CTSA, Successful Respondent agrees to take appropriate steps to correct these deficiencies. Upon request, Successful Respondent will furnish to DIR information regarding its non-discriminatory hiring and promotion policies, as well as specific information on the composition of its principals and staff, including the identification of minorities and women in management or other positions with discretionary or decision-making authority.

### 11.5 Historically Underutilized Business (HUB) Program

Successful Respondent shall comply with Chapter 2161, Texas Government Code, in the provision of Services under this CTSA. Successful Respondent's HUB Subcontracting Plan is incorporated into the CTSA. Successful Respondent shall comply with DIR's then-current Policy on Utilization of Historically Underutilized Businesses for reporting HUB subcontractor participation in the CTSA and for changing the HUB Subcontracting Plan throughout the Term.

## 12 PERFORMANCE MEASUREMENT

Successful Respondent performance of this CTSA will be measured by:

- a) Adherence to this CTSA, including all representations, warranties, certifications, and covenants;
- Delivery of the Services in accordance with the Service Level Agreements agreed to in Exhibit C Performance Management;
- c) Results of audits performed by DIR or its representatives in accordance with Article 13;
- d) Timeliness, completeness, and accuracy of required reports agreed to in this CTSA and as stated in the Master Vendor Reporting Guide; and
- e) Achievement of other performance measures developed and agreed to by Successful Respondent and DIR and as modified from time to time by written agreement.

#### 13 AUDIT AND FINANCIAL COMPLIANCE

#### 13.1 CTSA Record Retention and Audit

Successful Respondent agrees, and shall cause its agents and its Subcontractors to agree, to maintain complete and accurate records of, and supporting documentation for, all charges, transactions, authorizations, changes, implementations, reports, filings, returns, analyses, procedures, controls, records, data or information created, generates, collected, processed, or stored by Successful Respondent in the performance of its obligations this CTSA or any CSA, and are sufficient to ensure the accuracy and validity of Monthly Consolidated Invoices to DIR and direct billing to Customers for certain Direct Sales Transaction Services (collectively "CTSA Records"). Successful Respondent agrees, and shall cause its agents and Subcontractors to agree, to maintain and retain CTSA Records, including all original forms, records, and associated documentation, for a period of seven (7) years after the date of submission of the final billing or until the resolution of all litigation, claims, financial management reviews, or audits pertaining to this CTSA, whichever is longer.

Vendor	Contract No.	
--------	--------------	--

## 13.2 Access to Records, Books, and Documents

- a) Upon notice by DIR, Successful Respondent must provide, and cause its agents and its Subcontractors to provide, the officials and/or entities identified in this Section 13.2 with prompt, reasonable, and adequate access to any requested CTSA Records.
- b) Successful Respondent and its agents and its Subcontractors must provide the access described in this Section 13.2 upon DIR's request. This request may be for, but is not limited to, the following purposes:
  - 1) Examination:
  - 2) Audit;
  - 3) Investigation;
  - 4) CTSA administration;
  - 5) Compliance with the Texas Public Information Act; or
  - 6) The making of copies, excerpts, or transcripts.
- c) The access required must be provided to the following officials and/or entities:
  - 1) Any Independent Verification and Validation contractor or quality assurance contractor, when acting on behalf of DIR;
  - 2) The Office of the State Auditor, the CPA Claims Division, or their designee;
  - 3) A State or federal agency;
  - 4) A special or general investigating committee of the Texas Legislature or its designee; and
  - 5) Any auditor or other entity identified by DIR.
- d) Successful Respondent agrees, and will require its agents and its Subcontractors to agree, to provide the access described wherever Successful Respondent maintains such books, records, and supporting documentation. Successful Respondent further agrees, and will require its agents and its Subcontractors to agree, to provide such access in reasonable comfort and to provide any furnishings, Equipment, or other conveniences deemed reasonably necessary to fulfill the purposes described in this Section 13.2.

## **13.3** Audits by Governmental Authorities

Acceptance of funds under the CTSA by Successful Respondent acts as acceptance of the authority of the State Auditor's Office, the CPA Claims Division, and any successor governmental authorities, to conduct audits and investigations in connection with those funds. The State Auditor's Office and the CPA Claims Division shall at any time have access to and rights to examine, audit, excerpt, and transcribe any pertinent books, documents, working papers, and records of Successful Respondent relating to this CTSA, notwithstanding any other provision of this CTSA. Successful Respondent, and its agents and its Subcontractors, agree to cooperate fully with and provide all assistance requested by the State Auditor's Office or the CPA Claims Division or its successor in the conduct of such audits or investigations, including providing all records requested. In addition to and without limitation on the other audit provisions of this CTSA, pursuant to Section 2262.154, Texas Government Code, the State Auditor's Office may conduct an audit or investigation of Successful Respondent or any other entity or person receiving funds from the State directly through this CTSA or indirectly through a subcontract under this CTSA. The acceptance of funds by Successful Respondent or any other entity or person directly under this Agreement or indirectly through a subcontract under this CTSA acts as acceptance of the authority of the State Auditor's Office, under the direction of the Legislative Audit Committee, to conduct an audit or investigation in connection with those funds. This CTSA may be unilaterally amended by DIR upon notice to Successful Respondent to bring this CTSA into compliance with any rules and procedures of the

State Auditor's Office in the implementation and enforcement of Section 2262.154, Texas Government Code. Successful Respondent will ensure that this clause concerning the authority to audit funds received indirectly by Subcontractors through Successful Respondent and the requirement to cooperate is included in any subcontract it awards pertaining to the CTSA. Under the direction of the Legislative Audit Committee, a Successful Respondent that is the subject of an audit or investigation by the State Auditor's Office must provide the State Auditor's Office with access to any information the State Auditor's Office considers relevant to the investigation or audit.

#### 13.4 Audits of Services, Deliverables, and Inspections

- Upon notice from DIR, Successful Respondent will provide, and will cause its agents and its Subcontractors to provide, such auditors and inspectors as DIR may from time to time designate, with access to:
  - 1) Successful Respondent, agent, and Subcontractor Service locations, facilities, or installations; and
  - 2) Successful Respondent, agent, and Subcontractor Software and Equipment.
- b) Successful Respondent must provide, and shall cause its agents and its Subcontractors to provide, as part of the Services, any assistance that such auditors and inspectors may require to complete such audits or inspections.

#### 13.5 Response/Compliance with Audit or Inspection Findings

- a) If an audit by a governmental body, standards organization, or regulatory authority having jurisdiction over DIR, a Customer, or Successful Respondent results in a finding that Successful Respondent is not in compliance with any applicable law or standard, including any generally accepted accounting principle or other audit requirement relating to the performance of its obligations under this CTSA, Successful Respondent shall, if and to the extent such audit deficiency or finding of non-compliance results from Successful Respondent's failure to comply with its obligations under this CTSA, at its own expense and within the time period specified by such auditor, address and resolve the deficiency(ies) identified by such audit governmental body, standards organization, or regulatory authority, in the manner approved by DIR, to the extent necessary to comply with Successful Respondent's obligations under this CTSA.
- b) Successful Respondent must take action to ensure its employees', agents', or Subcontractors' compliance with or correction of any finding of non-compliance with any law, regulation, audit requirement, or generally accepted accounting principle relating to the Services or any other Deficiency contained in any audit, review, or inspection conducted under this Article 13. This action will include Successful Respondent's delivery to DIR, for DIR's approval, a Corrective Action Plan in accordance with Section 17.2 that addresses deficiencies identified in any audit(s), review(s), or inspection(s) within ten (10) days of the close of the audit(s), review(s), or inspection(s).
- c) As part of the Services, Successful Respondent must disclose to DIR, upon DIR's request, and will include in its Subcontracts a requirement that each of its Subcontractors will disclose upon request, a copy of those portions of its internal audit reports relating to the Services. If Successful Respondent determines as a result of its own internal audit that it has overcharged DIR or Customer, then Successful Respondent shall promptly pay to DIR or Customer the amount of such overcharge, together with interest at the rate specified by the CPA in accordance with Section 2251.025(b), Texas Government Code, from the date of receipt by Successful Respondent of the overcharged amount until the date of payment to DIR or Customer. In the

Vendor Contract No.	
---------------------	--

event such an audit results in a determination that Successful Respondent has undercharged DIR or Customer, then, subject to DIR's right to dispute the amount of such undercharge, Successful Respondent shall promptly report such undercharge to DIR and may invoice DIR for such undercharged amounts.

## 13.6 Audit of Telecommunications Fees and Surcharges, Rates to DIR, Prices, and Assessments Conducted Pursuant to Section 13.3 of the CTSA

- a) Successful Respondent will provide, and will cause its agents and its Subcontractors to provide, to DIR and its designees access to such financial records and supporting documentation requested by DIR to audit telecommunications fees and surcharges, Rates to DIR, Prices, and any assessment conducted pursuant to Section 13.3 of the CTSA. Successful Respondent's obligations hereunder with respect to its Subcontractors shall be fulfilled by complying with the Subcontractor Compliance Obligations.
- b) In addition to its monthly review and payment of Monthly Consolidated Invoices under the CTSA, DIR may audit the applicable telecommunications fees and surcharges, Rates to DIR, Prices, and any assessments conducted under Section 13.3 of the CTSA to determine that such telecommunications fees and surcharges, Rates to DIR, Prices to Direct Sales Transaction Service Customers, and any assessments conducted pursuant to Section 13.3 of the CTSA are accurate and in accordance with this CTSA.
- c) If, as a result of such audit, DIR determines that Successful Respondent has overcharged the State, DIR will notify Successful Respondent in writing of the amount of such overcharge, and provide to the Successful Respondent the work papers supporting such position. Successful Respondent will promptly pay to DIR the undisputed amount of the overcharge, plus interest as calculated in accordance with Chapter 2251, Texas Government Code, for late payments, and the Parties will use good faith efforts to resolve any disputed amounts as soon as practicable. In the event such audit reveals an overcharge of five percent (5%) or more to DIR which is not disputed and is in excess of five percent (5%) of the total invoiced amounts for such period, Successful Respondent will reimburse DIR for the reasonable cost of this audit.

## 14 TERMS AND CONDITIONS OF SUCCESSFUL RESPONDENT BILLING AND DIR AND CUSTOMER PAYMENT

## **14.1** Monthly Consolidated Invoice to DIR

The Successful Respondent shall provide DIR with a Monthly Consolidated Invoice for all Services whereby DIR acts as the Successful Respondent's billing agent provided by the Successful Respondent, at the Rates to DIR and with any applicable telecommunications fees and surcharges as allowed by this CTSA, in electronic format, with the ability to batch load. All such Monthly Consolidated Invoices shall conform to the standards as set forth in Appendix B, the Master Vendor Reporting Guide and Exhibit D, Billing Plan.

#### 14.2 Invoices for Direct Sales Transaction to Customers

The Successful Respondent shall provide to each Direct Sales Transaction Service Customer a monthly Invoice for the Direct Sales Transaction Service Services provided to such Customer, in electronic format, at the Prices (Rates to DIR plus Administrative Fee) and with any applicable telecommunications fees and surcharges as allowed by this CTSA.

Vendor (	Contract No.	
----------	--------------	--

## 14.3 Limitations on the Right to Bill

The Successful Respondent shall not bill for any Services prior to the Service Order Completion Notice (SOCN) date for new Orders, and for disconnect Orders, the billing end date shall equal the disconnect date on a SOCN.

## 14.4 Telecommunications Fees, Surcharges, and Taxes

- a) All applicable Federal Communication Commission (FCC), Texas Public Utilities Commission (PUC), or other authorized telecommunications fees and surcharges in effect as of the date of this CTSA which may appear on billings to Customers are disclosed in the appropriate section per Service in Exhibit B to the CTSA. Any additional FCC, PUC, or other authorized telecommunications fees or surcharges applicable to any Service may only be imposed during the Term, upon a written amendment of the CTSA to authorize such imposition under the CTSA. Successful Respondent acknowledges that certain Customers that are political subdivisions of the State, are exempt from the imposition and collection of certain State telecommunications fees, including the Texas Universal Service Fund charge and the Texas Infrastructure Fund assessment. In addition, State agency Customers have additional exemptions from State telecommunications fees, including the 9-1-1 emergency service fee, 9-1-1 equalization surcharge, poison control surcharge, and late charges imposed under Section 55.010, Texas Utilities Code. Successful Respondent agrees to not bill for any items which are not mandated by the FCC, PUC, or other proper authority and which are otherwise not applicable to the Services and for which Successful Respondent has requested and received valid exemption certificates from Customers. Successful Respondent agrees to promptly correct any incorrect billings of telecommunications fees and surcharges that occur. Successful Respondent acknowledges that DIR makes no representation about the exemption status of any Customers. Successful Respondent must request and receive any exemption certificates that may apply from each assistance organization directly.
- b) During the Term, all changes in the law or fee structures, which creates or authorizes Successful Respondent to impose an unlisted telecommunications fee or surcharge on the Services, which Successful Respondent desires to impose, shall require an amendment in order to be effective against the State, DIR, and Customers. In the event of a change in the law or telecommunications fees and surcharges structures, which results in an exemption from payment in favor of the State, DIR, and/or Customers, Successful Respondent shall give effect to the exemption without the necessity of an amendment hereto.
- c) Successful Respondent acknowledges that certain Government Entity Customers are exempt from State sales, use, and excise taxes, Section 151.309, Texas Tax Code, and federal excise tax, 26 USC Sections 4253 (i) and (j). Successful Respondent further acknowledges that State agency Customers are exempt from the assessment and collection of sales taxes imposed by political subdivisions of the State, in accordance with Sections 321.208 (municipalities) and 323.207 (counties), Texas Tax Code. The Government Entity Customers shall issue a tax exemption certificate upon request to Successful Respondent. Successful Respondent acknowledges that DIR makes no representation about the exemption status of any Customers. Successful Respondent must request and receive any exemption certificates that may apply from each assistance organizations directly.

## 14.5 No Rights of Set-off

There is no general right of set-off for Successful Respondent against amounts owed to DIR under this CTSA. To collect Credits under Article 17, Remedies and Disputes, DIR may set-off the amount of

Vendor Contract No.	
---------------------	--

assessed Credits from amounts otherwise owed to Successful Respondent in accordance with the terms of Section 17.10. There is no general right of set-off for Successful Respondent against amounts owed by a Customer under any CSA.

## 14.6 Expenses

Except as provided in this CTSA with respect to Rates to DIR, Prices, and authorized telecommunications fees and surcharges, all other expenses incurred by Successful Respondent in connection with its provision of the Services (including management; travel and lodging; document reproduction and shipping; Equipment, Software required by Successful Respondent Personnel; and long-distance telephone) will not be paid by DIR or Customer unless agreed upon by DIR or Customer in writing. Successful Respondent hereby waives any and all claims for additional compensation related to the Services except as set forth in Exhibit B hereto.

## **14.7** Dispute Handling and Adjustments

- a) A Billing Inquiry is any question or issue about a Monthly Consolidated Invoice that may lead to a Billing Dispute. It is DIR's responsibility to validate Successful Respondent Monthly Consolidated Invoices each month. During that process, Billing Inquiries will arise, and the Successful Respondent shall timely answer or respond to billing inquiries. The Successful Respondent shall respond within one (1) Business Day of receipt of the Billing Inquiry.
- b) DIR may withhold any amount of any Invoice in dispute as provided in and in accordance with Chapter 2251, Texas Government Code. DIR shall comply with Chapter 2251, Texas Government Code, with respect to timely notice of such disputed amounts.
- c) For Invoice amounts that have been paid by DIR that become the subject of a Billing Dispute, in addition to any amounts that Successful Respondent may subsequently credit to DIR arising from resolution of such dispute, Successful Respondent shall include interest on the reimbursed amounts accrued monthly at the rate specified by the CPA in accordance with Section 2251.025(b), Texas Government Code, calculated from the ninetieth (90th) day following the date of DIR's Billing Dispute.
- d) DIR and Customer have the right to file a Billing Dispute. The Successful Respondent shall promptly resolve Billing Disputes that involve amounts less than or equal to \$15,000 no later than sixty (60) calendar days from the date DIR or Customer notifies Successful Respondent of the Billing Dispute and, for Billing Disputes that involve amounts greater than \$15,000, no later than ninety (90) calendar days from the date DIR or Customer notifies Successful Respondent of the Billing Dispute. This time frame starts with the notification of the Billing Dispute by DIR and ends with the satisfactory resolution of the Billing Dispute. Billing Disputes not resolved within one-hundred twenty (120) days of submission of the Billing Dispute will automatically resolve in favor of DIR or Customer. Successful Respondent shall issue Adjustment(s) within thirty (30) calendar days of resolution. DIR may require a revised Monthly Consolidated Invoice that does not contain the disputed item(s). Successful Respondent shall allow non-payment by DIR for the disputed amount until there is a dispute resolution, as allowed by Chapter 2251, Texas Government Code.
- e) Direct Sales Transaction Service Customers shall have the same rights as DIR under Section 14.7, Subsection (a) and (b) above, for the Invoices they receive for Direct Sales Transaction Services.
- f) If DIR or a Direct Sales Transaction Service Customer disputes payment of all or any portion of an Invoice from the Successful Respondent, DIR or the Direct Sales Transaction Service Customer, respectively, will notify Successful Respondent of such Billing Dispute and the

Parties, either DIR and Successful Respondent or the Direct Sales Transaction Service Customer and Successful Respondent, as appropriate, will attempt in good faith to resolve the Billing Dispute in conformance with Chapter 2251, Texas Government Code. DIR and the Direct Sales Transaction Service Customer will not be required to pay any disputed portion of a Successful Respondent Invoice. Notwithstanding any such Billing Dispute, the Successful Respondent must continue to perform the Services in compliance with the terms of this CTSA pending resolution of Billing Disputes so long as all undisputed amounts continue to be paid to Successful Respondent.

# **14.8** Billing Adjustments

- a) A billing Adjustment is a modification or correction of a billing amount of an element of a Service or a Service Level Agreement (SLA) Credit. Billing Adjustments may be made on both recurring and non-recurring charges, and are used to settle Disputes and financial discrepancies.
- b) Successful Respondent shall include in its Monthly Consolidated Invoice to DIR or its Invoices to Direct Sales Transaction Service Customers, the following minimum items as billing Adjustments:
  - Correction of a billing error and any related interest discovered by Successful Respondent, DIR, or the Direct Sales Transaction Service Customer, as appropriate;
  - 2) Back-billing, as discussed below;
  - 3) SLA Credits; and
  - 4) Any other elements Successful Respondent considers Adjustments as set forth in its Billing Plan, Exhibit D hereto.
- c) The Successful Respondent shall resolve all Billing Disputes by any one of the three following approaches:
  - 1) Issue a proposed dispute resolution for the full amount;
  - 2) Provide evidence acceptable to DIR or the Direct Sales Transaction Service Customer, as appropriate that the disputed amount will be reduced; or
  - 3) Provide evidence acceptable to DIR or the Direct Sales Transaction Service Customer, as appropriate, that the Billing Dispute is not valid.
- d) For all Billing Disputes, Successful Respondent shall receive agreement or denial from DIR or the Direct Sales Transaction Service Customer, in their sole and absolute discretion, as appropriate, on the proposed resolution before issuing a billing Adjustment.

## **14.9** Back-billing and POP Moves

- a) Rates to DIR or Prices for Services inadvertently left off previous Monthly Consolidated Invoices or direct Invoices by the Successful Respondent shall be back-billed no more than ninety (90) calendar days for usage and one-hundred twenty (120) calendar days for circuits (however, in no case may the back-billing include a time period prior to the delivery of the SOCN). Pursuant to PUC Rule, 16 Texas Administrative Code 26.27, a six (6) month limit from the date of discovery of an error applies to back-billing of all billed Services. Back-billing shall be included in the Adjustments.
- b) Successful Respondent Initiated POP location moves shall not adversely affect the existing Rates to DIR and/or Prices to Direct Sales Transaction Service Customers for a Service.

Vandor	Contract No.	
venaor	Contract No.	

## 14.10 Liability for Taxes, Insurance, and Indemnification

- a) SUCCESSFUL RESPONDENT AGREES AND ACKNOWLEDGES THAT DURING THE EXISTENCE OF THIS CTSA, SUCCESSFUL RESPONDENT WILL BE ENTIRELY RESPONSIBLE FOR THE LIABILITY AND PAYMENT OF SUCCESSFUL RESPONDENT PERSONNEL'S TAXES OF WHATEVER KIND, ARISING OUT OF THE PERFORMANCES IN THIS CTSA. SUCCESSFUL RESPONDENT AGREES TO COMPLY WITH ALL STATE AND FEDERAL LAWS APPLICABLE TO ANY SUCH PERSONS, INCLUDING LAWS REGARDING WAGES, TAXES, INSURANCE, AND WORKERS' COMPENSATION. SUCCESSFUL RESPONDENT AGREES AND ACKNOWLEDGES THAT SUCCESSFUL RESPONDENT'S PERSONNEL WILL NOT BE ENTITLED TO ANY STATE BENEFIT OR BENEFIT OF ANOTHER GOVERNMENTAL ENTITY CUSTOMER. DIR AND/OR THE STATE WILL NOT BE LIABLE TO THE SUCCESSFUL RESPONDENT, SUCCESSFUL RESPONDENT PERSONNEL, OR OTHERS FOR THE PAYMENT OF TAXES OR THE PROVISION OF UNEMPLOYMENT INSURANCE AND/OR WORKERS' COMPENSATION OR ANY BENEFIT AVAILABLE TO A STATE EMPLOYEE OR EMPLOYEE OF ANOTHER GOVERNMENTAL ENTITY CUSTOMER.
- b) SUCCESSFUL RESPONDENT AGREES TO INDEMNIFY AND HOLD HARMLESS DIR, ALL CUSTOMERS, THE STATE, AND/OR EACH OF THEIR EMPLOYEES, AGENTS, REPRESENTATIVES, CONTRACTORS, ASSIGNEES, AND/OR DESIGNEES FROM ANY AND ALL LIABILITY, ACTIONS, CLAIMS, DEMANDS, OR SUITS, AND ALL RELATED COSTS, ATTORNEY FEES, AND EXPENSES, RELATING TO TAX LIABILITY, UNEMPLOYMENT INSURANCE, AND/OR WORKERS' COMPENSATION OR EXPECTATIONS OF BENEFITS BY SUCCESSFUL RESPONDENT OR SUCCESSFUL RESPONDENT PERSONNEL IN THEIR PERFORMANCE UNDER THIS CTSA. SUCCESSFUL RESPONDENT WILL BE LIABLE TO PAY ALL COSTS OF DEFENSE INCLUDING ATTORNEYS' FEES AND OUT OF POCKET EXPENSES. THE DEFENSE WILL BE COORDINATED BY THE OFFICE OF THE ATTORNEY GENERAL FOR TEXAS STATE AGENCY CUSTOMERS AND BY CUSTOMER'S LEGAL COUNSEL FOR NON-STATE AGENCY CUSTOMERS.
- c) NOTWITHSTANDING ANYTHING CONTAINED HEREIN TO THE CONTRARY, THE PARTIES AGREE THAT TO THE EXTENT THE FOREGOING PROVISIONS IMPOSE OBLIGATIONS WITH RESPECT TO SUBCONTRACTORS UPON SUCCESSFUL RESPONDENT, SUCCESSFUL RESPONDENT'S OBLIGATIONS WILL BE MET BY FULFILLMENT OF THE SUBCONTRACTOR COMPLIANCE OBLIGATIONS. AFTER COMPLIANCE WITH THE SAME, SUCCESSFUL RESPONDENT SHALL HAVE NO LIABILITY TO ANYONE UNDER THIS PROVISION WITH RESPECT TO ITS SUBCONTRACTORS.

## **14.11** Independent Contractor Status

Successful Respondent, in furnishing Services hereunder, is acting as an independent contractor, and Successful Respondent has the sole obligation to supervise, manage, contract, direct, procure, perform, or cause to be performed, all work to be performed by Successful Respondent under this CTSA. The relationship of the Parties under this CTSA shall not constitute a partnership or joint venture for any purpose. Successful Respondent is not an agent of DIR or the Customers and has no right, power, or authority, expressly or impliedly, to represent or bind DIR or any Customer as to any matters.

Vandor	Contract No.	
vendor	Contract No.	

### 14.12 No Additional Consideration

Successful Respondent, its agents, its Subcontractors, and Successful Respondent Personnel will not be entitled to nor receive from DIR any additional consideration, compensation, salary, wages, or any other type of remuneration for Services rendered under this CTSA, except as set forth in Exhibit B. Specifically, Successful Respondent will not be entitled by virtue of this CTSA to any consideration in the form of overtime, health insurance benefits, retirement benefits, disability retirement benefits, sick leave, vacation time, paid holidays, or other paid leaves of absence of any type or kind whatsoever. In addition, all costs associated with transportation, delivery, and insurance relating to the Successful Respondent's, agents', Subcontractors', or Successful Respondent Personnel's performance of this CTSA will be paid for by the Successful Respondent.

## 14.13 Prompt Payment Act

When payments by DIR or Customers are necessary under this CTSA, or a CSA, payment by Government Entities will be made in accordance with the Texas Prompt Payment law, Chapter 2251, Texas Government Code. Non-governmental entities shall follow Chapter 2251, Texas Government Code, in making payments due hereunder.

## **14.14** Payments for Services

During the term of this CTSA, Successful Respondent will receive funds in consideration for the Services, calculated in accordance with Exhibit B. Payment for Disentanglement Services will be in accordance with a written agreement between the Parties.

## 14.15 DIR Administrative Fee and Direct Sales Transaction Services Reporting

#### 14.15.1 DIR Administrative Fee

DIR's TEX-AN 2021 Program is a cost recovery program. The Successful Respondent shall pay an Administrative Fee to DIR to defray the DIR costs of operating and administering the TEX-AN 2021 Program.

- a) DIR, in its discretion with DIR Board approval, formulates the Administrative Fee per Service designed to fulfill its program needs, as authorized in Chapter 2170, Texas Government Code. Such Administrative Fee may change over time, and DIR may change the amount of the Administrative Fee upon thirty (30) calendar days written notice to Successful Respondent without the need for a formal CTSA amendment.
- b) With the exception of Direct Sales Transaction Services, DIR shall bill for such Administrative Fee and collect same from the Customers.
- c) Successful Respondent shall include the applicable Direct Sales Transaction Service Administrative Fee, as a component of its Price, in its bills to the Direct Sales Transaction Service Customers and shall remit the Administrative Fee to DIR on a monthly basis. See Section 14.15.2(b), Reporting of Direct Sales Transactions and Payment of Administrative Fees, below.
- d) Successful Respondent consents to DIR retaining the Administrative Fee portion of receipts received from Direct Sales Transaction Service Customers and consents to receiving the net proceeds as payment in full for its non-Direct Sales Transaction Service sold hereunder. Such payment shall be made as long as the net amount received by Successful Respondent is equal to the undisputed billed amount.

Vandor	Contract No.	
vendor	Contract No.	

#### 14.15.2 Direct Sales Transaction Service Reporting

- a) Successful Respondent shall be responsible for reporting all Direct Sales Transaction Services purchased under the CTSA. Successful Respondent shall file the monthly reports and Subcontract reports in accordance with the due dates specified in this Section 14.15.
- b) Successful Respondent shall electronically provide DIR with a detailed monthly report in the format required by DIR showing the dollar volume of any and all sales under the CTSA for the previous calendar month period. Reports are due on the fifteenth (15th) calendar day of the month following the month of the sale. If the fifteenth (15th) calendar day falls on a weekend or State or federal holiday, the report shall be due on the next Business Day. The monthly report shall include, per transaction: the detailed sales for the period, Customer name, Invoice date, Invoice number, description, quantity, MSRP or List Price, unit price, extended price, Customer Purchase Order number, contact name, Customer's complete billing address, the estimated Administrative Fee for the reporting period, Subcontractor name, the electronic product environmental assessment Tool (EPEAT) designation (if applicable), configuration (if applicable), CTSA discount percentage, actual discount percentage, negotiated CTSA price (if fixed price is offered instead of discount off of MSRP), and other information as required by DIR. Each report must contain all information listed above per transaction or the report will be rejected and returned to the Successful Respondent for correction in accordance with this section. Successful Respondent shall report in a manner required by DIR which is subject to change dependent upon DIR's business needs. Failure to do so may result in CTSA termination.
- c) DIR shall have the right to verify required reports and to take any actions necessary to enforce its rights under this Section, including but not limited to, audits and investigations in accordance with terms and conditions of Article 13.
- d) DIR will review monthly vendor sales reports (VSR), close the sales period, and notify the Successful Respondent of the Administrative Fee due to DIR. The monthly VSR is due the fifteenth (15th) of the month following the reporting month, or the next business day if the 15th falls on a weekend or State holiday. DIR will send notification of the final Administrative Fee and any late fees by the fourteenth (14th) calendar day in the second month following the reporting month. Successful Respondent shall pay the total fee by the twenty-fifth (25th) calendar day in the second month following the reporting month. For example, vendor sales reports for invoiced sales during the month of January is due by February 15. DIR reviews and closes January sales reports and notifies Successful Respondent of final Administrative Fees by March 14th and Successful Respondent must submit payment by March 25th.
- e) Successful Respondent shall reference the DIR Contract number, reporting period, and Administrative Fee amount on any remittance instruments.
- f) Successful Respondent shall correct any inaccurate reports or Administrative Fee payments and deliver such corrected reports and Administrative Fee payments within three (3) Business Days upon written notification by DIR. If Successful Respondent is unable to correct inaccurate reports or Administrative Fee payments within three (3) Business Days or delivers late reports or Administrative Fee payments, Successful Respondent must contact DIR and provide a Corrective Action Plan, including the timeline for completion of fee payments correction. The Corrective Action Plan shall be subject to DIR approval and discretion.
- g) Should Successful Respondent fail to correct inaccurate reports or cure the delay in timely delivery of reports and payments within the Corrective Action Plan timeline, DIR reserves the right to require an independent third party audit of the Successful Respondent's records as specified in CTSA Section 13.5, at Successful Respondent's expense.

Vandor	Contract No.	
venaor	Contract No.	

### 15 DISCLOSURE AND CONFIDENTIALITY OF INFORMATION

## **15.1** Confidentiality

- a) Confidential Information.
  - Successful Respondent and DIR each acknowledge that the other possesses and shall continue to possess information that has been developed or received by it, has commercial, proprietary, or other value in its or its constituents' or customers' activities or operations, and is not generally available to the public, subject, however to the applicability of the Texas Public Information Act (Chapter 552, Texas Government Code) and other applicable law. Except as otherwise specifically agreed in writing by the Parties, "Confidential Information" means:
    - i. all information marked confidential, restricted, or proprietary by either Party and
    - ii. any other information that is treated as confidential by the disclosing Party and would reasonably be understood to be confidential, whether or not so marked.
  - In the case of DIR and the Customers, Confidential Information also shall include developed materials, DIR or Customer Data, Customer information, attorney-client privileged materials, attorney work product, research information, information that contains trade secrets, human resources and personnel information, or other information or data obtained, received, transmitted, processed, stored, archived, or maintained by Successful Respondent under this CTSA. Successful Respondent's Confidential Information shall not include Confidential Information of DIR or Customers, and provided further, however characterization of information as Confidential Information of Successful Respondent shall not limit or restrict the rights of DIR or Customers to exercise their rights (including rights related to auditing and benchmarking) provided for under this CTSA.
  - 3) Successful Respondent shall cause its agents and Subcontractors to treat all information that is obtained through performance of the Services under this CTSA, including information relating to DIR and Customers as Confidential Information to the extent that confidential treatment is provided under State and federal law, regulations, or administrative rules.
- b) Successful Respondent is responsible for understanding and causing its agents and Subcontractors to understand the degree to which information obtained through performance of this CTSA is confidential under State and federal law, regulations, or administrative rules.
- c) Successful Respondent shall not and shall cause its agents and Subcontractors to not use or access any information obtained through performance of this CTSA in any manner except as is necessary for the fulfillment of requested Services under this CTSA.
- d) Successful Respondent shall securely store DIR's and Customer's Confidential Information until such Confidential Information is returned or destroyed in accordance with this section. Successful Respondent must have systems in effect to protect all records and all other documents deemed Confidential Information that are maintained in connection with the Services. Any disclosure or transfer of Confidential Information by Successful Respondent, including information required by DIR, will be in accordance with applicable State and federal law, regulations, or administrative rules. If the Successful Respondent receives a request for information deemed to be Confidential Information, the Successful Respondent shall immediately notify DIR of such request, and shall make best efforts to protect such information from public disclosure.

- e) In addition to the requirements expressly stated in this Section 15.1, Successful Respondent must comply with any policy, rule, or requirement of DIR that relates to the safeguarding or disclosure of information relating to the TEX-AN 2021 Program recipients, Successful Respondent's operations, or Successful Respondent's performance of this CTSA.
- f) Within thirty (30) days of the expiration or termination of this CTSA for any reason, all Confidential Information of a Party, disclosed to another Party, including all copies made by the other Party, will be returned to the disclosing Party or, and to the extent authorized by applicable record retention laws and policies, at the disclosing Party's option, erased or destroyed. The recipient of the Confidential Information will provide the disclosing Party certificates evidencing such erasure or destruction. Notwithstanding anything in this CTSA to the contrary, the Successful Respondent may retain, for the duration of the seven (7) year audit period, safeguarded (and encrypted if Sensitive Personal Information as defined under the laws of the State is included) copies of the following records and Data it receives or prepares during or in connection with the TEX-AN 2021 Program and CTSA, to comply with the record keeping requirements of Section 13.1 and for archival and defense purposes:
  - Supporting financial information and documents that are adequate to ensure CTSA
    compliance and are sufficient to ensure the accuracy and validity of Successful Respondent
    Invoices;
  - 2) Financial information, records and reports;
  - 3) Transaction logs;
  - 4) Successful Respondent's customer database data files;
  - 5) Successful Respondent human resources documentation;
  - 6) Audit reports:
  - 7) Operational reporting (e.g., security logs, security metrics reporting/audits, and relevant network reports/logs); and
  - 8) Data for the TEX-AN 2021 Program and CTSA and its applications, including maintenance log files, uptime log files, relevant network access log files, and other required analytics.
- g) With the exception of Confidential Information of Customers, Confidential Information of a Party will not be afforded the protection of this CTSA if such data:
  - 1) is, at the time of disclosure, generally available to the public other than through a breach of the receiving Party's or a third party's confidentiality obligations;
  - 2) after disclosure, is published by the disclosing Party or otherwise becomes generally available to the public other than through a breach of the receiving Party's or a third party's confidentiality obligations;
  - 3) was lawfully in the possession of the receiving Party immediately prior to the time of disclosure to it;
  - 4) is received from a third party having a lawful right to disclose such information; or
  - 5) is independently developed by the receiving Party without reference to the disclosing Party's Confidential Information. The exclusions in this section shall not apply to sensitive or regulated Data such as personal health information and personal identifying information
- h) DIR will require its agents, Subcontractors, and Successful Respondent Personnel to comply with the terms of this Section 15.1 as it relates to Successful Respondent Confidential Information. Prior to disclosing Successful Respondent Confidential Information to other Contract Holders, DIR will obtain an executed confidentiality agreement in a form acceptable to Successful Respondent, whose consent to use will not be unreasonably withheld or delayed.

Vendor	Contract No.		
vendor	COHITACI INO.		

### 15.2 Disclosure or Loss of DIR's and Customers' Confidential Information

- a) Successful Respondent shall immediately report to DIR and the affected Customer any and all unauthorized disclosures, exposures, uses, or losses of DIR's and/or the Customer's Confidential Information, including Security Incidents, of which it or Successful Respondent Personnel is aware or has knowledge, including the Successful Respondent's reasonable belief that the Confidential Information may have been disclosed or exposed in an unauthorized manner. Successful Respondent acknowledges that any publication, exposure, loss, or disclosure of DIR's and/or Customer's Confidential Information may cause immediate and irreparable harm to DIR and /or the Customer and may constitute a violation of State or federal laws.
- b) In the event Successful Respondent discovers or is notified of a breach, potential breach, exposure, loss, or inappropriate disclosure of Confidential Information, including Security Incidents, Successful Respondent shall, in addition to its other obligations in this CTSA, immediately once Successful Respondent knows or should have known such a breach, potential breach, exposure, loss, or inappropriate disclosure has occurred:
  - 1) notify DIR of such breach, potential breach, exposure, loss, or inappropriate disclosure;
  - 2) investigate (with DIR's participation if so desired by DIR) such breach, potential breach, exposure, loss, or inappropriate disclosure and perform a risk assessment, root cause analysis, and Corrective Action Plan thereon;
  - provide a written report to DIR of such rick assessment, root cause analysis, and Corrective Action Plan;
  - 4) remediate the effects of such breach, potential breach, exposure, loss, or inappropriate disclosure as soon as practicable or coordinate such remediation if Successful Respondent does not have responsibility for the matters which are the source of the breach, potential breach, exposure, loss or inappropriate disclosure; and
  - 5) provide DIR with reasonable assurances that such breach, potential breach, exposure, loss, or inappropriate disclosure shall not recur.
- If Successful Respondent or Successful Respondent Personnel should publish, expose, lose, or disclose such Confidential Information without authorization, DIR and/or the affected Customer will immediately be entitled to seek injunctive relief or any other remedies to which it is, or they are, entitled under law or equity without requiring a cure period as described in Article 17. DIR and/or Customers will have the right to recover from Successful Respondent all damages and liabilities caused by or arising from Successful Respondent's, its agent(s) and/or its Subcontractors', failure to protect DIR's and Customers' Confidential Information, including but not limited to the cost of providing notice to affected individuals, the cost of providing thirty-six (36) months of credit-monitoring services, the cost of creating a call center to support affected individuals for thirty (30) days past notification, any related governmental fees or fines assessed against DIR or Customers, any damages assessed by a court of competent jurisdiction against DIR or Customer(s) related to such event, and any other losses related to such event. Successful Respondent will defend with counsel approved by DIR and/or the affected Customer, indemnify and hold harmless DIR and the affected Customer from all damages, costs, liabilities, and expenses as described above (including without limitation reasonable attorneys' fees and costs) caused by or arising from Successful Respondent Personnel's failure to protect DIR's and/or Customers' Confidential Information. Defense and settlement matters shall be coordinated by the Texas Office of the Attorney General for State agency Customers and by counsel of Customer's choice for non-State agency Customers.

Vendor Contract No.	
---------------------	--

d) Successful Respondent will require its agent(s) and/or its Subcontractor(s), to comply with the terms of this Article 15.

## **15.3** Public Information

- a) Successful Respondent acknowledges that DIR and some of the Customers are subject to the Texas Public Information Act (Chapter 552, Texas Government Code). Notwithstanding the provisions of this article, Successful Respondent also acknowledges that DIR will comply with the Texas Public Information Act, and with all opinions of the Texas Office of the Attorney General concerning the Public Information Act. DIR agrees that it will promptly notify Successful Respondent of each request for disclosure of public information filed in accordance with the Texas Public Information Act that requests any of the Successful Respondent's Confidential Information, including data to which Successful Respondent has a proprietary or commercial interest. DIR will deliver all copies of such requests for public information to Successful Respondent in accordance with Chapter 552, Texas Government Code.
- b) With respect to any information that is subject to the Texas Public Information Act, Successful Respondent is solely responsible for demonstrating to the Texas Office of the Attorney General, the specific reasons why the requested information is confidential or otherwise exempted from required public disclosure under the Texas Public Information Act. Successful Respondent will provide DIR with copies of all such communications.
- c) Under the terms of the CTSA, DIR may provide Successful Respondent with information related to Customers and/or the TEX-AN 2021 Program. Successful Respondent will not re-sell or otherwise distribute or release any such information to any party in any manner without DIR's or the Customer's, as appropriate, express written consent (which consent shall be in their sole and absolute discretion), unless such disclosure is required by law or pursuant to Section 15.1(g) herein.

### 15.4 Unauthorized Acts

## Each Party agrees to:

- Notify the other Party promptly of any unauthorized exposure, disclosure, possession, use, or knowledge, or any unauthorized attempt to possess, use, or gain knowledge, of any Confidential Information by any person or entity that may become known to it;
- b) Promptly furnish to the other Party full details of the unauthorized exposure, disclosure, possession, use, or knowledge, or any unauthorized attempt to possess, use or gain knowledge, and use reasonable efforts to assist the other Party in investigating or preventing the recurrence of any unauthorized possession, use, or knowledge, or attempt thereof, of Confidential Information. For DIR the reasonable efforts to assist Successful Respondent shall be subject to and limited by the Texas Public Information Act and any other applicable laws; and
- c) Cooperate with the other Party in any litigation and investigation against third parties deemed necessary by such Party to protect its proprietary rights. For DIR the reasonable efforts to assist the Successful Respondent shall be subject to and limited by the Texas Public Information Act and any other applicable laws.

### 15.5 Legal Action

Successful Respondent may not commence any legal action or proceeding in respect to any unauthorized possession, use, or knowledge, or any unauthorized attempt to possess, use, or gain knowledge, of

Vendor Contract No.	
---------------------	--

Confidential Information by any person or entity, which action or proceeding identifies DIR and/or the Customers or their Confidential Information without consent from DIR and/or Customers, which consent shall be in their sole and absolute discretion.

## 16 LIABILITY

## **16.1** Property Damage

- a) Successful Respondent will protect DIR's and Customers' real and personal property from damage arising from Successful Respondent's, Successful Respondent Personnel's, and its Subcontractors' performance of this CTSA, and Successful Respondent will be responsible for any loss, destruction, or damage to DIR's and Customers' property that results from or is caused by Successful Respondent's, Successful Respondent Personnel's, or its Subcontractors' negligent or wrongful acts or omissions. Upon the loss of, destruction of, or damage to any property of DIR and Customers, Successful Respondent will notify DIR and the applicable Customer promptly and, subject to direction from DIR and the applicable Customer, will take all steps to protect that property from further damage.
- b) Successful Respondent agrees to observe and require Successful Respondent Personnel and Subcontractors to observe no less than industry standard safety measures and proper operating procedures at DIR's and Customers' sites at all times and otherwise to comply with the procedures and safety standards imposed by DIR and Customers, whichever is the more rigorous standard.
- c) Successful Respondent will distribute a policy statement to all of Successful Respondent Personnel and Subcontractors that directs the Successful Respondent Personnel or Subcontractor to immediately report to DIR and the applicable Customer and Successful Respondent the existence of any special defect or unsafe condition encountered while on DIR's or Customers' premises. Successful Respondent shall immediately report to DIR and the applicable Customer the existence of any special defect or an unsafe condition it encounters or otherwise learns about.

## 16.2 Risk of Loss

As applicable to any Equipment used in the performance of Services by Successful Respondent, during the period such Equipment is in possession of Successful Respondent, its carriers, or DIR or Customers, prior to being Accepted by DIR and/or the applicable Customer, Successful Respondent will insure and bear the risk of loss or damage thereto, unless such loss or damage is caused by the negligence or intentional misconduct of DIR or the applicable Customer.

## **16.3** Limitation of Liability

- a) For any claim or cause of action arising under or related to the CTSA, neither Party will be liable to the other Party for indirect, punitive, special, or consequential damages, even if it is advised of the possibility of such damages.
- b) Items Not Considered as Damages. The following will not be considered damages subject to, and will not be counted toward the liability exclusion or cap specified herein:
  - 1) SLA Credits assessed against Successful Respondent pursuant to this CTSA.

Vendor Contract No.	
---------------------	--

- 2) Amounts withheld by DIR in accordance with this CTSA due to incorrect charges or Services not provided.
- 3) Amounts paid by DIR but subsequently recovered from Successful Respondent due to incorrect charges or services not provided.
- Successful Respondent Invoiced Rates to DIR, Prices, telecommunications fees and surcharges and other amounts that are due and owing to Successful Respondent for Services under this CTSA.
- c) Duty to Mitigate. Each Party shall use appropriate efforts to mitigate its damages to the extent within its reasonable control and consistent with the Parties' respective performance obligations under this Agreement; provided, however, this provision is not intended to expand or diminish a Party's rights or obligations under this Agreement, alter the plain meaning of the provisions contained herein, or limit a Party's rights to act in its own self-interest.

## 17 REMEDIES AND DISPUTES

## 17.1 Understanding and Expectations

DIR can choose any and all remedies including all others afforded by law at its own discretion.

### 17.2 Tailored Remedies

#### 17.2.1 Understanding of the Parties

Successful Respondent agrees and understands that DIR may pursue tailored contractual remedies for noncompliance with this CTSA. At any time and at its sole and absolute discretion, DIR may impose or pursue one or more remedies for each item of noncompliance and will determine remedies on a case-by-case basis. DIR's pursuit or non-pursuit of a tailored remedy does not constitute a waiver of any other remedy that DIR may have at law or equity, including, but not limited to, the remedies set forth in Section 17.3 below. Prior to imposing any remedies, DIR will provide notice to the Successful Respondent of the non-compliance, and such notice shall include a reasonable cure period in accordance with Section 17.2.2.2.5) ii, or as specified elsewhere in this CTSA. In all instances under which DIR may pursue multiple remedies, DIR has the option, in its sole and absolute discretion, to collect either Credits or actual damages.

### 17.2.2 Notice and Opportunity to Cure for Deficiencies

- a) In the event that DIR reasonably determines that Successful Respondent has failed or is reasonably likely to fail to meet performance expectations, standards, SLAs, or schedules (a Deficiency), but that, in the determination of DIR, in its sole and absolute discretion, do not result in a Material Breach or delay in the delivery or operation of the Services, DIR shall notify Successful Respondent in writing. If Successful Respondent has determined that it has failed or is reasonably likely to fail to meet performance expectations, standards, SLAs, or schedules, Successful Respondent shall notify DIR.
- b) Upon notice or discovery, Successful Respondent will take immediate steps to mitigate any harmful effects of such Deficiency and promptly perform a root cause analysis. Successful Respondent will, within ten (10) calendar days (or another date approved in writing by DIR) of receipt of written notice of a Deficiency from DIR or delivery of such notification to DIR, provide DIR a written response that:
  - 1) Explains the reasons for the Deficiency, Successful Respondent's plan to address or cure the Deficiency, and the date and time by which the Deficiency will be cured; or

- 2) If Successful Respondent disagrees with DIR's findings, its reasons for disagreeing with DIR's findings.
- c) Successful Respondent's proposed cure of a Deficiency is subject to the approval of DIR, in its sole and absolute discretion. Successful Respondent's repeated commission of Deficiencies or failure to resolve any Deficiencies may be regarded by DIR as a Material Breach and entitle DIR to pursue any other remedy provided in this CTSA or any other appropriate remedy DIR may have at law or equity. At any time, DIR may, in its own discretion and based on the passage of time and information discovered by DIR or provided by Successful Respondent, determine that a Deficiency is a Material Breach. In that instance, DIR will notify Successful Respondent in accordance with the below requirements.

## 17.2.2.1 Notice and Opportunity to Cure for Material Breach

- a) DIR will notify Successful Respondent in writing of specific areas of Successful Respondent performance that fail to meet performance expectations, standards, SLAs, or schedules, and that, in the determination of DIR, in its sole and absolute discretion, do constitute a Material Breach in the delivery or operation of the Services or a Material Breach of the CTSA. Successful Respondent may also commit a Material Breach if it attempts or threatens to commit a Material Breach
- b) Upon notice or discovery of a Material Breach, Successful Respondent will take immediate steps to mitigate any harmful effects of such Material Breach and promptly perform a root cause analysis. Successful Respondent will, within ten (10) calendar days (or another date approved by DIR) of receipt of written notice of a Material Breach, provide DIR a detailed written Corrective Action Plan to correct or resolve a Material Breach of this CTSA.
- c) The Corrective Action Plan must provide:
  - 1) A detailed explanation of the reasons for the cited Material Breach as determined by the root cause analysis;
  - 2) A specific proposal to cure or resolve the Material Breach promptly and completely.
- d) The Corrective Action Plan is subject to approval or modification by DIR, in its sole and absolute discretion.
- e) DIR will notify Successful Respondent in writing of DIR's final disposition of the Corrective Action Plan. If DIR accepts Successful Respondent's proposed Corrective Action Plan, DIR may:
  - 1) Condition such approval on completion of tasks in the order or priority that DIR may prescribe;
  - 2) Disapprove portions of Successful Respondent's proposed Corrective Action Plan; or
  - 3) Require additional or different Corrective Action Plan(s).
- f) At any time during this process, DIR reserves the right, in its sole and absolute discretion, to:
  - 1) Suspend all, or part of, this CTSA; or
  - 2) Issue a written Stop Marketing Order which prohibits Successful Respondent or any of its agents, or Subcontractors, third party providers or from further marketing the Services during investigation of the alleged Material Breach and pending corrective action, if necessary, by Successful Respondent, or a decision by DIR to terminate the CTSA for Cause. DIR may delay the implementation of the Stop Marketing Order if it

affects the completion of any of the Services in accordance with a Customer's approved schedule under a CSA.

- g) DIR's acceptance of a Corrective Action Plan under this Section 17.2 will not:
  - 1) Excuse Successful Respondent's prior Material Breach(es);
  - 2) Relieve Successful Respondent of its duty to comply with performance standards; or
  - 3) Prohibit DIR from assessing additional tailored remedies or pursuing other appropriate remedies for continued Deficiencies or Material Breaches.

#### 17.2.2.2 Administrative Remedies

- a) At its sole and absolute discretion, DIR may impose one or more of the following remedies for each item of non-compliance, both Deficiencies and Material Breaches, and will determine the scope and severity of the remedy on a case-by-case basis:
  - 1) Assess Credits in accordance with the terms of this CTSA;
  - 2) Conduct accelerated monitoring of Successful Respondent including more frequent or more extensive monitoring by DIR or its agents;
  - 3) Require additional, more detailed, marketing, and/or performance reports to be submitted by Successful Respondent;
  - 4) Decline to extend this CTSA; and/or
  - 5) Terminate this CTSA, in its entirety or in part, in accordance with Section 17.3, provided, however, that the right to terminate hereunder shall be limited to those instances where:
    - i. The non-compliance constitutes a Material Breach;
    - ii. DIR gives Successful Respondent written notice of such Material Breach and a thirty (30) calendar day cure period, or other cure period that DIR in its sole discretion agrees to in writing; and
    - iii. Successful Respondent does not cure the Material Breach within the applicable cure period.
- b) For purposes of this CTSA, an item of non-compliance, both Deficiencies and Material Breaches, means a specific action of Successful Respondent that:
  - 1) Violates a provision of this CTSA and/or a TEX-AN 2021 Customer Service Agreement;
  - 2) Fails to meet an agreed SLA; or
  - 3) Represents a failure of Successful Respondent to be responsive to a request of DIR and/or a Customer relating to the Services for information, assistance, or support within the timeframe specified by DIR and/or the Customer provided that such request for information, assistance, or support is within the Services specified to be provided in this CTSA and/or CSA, as amended from time to time.
- c) DIR will provide notice to Successful Respondent of the imposition of an administrative remedy in accordance with this Section 17.2, with the exception of accelerated monitoring, which may be unannounced. DIR may require Successful Respondent to file a written response in accordance with this Section.

37 1 0		
venaor C	ontract No.	

### 17.2.2.3 Damages

- a) DIR will be entitled to actual damages resulting from the Successful Respondent's failure to comply with any of the terms of this CTSA. In some cases, the actual damage to DIR or the Customers as a result of Successful Respondent's failure to meet any provision of, or obligation under, this CTSA and/or to meet specific SLAs or performance standards set forth in this CTSA are difficult or impossible to determine with precise accuracy. Therefore, agreed Credits may be assessed in writing against and in such event will be paid by Successful Respondent for failure to meet any aspect of Successful Respondent's responsibilities under this CTSA and/or to meet the specific performance standards identified in Exhibit C Performance Management. Credits will be assessed if DIR determines, in its sole and absolute discretion, such failure is the fault of Successful Respondent (including the Successful Respondent Personnel) and is not materially caused or contributed to by DIR, Customers, or their agents. If at any time, DIR determines, in its sole and absolute discretion, that Successful Respondent has not met a provision of, or has not fulfilled an obligation under this CTSA and/or the specific performance standards due to mitigating circumstances, DIR reserves the right to waive all or part of the Credits. All such waivers must be in writing, contain the reasons for the waiver, and must be signed by the appropriate executive of DIR.
- b) The Credits prescribed in this Section 17.2 are not intended to be in the nature of a penalty, but are intended to be reasonable estimates of DIR's projected financial loss and damage resulting from the Successful Respondent's nonperformance, including financial loss as a result of Service delivery delays to DIR and/or the Customers. Accordingly, in the event Successful Respondent fails to perform in accordance with this CTSA, DIR may assess Credits as provided in this Section 17.2.
- c) If Successful Respondent fails to perform any of the Services as ordered by DIR or the Customers, DIR may assess Credits for each occurrence of such failure, to the extent consistent with DIR's tailored approach to remedies and State law.
- d) In accordance with Exhibit C Performance Management, DIR may elect, in its sole and absolute discretion, to collect Credits:
  - 1) Through direct assessment and demand for payment; or
  - 2) By deduction of amounts assessed as Credits as set-off against payments then due to Successful Respondent for the Services or that become due at any time after assessment of the Credits. DIR will make deductions until the full amount payable by the Successful Respondent is received by DIR or Customer.
- e) Credits and actual damages imposed under any provision of this CTSA will be limited to the amounts and triggering events specified in this CTSA.
- f) Acknowledgment of Actual Damages
  - The absence of actual damages listed in this section shall not be construed or interpreted as an agreement to exclude it as an actual damage under this CTSA. For the avoidance of doubt, the following shall be considered actual damages and Successful Respondent shall not assert that they are indirect, incidental, collateral, consequential, or special damages, or lost profits to the extent they result directly from the Successful Respondent's failure to perform in accordance with this CTSA:
  - 1) Costs and expenses of restoring or reloading any lost, stolen, or damaged Data;
  - 2) Costs and expenses of implementing a work-around in respect of a failure to provide the Services or any part thereof;

Vendor Contract No.	
---------------------	--

- 3) Costs and expenses of replacing lost, stolen, or damaged Equipment and materials;
- 4) Cover damages, including the costs and expenses incurred to procure the Services or corrected Services from an alternate source;
- 5) Costs and expenses incurred to bring the Services in-house or to contract to obtain the Services from an alternate source;
- 6) Straight time, overtime, or related expenses incurred by DIR or Customer in performing 1) through 5) above, including overhead allocations for employees, wages, and salaries of additional employees, travel expenses, overtime expenses, telecommunication charges, and similar charges;
- 7) Fines, penalties, sanctions, interest, or other monetary remedies incurred as a result of a failure to comply with applicable laws;
- 8) Credits assessed against Successful Respondent in accordance with this CTSA; and
- 9) Costs and expenses of protecting and compensating the State and its constituents after a Security Incident, including but not limited to notifications, fines and penalties, damages established by a court of law, costs to establish a call center, and thirty-six (36) months of credit monitoring for effected individuals.
- g) Liquidated Damages [Requirements regarding Liquidated Damages will be negotiated per contract.]

### 17.2.2.4 Equitable Remedies

- a) Successful Respondent acknowledges that, if Successful Respondent commits a Material Breach (or attempts or threatens to commit a Material Breach) of its obligations under this CTSA, DIR, the Customers, and the State will be irreparably harmed. In such a circumstance, DIR, for and on behalf of its Customers and the State, may proceed directly to court.
- b) If a court of competent jurisdiction finds that Successful Respondent has committed an uncured Material Breach (or attempted or threatened to commit a Material Breach), Successful Respondent agrees that, without any additional findings of irreparable injury or other conditions to injunctive relief, it will not oppose the entry of an appropriate injunctive order compelling performance by Successful Respondent and restraining it from committing any further Material Breaches (or attempted or threatened Material Breaches).

# 17.2.2.5 Suspension of CTSA

- a) DIR may suspend performance of all or any part of this CTSA if:
  - DIR determines that Successful Respondent has committed a Material Breach of this CTSA and, after written notice from DIR, such Material Breach is not cured by Successful Respondent during the cure period as further described in Section 17.2.2.2 5) ii of the CTSA; provided however, if no specific cure period is specified under this CTSA, such cure period will be not less than ten (10) calendar days.
  - 2) DIR has reason to believe that Successful Respondent, Successful Respondent Personnel, and/or Successful Respondent's Subcontractors, have committed, assisted in the commission of, or failed to take appropriate action concerning fraud, malfeasance, or misfeasance concerning this CTSA and if DIR, in its sole and

absolute discretion, believes that the suspension is necessary to prevent a material loss or material damages to the State, and the portion of the CTSA being suspended has a direct relationship to the alleged fraud, malfeasance, or misfeasance concerning this CTSA.

- 3) DIR determines that suspension of this CTSA in whole or in part is convenient or in the best interests of the State or the TEX-AN 2021 Program.
- b) DIR will notify Successful Respondent in writing of its intention to suspend this CTSA in whole or in part. Such notice will:
  - 1) Be delivered in writing to Successful Respondent; and
  - 2) Include a concise description of the facts or matter leading to DIR's decision.

#### 17.3 Termination of CTSA

In addition to other provisions of this Article 17 allowing termination, this CTSA will terminate upon the Expiration Date unless extended in accordance with the terms of this CTSA, or terminated sooner under the terms of this CTSA. Prior to completion of the Term, all or a part of this CTSA may be terminated for any of the following reasons:

## 17.3.1 Termination by Mutual Agreement of the Parties.

This CTSA may be terminated by mutual agreement of the Parties. Such agreement must be in writing.

#### 17.3.2 Termination in the Best Interest of the State.

DIR may terminate this CTSA, in whole or in part, without penalty to or other liability on DIR, at any time when, in its sole discretion, DIR determines that termination is in the best interests of the State. The termination will be effective on the date specified in DIR's notice of termination.

### 17.3.3 Termination for Cause

Subject to the applicable notice and cure provisions in Section 17.2, DIR reserves the right to terminate this CTSA, in whole or in part, without recourse, upon the following conditions (each a Cause):

### 17.3.3.1 Successful Respondent Insolvency

Subject to relevant law, DIR may terminate this CTSA if Successful Respondent:

- a) makes an assignment for the benefit of all or substantially all of its creditors;
- b) admits in writing its inability to pay its debts generally as they become due, including passing a resolution for voluntary liquidation;
- c) consents to or is subject to the appointment of a receiver, trustee, manager, or liquidator over all or any part of its property;
- d) becomes or is declared insolvent or is the subject of any bona fide proceedings related to its liquidation, administration, provisional liquidation, insolvency, or the appointment of a receiver or similar officer for it; or
- e) enters into an agreement or arrangement for the composition, extension, or readjustment of substantially all of its obligations or any class of such obligations.

## 17.3.3.2 Failure to Adhere to Laws, Rules, Ordinances, or Orders

a) DIR may terminate this CTSA upon making a determination, in its sole and absolute discretion, that Successful Respondent failed to adhere to any laws, ordinances, rules,

regulations, or orders of any public authority having jurisdiction and such violation prevents or substantially impairs performance of Successful Respondent's duties under this CTSA.

b) DIR may terminate this CTSA if a court of competent jurisdiction finds Successful Respondent failed to adhere to any laws, ordinances, rules, regulations, or orders of any public authority having jurisdiction and such violation prevents or substantially impairs performance of Successful Respondent's duties under this CTSA.

### 17.3.3.3 Breach of Confidentiality

DIR may terminate this CTSA if DIR determines, in its sole and absolute discretion, that Successful Respondent breached Article 15 or any confidentiality or privacy laws with respect to the Services provided under this CTSA and related Data.

## 17.3.3.4 Failure to Maintain Adequate Personnel or Resources

DIR may terminate this CTSA if, after providing notice and an opportunity to cure, DIR determines, in its sole and absolute discretion, that Successful Respondent has failed to supply sufficient personnel or resources and such failure results in Successful Respondent's inability to fulfill its duties to provide Services in any way under this CTSA.

#### 17.3.3.5 Termination for Gifts and Gratuities

- a) DIR may terminate this CTSA following its determination that Successful Respondent, Successful Respondent Personnel, or Successful Respondent's Subcontractors have either offered or given any economic opportunity, future employment, gift, loan, gratuity, special discount, trip, favor, or service (collectively "Thing of Value") to an officer or employee of DIR, a Customer, or the State in violation of State law. DIR may terminate this CTSA following a determination by a competent judicial or quasi-judicial authority and Successful Respondent's exhaustion of all legal remedies that Successful Respondent, Successful Respondent Personnel, or Successful Respondent's Subcontractors have either offered or given a Thing of Value to an officer or employee of DIR, a Customer, or the State in violation of State law.
- b) Termination of a Subcontract by Successful Respondent pursuant to this provision will not be a cause for termination of this CTSA unless:
  - 1) Successful Respondent fails to replace such terminated Subcontractor within a reasonable time; and
  - 2) Such Failure constitutes Cause as described in this Section 16.3.

## 17.3.3.6 Material Adverse Change in Successful Respondent's Financial Condition

If (1) Successful Respondent receives a "going concern" qualification from its external auditor or (2) Standard & Poor's lowers Successful Respondent's long term credit rating to lower than BB-and in the reasonable opinion of DIR such change in the financial condition of Successful Respondent may impair or otherwise compromise the ability of Successful Respondent to perform its obligations under this Agreement, then DIR may, in its sole discretion, terminate this Agreement by giving Successful Respondent at least thirty (30) calendar days prior notice. With respect to the events described in (2) above, prior to exercising its right to terminate, DIR shall meet with Successful Respondent within ten (10) calendar days following notification (or awareness) of such event and permit Successful Respondent to submit to DIR within fifteen (15) calendar days of such meeting a plan that comprehensively addresses DIR's concerns related to Successful Respondent's ability to perform its obligations under this Agreement (the "Service Delivery Plan"). DIR shall not unreasonably withhold its approval of such Service Delivery Plan.

Vendor Contract No.	
---------------------	--

If DIR does not approve the initial Service Delivery Plan it shall inform the Successful Respondent of its reasons and the Successful Respondent shall take those reasons into account in the preparation of a further Service Delivery Plan, which shall be resubmitted to DIR within ten (10) calendar days. If in DIR's reasonable discretion the updated Service Delivery Plan does not address DIR's concerns related to Successful Respondent's ability to continue delivering the Services, then DIR will have the right to terminate this Agreement as described above.

### 17.3.3.7 Termination for Successful Respondent's Material Breach of this CTSA

DIR has the right to terminate this CTSA, in whole or in part, if DIR determines, in its sole and absolute discretion, that Successful Respondent has committed a Material Breach. Prior to exercising its right to terminate all or a part of this CTSA, DIR will provide the Successful Respondent written notice and an opportunity to cure the Material Breach in accordance with Section 17.2, and if such cure is made and accepted by DIR, in its sole and absolute discretion, DIR will not terminate the CTSA.

#### 17.3.3.8 Prohibited Vendor's List

DIR has the absolute right to terminate this CTSA, without recourse, in the event that:

- a) Successful Respondent becomes listed on the prohibited vendors list authorized by Executive Order #13224, "Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism," published by the United States Department of the Treasury, Office of Foreign Assets Control, or
- b) Successful Respondent becomes suspended or debarred from doing business with the federal government as listed in the Excluded Parties List System (EPLS) maintained by the General Services Administration, or
- c) Successful Respondent is found by DIR to be ineligible to hold this CTSA under Section 2155.006(b), Texas Government Code.

## 17.4 DIR Rights Upon Successful Respondent's Bankruptcy

### 17.4.1 General Rights

In the event of Successful Respondent's bankruptcy or other formal procedure or of the filing of any petition under bankruptcy laws affecting the rights of Successful Respondent which is not stayed or dismissed within thirty (30) days of filing, in addition to the other rights and remedies set forth herein, to the maximum extent permitted by law, DIR and Customers shall have the immediate right to retain and take possession for safekeeping all DIR and Customer Data, DIR and Customer Confidential Information, DIR and Customer licensed third party materials, DIR and Customer owned Equipment, DIR and Customer owned Materials, DIR and Customer owned developed materials, and all other materials, Equipment, or systems to which DIR and/or the Customers are or would be entitled during the Term or upon the expiration or any termination of this CTSA. Successful Respondent shall cooperate fully with DIR and the Customers and assist DIR and the Customers in identifying and taking possession of the items listed in the preceding sentence. DIR and Customers shall have the right to hold such Data, Confidential Information, materials, Equipment, and systems until such time as the trustee or receiver in bankruptcy or other appropriate insolvency office holder can provide adequate assurances and evidence to DIR and Customer that they shall be protected from sale, release, inspection, publication, or inclusion in any publicly accessible record, document, material, or filing. Successful Respondent and DIR agree that without this material provision, DIR would not have entered into this CTSA or provided any right to the possession or use of such Data, Confidential Information, or materials covered by this CTSA.

Vendor	Contract No.	
vendor	Contract No.	

### 17.4.2 DIR Rights in Event of Bankruptcy Rejection

Notwithstanding any other provision of this CTSA to the contrary and to the maximum extent permitted by applicable laws, in the event that Successful Respondent becomes a debtor under the United States Bankruptcy Code (11 U.S.C. §101 et. seq. or any similar Law in any other country (the "Bankruptcy Code")) and rejects this Agreement pursuant to Section 365 of the Bankruptcy Code (a "Bankruptcy Rejection"):

- a) any and all of the licensee and sublicensee rights of DIR and the Customers arising under or otherwise set forth in this CTSA, including the rights of DIR and the Customers shall be deemed fully retained by and vested in DIR and the Customers as protected intellectual property rights under Section 365(n)(1)(B) of the Bankruptcy Code and further shall be deemed to exist immediately before the commencement of the bankruptcy case in which Successful Respondent is the debtor;
- b) DIR and the Customers shall have all of the rights afforded to non-debtor licensees and sublicensees under Section 365(n) of the Bankruptcy Code; and
- c) to the extent any rights of DIR and the Customers under this CTSA which arise after the expiration or any termination of this CTSA are determined by a bankruptcy court not to be "intellectual property rights" for purposes of Section 365(n),

all of such rights shall remain vested in and fully retained by DIR and/or the Customers after any bankruptcy rejection as though this CTSA were terminated or expired. DIR shall under no circumstances be required to terminate this CTSA, in whole or in part, after a bankruptcy rejection in order to enjoy or acquire any of its rights under this CTSA, including any of the rights of DIR or the Customers unless and to the extent required by applicable law.

#### 17.5 Effective Date of Termination

Termination will be effective as of the date specified in the notice of termination.

#### 17.6 Extension of Termination Effective Date

Any termination date may be extended upon the mutual agreement of the Parties; however, such date shall not be extended beyond the Expiration Date of this CTSA.

## 17.7 Payment and Other Provisions at CTSA Termination

- a) DIR further agrees to negotiate in good faith with Successful Respondent to equitably adjust and settle any accrued or outstanding liabilities for a Service which has not been Accepted that:
  - 1) Is due or delivered prior to or upon termination of this CTSA; and
  - 2) Benefits DIR, a Customer, or the State, notwithstanding its status as not having been Accepted.
- b) Successful Respondent agrees to negotiate in good faith with DIR and/or Customer to equitably adjust and settle any outstanding disputes currently on file between the parties.
- c) Successful Respondent must provide DIR access to records, facilities, and documentation as is required to efficiently and expeditiously close out the Services under this CTSA.
- d) Successful Respondent must prepare a Disentanglement Plan, upon notice of termination and within ten (10) Business Days, which is acceptable to and approved by DIR, in its sole and

Vendor Contract No.	
---------------------	--

- absolute discretion. Such Disentanglement Plan will be implemented during the time period between DIR's Acceptance of the Disentanglement Plan and the termination date.
- e) Subsections 17.7 a) and 17.7 b) shall only be applicable to time and materials work performed under the CTSA and/or CSAs or as otherwise allowed hereunder.
- f) For all other Rates to DIR and Prices set forth in Exhibit B, DIR or Direct Sales Transaction Service Customers, as appropriate, will pay for all Services provided as of the effective termination date. Article 14 Terms and Conditions of Successful Respondent Billing and DIR and Customer Payment, shall govern the billing and payment for Services under this Section 11.7.

#### 17.8 Amendment of CTSA in the Event of Remedies

DIR may propose an amendment of this CTSA in response to the imposition of a remedy under this Article 17. Any amendments offered pursuant to this Section 17.8 must be limited to the matters causing the exercise of a remedy and in writing. Successful Respondent must negotiate such proposed amendments in good faith.

## **17.9** Disentanglement Assistance

Upon receipt of notice of full or partial termination of this CTSA by DIR, Successful Respondent shall provide all Disentanglement assistance necessary to enable DIR and/or its designee to effectively close out this CTSA and transition the Services and Customers to another TEX-AN 2021 Contract Holder as provided in the Disentanglement Plan (see Article 21. Transition Plan for Expiration).

## 17.10 Successful Respondent Responsibility for Associated Costs

If DIR terminates this CTSA for Cause, the Successful Respondent will be responsible to DIR for all costs incurred by DIR, the State, or any of its Customers to replace the Successful Respondent. These costs include, but are not limited to, the costs of transitioning all Customers to a new TEX-AN 2021Contract Holder or procuring a substitute, and the cost of any claim or litigation that is attributable to Successful Respondent's failure to perform the Services in accordance with the terms of this CTSA. For purposes of clarity, the Successful Respondent's liability for State's costs of transitioning all Customers to a new TEX-AN 2021 Contract Holder will be limited to the total charges incurred by DIR and the Customers from the new TEX-AN 2021 Contract Holder in transitioning.

## 17.11 Dispute Resolution

### 17.11.1 General Agreement of the Parties

The Parties mutually agree that the interests of fairness, efficiency, and good business practices are best served when the Parties employ all reasonable and informal means to resolve any Dispute under this CTSA. The Parties express their mutual commitment to using all reasonable and informal means of resolving Disputes prior to invoking a remedy provided elsewhere in this Article 17.

### 17.11.2 Duty to Negotiate in Good Faith.

Any Dispute that in the judgment of any Party to this CTSA may materially or substantially affect the performance of any Party will be reduced to writing and delivered to the other Party per the formal notice procedures discussed above. The Parties must then negotiate in good faith and use reasonable efforts to resolve such dispute and the Parties will not resort to pursuing further remedies unless they have reasonably determined that a negotiated resolution is not possible. The resolution of any dispute disposed

Vendor Co	ntract No.	
-----------	------------	--

of by agreement between the Parties will be reduced to writing and delivered to all Parties within ten (10) Business Days.

#### 17.11.3 Claims for Breach of CTSA

#### 17.11.3.1 General Requirement

As required by and in accordance with Chapter 2260, Texas Government Code, Successful Respondent's claim for breach of this CTSA must be resolved in accordance with the dispute resolution process established by DIR.

## 17.11.3.2 Negotiation of Claims

- a) The Parties expressly agree that the Successful Respondent's claim for breach of this CTSA that the Parties cannot resolve in the ordinary course of business or through the use of all reasonable and informal means will be submitted to the negotiation process provided in Chapter 2260, Subchapter B, Texas Government Code.
- b) The Parties expressly agree that the Successful Respondent's compliance with Chapter 2260, Subchapter B, Texas Government Code, is a condition precedent to the filing of a contested case proceeding under Chapter 2260, Subchapter C, Texas Government Code.

## 17.11.3.3 Contested Case Proceedings

The contested case process provided in Chapter 2260, Subchapter C, Texas Government Code, will be Successful Respondent's sole and exclusive process for seeking a remedy for any and all alleged breaches of the CTSA by DIR if the Parties are unable to resolve their disputes under Subsection 17.11.3.2 of this section. The Parties expressly agree that compliance with the contested case process provided in Chapter 2260, Subchapter C, Texas Government Code, is a condition precedent to seeking consent to sue from the Texas Legislature under Chapter 107, Texas Civil Practices & Remedies Code. Neither the execution of this CTSA by DIR nor any other conduct of any representative of DIR relating to this CTSA shall be construed as or be considered a waiver of the State's sovereign immunity.

### 17.11.3.4 DIR Rules

The submission, processing, and resolution of Successful Respondent's claim is governed by the rules adopted by DIR pursuant to Chapter 2260, Texas Government Code, found at Title 1, Chapter 201, Texas Administrative Code.

## 17.11.3.5 Successful Respondent's Duty to Perform

Neither the occurrence of an event constituting an alleged breach of CTSA nor the pending status of any claim for breach of CTSA is grounds for the suspension of performance, in whole or in part, by Successful Respondent of any duty or obligation with respect to the performance of Services under this CTSA and all CSAs. Any changes to this CTSA as a result of a dispute resolution will be implemented in accordance with Article 9.

# 17.12 Liability of Successful Respondent

- a) Successful Respondent bears all risk of loss or damage due to:
  - 1) Defects in Services;
  - 2) Unfitness or obsolescence of Services; and
  - 3) The negligence or intentional misconduct of Successful Respondent or its employees, agents, or Subcontractors;

- b) Successful Respondent shall defend, indemnify and hold harmless the State, DIR and Customers, and/or their employees, agents, representatives, contractors, Assignees, officers, directors, successors, and/or designees from any and all liability, actions, claims, demands, or suits and all related costs, attorney fees, and expenses arising out of, or resulting from any acts or omissions of the Successful Respondent, Successful Respondent Personnel, Successful Respondent's Subcontractors, or suppliers of Subcontractors in the execution or performance of the CTSA, a CSA and any Purchase Orders issued under the CTSA regardless of the negligence of the Customer, the State, DIR, and/or their employees, agents, representatives, contractors, Assignees, officers, directors, successors, and/or designees. The defense shall be coordinated by the Texas Office of the Attorney General for State agency Customers and by Customer's legal counsel for non-State agency Customers.
- c) Successful Respondent will ship all Equipment, if any, related to the provision of Services under this CTSA and any CSAs entered into pursuant hereto, freight prepaid, freight on board (FOB) DIR's or Customer's destination or other designated location. The method of shipment will be consistent with the nature of the Equipment and hazards of transportation. Regardless of FOB point, Successful Respondent agrees to carry adequate freight insurance and bear all risks of loss, damage, or destruction of Equipment, in whole or in part, ordered hereunder that occurs prior to written Acceptance, except loss or damage attributable to DIR's or Customer's fault or negligence; and such loss, damage, or destruction will not release Successful Respondent from any obligation hereunder. After written Acceptance, the risk of loss or damage will be borne by DIR or the Customer, as appropriate, except loss or damage attributable to Successful Respondent's fault or negligence.

## 18 ASSURANCES AND CERTIFICATIONS

### **18.1** Conflicts of Interest

## 18.1.1 Representation

Successful Respondent agrees to comply with applicable State and federal laws, rules, and regulations regarding conflicts of interest in the performance of its duties under this CTSA. Successful Respondent warrants that it has no interest and will not acquire any direct or indirect interest that would conflict in any manner or degree with its performance under this CTSA.

#### 18.1.2 General Duty Regarding Conflicts of Interest

- a) Successful Respondent will establish safeguards to prohibit Successful Respondent Personnel and Subcontractors from using their positions for a purpose that constitutes or presents the appearance of personal or organizational conflict of interest, or personal gain. Successful Respondent shall, and Successful Respondent shall cause its Successful Respondent Personnel and Subcontractors, to operate with complete independence and objectivity without actual, potential, or apparent conflict of interest with respect to the activities conducted under this CTSA. Successful Respondent has disclosed, in its Response to the RFO, any existing conflicts of interest, including any situations that create an appearance of a conflict of interest. Successful Respondent has also disclosed the current or past employment of former state employees within two (2) years of the date of its Response to the RFO, under the terms of the disclosure requirement contained in the RFO.
- b) Successful Respondent shall for itself and on behalf of its Subcontractors, has identified (and will identify for the duration of the Term): A. all current or former employees of the State assigned or proposed to work on the CTSA twenty percent (20%) or more of their time and has disclosed

Vendor Contract No.	
---------------------	--

them to DIR and has disclosed or does not employ any relative of a current or former State employee within two (2) degrees of consanguinity or affinity, and, if these facts change during the course of the CTSA, Successful Respondent certifies for itself and on behalf of its Subcontractors, it shall disclose the name and other pertinent information about the employment of current and former employees and their relatives within two (2) degrees of consanguinity or affinity, and B. any former executive head of a State agency employed by Successful Respondent;

# 19 REPRESENTATIONS, WARRANTIES, AND COVENANTS

### 19.1 Authorization

Successful Respondent hereby represents and warrants the following:

- a) The execution, delivery, and performance of this CTSA has been duly authorized by Successful Respondent and no approval, authorization, or consent of any governmental or regulatory agency is required to be obtained in order for Successful Respondent to enter into this CTSA and perform its obligations under this CTSA.
- b) Successful Respondent has obtained all licenses, assignments, certifications, permits, and authorizations necessary to perform the Services under this CTSA and currently is in good standing with all regulatory agencies that regulate any or all aspects of Successful Respondent's performance of Services under this CTSA. Successful Respondent will maintain all required certifications, licenses, permits, and authorizations during the term of this CTSA.

## **19.2 Ability to Perform**

Successful Respondent warrants that it has the financial resources to fund the expenditures required to provide the Services under this CTSA without advances by DIR or assignment of any payments by DIR to a financing source.

## **19.3** Successful Respondent Certifications

Successful Respondent certifies, for itself, its agents, and on behalf of Subcontractors, as follows (all references within this section to Successful Respondent include Successful Respondent's agents and Subcontractors):

- a) Successful Respondent has not given, offered to give, and does not intend to give at any time hereafter any economic opportunity, future employment, gift, loan, gratuity, special discount, trip, favor, or service to a public servant in connection with the CTSA;
- b) Successful Respondent is not currently delinquent in the payment of any franchise tax owed the State and is not ineligible to receive payment under Section 231.006 of the Texas Family Code and acknowledges the CTSA may be terminated and payment withheld if this certification is inaccurate;
- neither Successful Respondent, nor anyone acting for it, has violated the antitrust laws of the
  United States or the State, nor communicated directly or indirectly to any competitor or any other
  person engaged in such line of business for the purpose of obtaining an unfair price advantage;
- d) Successful Respondent has not received payment from DIR or any of its employees for participating in the preparation of the CTSA;

- e) under Sections 2155.004, 2155.006, and 2261.053, Texas Government Code, Successful Respondent is not ineligible to receive the specified Contract and acknowledges that this CTSA may be terminated and payment withheld if this certification is inaccurate;
- f) to the best of its knowledge and belief, Successful Respondent knows there are no suits or proceedings pending or threatened against or affecting it, which if determined adversely to it will have a material adverse effect on the ability to fulfill its obligations under the CTSA;
- g) Successful Respondent and its principals are not suspended or debarred from doing business with the federal government as listed in the System for Award Management (SAM) maintained by the General Services Administration;
- h) as of the Effective Date of the CTSA, Successful Respondent is not listed in any of the divestment statute lists published on the CPA's website (https://comptroller.texas.gov/purchasing/publications/divestment.php);
- as of the Effective Date, Successful Respondent is not listed in the prohibited vendors list authorized by Executive Order #13224, "Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism," published by the United States Department of the Treasury, Office of Foreign Assets Control;
- j) Successful Respondent has identified (and will identify for the duration of the Term):
  - all current or former employees of the State assigned or proposed to work on the CTSA twenty percent (20%) or more of their time and has disclosed them to DIR and has disclosed or does not employ any relative of a current or former state employee within two (2) degrees of consanguinity or affinity, and, if these facts change during the course of the Agreement, Successful Respondent certifies for itself and on behalf of its Subcontractors, it shall disclose the name and other pertinent information about the employment of current and former employees and their relatives within two (2) degrees of consanguinity or affinity, and
  - 2) any former executive head of a State agency employed by Successful Respondent;
- k) that, for its performance of this CTSA, Successful Respondent shall purchase products and materials produced in the State when available at the price and time comparable to products and materials produced outside the State, to the extent that such is required under Section 2155.4441, Texas Government Code;
- Successful Respondent agrees that all equipment and materials used in fulfilling the requirements
  of this CTSA are of high-quality and consistent with or better than applicable industry standards,
  if any. All work and Services performed pursuant to this CTSA shall be of high professional
  quality and workmanship and according consistent with or better than applicable industry
  standards, if any;
- m) to the extent applicable to this scope of this CTSA, Successful Respondent is in compliance with Subchapter Y, Chapter 361, Health and Safety Code, related to the Computer Equipment Recycling Program and its rules, 30 Texas Administrative Code Chapter 328;
- n) Successful Respondent agrees that any payments due under this CTSA will be applied towards any debt, including but not limited to delinquent taxes and child support that is owed to the State;
- o) Successful Respondent is in compliance Section 669.003, Texas Government Code, relating to contracting with executive head of a State agency;
- p) the provision of goods and services or other performance under the CTSA will not constitute an actual or potential conflict of interest, Successful Respondent will not create the appearance of

Vendor Contract No.	
---------------------	--

- impropriety, and, if these facts change during the course of the CTSA, Successful Respondent certifies it shall disclose the actual or potential conflict of interest and any circumstances that create the appearance of impropriety;
- q) Successful Respondent has complied with the Section 556.0055, Texas Government Code, restriction on lobbying expenditures.
- r) Successful Respondent understands and acknowledges the applicability of Sections 2155.444 and 2155.4441, Texas Government Code, in fulfilling the terms of the CTSA;
- s) DIR's or the Customer's payment to Successful Respondent and Successful Respondent's receipt of appropriated or other funds under this CTSA are not prohibited by Sections 556.005 or 556.008, Texas Government Code;
- t) to the extent applicable to this scope of this CTSA, Successful Respondent is authorized to sell and provide warranty support for all products and Services listed in Exhibit B of this CTSA;
- in accordance with Section 2271.002 of the Texas Government Code, by signature hereon, Successful Respondent does not boycott Israel and will not boycott Israel during the term of this CTSA. Successful Respondent agrees to take all necessary steps to ensure this certification remains true during the Term of this CTSA;
- Successful Respondent represents and warrants that it is not engaged in business with Iran, Sudan, or a foreign terrorist organization, as prohibited by Section 2252.152 of the Texas Government Code.
- w) that in accordance with Section 2155.0061, Texas Government Code, the individual or business entity named in this CTSA is not ineligible to receive the specified contract and acknowledges that this CTSA may be terminated and payment withheld if this certification is inaccurate; and
- x) during the Term, Successful Respondent shall promptly disclose to DIR all changes that occur to the foregoing certifications, representations, and warranties. Successful Respondent covenants to fully cooperate in the development and execution of resulting documentation necessary to maintain an accurate record of the certifications, representations, and warranties.

# 19.4 Workmanship and Performance

- a) All Services provided under this CTSA will be provided in a manner consistent with the standards of quality as outlined in this CTSA.
- b) All Services must meet or exceed the required levels of performance specified in or pursuant to this CTSA, and will meet or exceed DIR's Goals and Objectives, as set forth in the RFO, Section 1.3
- c) Successful Respondent shall, and it shall cause Successful Respondent Personnel and Subcontractors, to perform the Services in a workmanlike manner, in accordance with best practices and high professional standards consistent with applicable industry standards used in well-managed operations performing Services similar to the Services awarded in this CTSA.

### 19.5 Compliance with Agreement

Successful Respondent represents, warrants, and covenants that it will not take any action inconsistent with any of the terms and conditions set forth in this CTSA without the express, prior written approval of DIR, in DIR's sole and absolute discretion.

Vendor	Contract No.	
vendor	COMPACE NO.	

## 19.6 Infringement and Misappropriation

- a) Successful Respondent represents, warrants, and covenants that all Services provided by Successful Respondent will not infringe or misappropriate any right of, and will be free of any claim of, any third person or entity based on copyright, patent, trade secret, or other Intellectual Property rights. Successful Respondent agrees that it shall include the indemnity provision provided in Section 14.10 b) into its Subcontracts, in which Subcontractors participate in the development or delivery of the Services or Software.
- b) As of the Effective Date, there is, to Successful Respondent's knowledge, no pending claim, suit, or proceeding against or affecting Successful Respondent or any of its Affiliates or Subcontractors that would reasonably be expected to adversely affect Successful Respondent's ability to perform and fulfill its obligations under this CTSA including, without limitation, actions pertaining to the proprietary rights to the relevant intellectual property. Successful Respondent shall notify DIR within fifteen (15) days of Successful Respondent's knowledge of any such claim, suit, or proceeding. Successful Respondent shall notify DIR, within forty-eight (48) hours, if process is served on Successful Respondent in connection with this CTSA where such matter may affect the Services or a Party's rights including any subpoena for Successful Respondent's records, and shall send a written notice of the service together with a copy of the same to DIR within seventy-two (72) hours of such service.

## 19.7 Organizational Conflicts of Interest

#### 19.7.1 Definition

An organizational conflict of interest is a set of facts or circumstances, a relationship, or other situation under which Successful Respondent, agent or Subcontractor has past, present, or currently planned personal or financial activities or interests that either directly or indirectly:

- a) Impairs or diminishes Successful Respondent's, agent's, or Subcontractor's ability to render impartial or non-discriminatory Services to DIR and the Customers; or
- b) Provides the Successful Respondent, agent, or Subcontractor an unfair competitive advantage in the Procurement and future DIR procurements.

#### 19.7.2 Warranty

Except as otherwise disclosed and approved by DIR prior to the Effective Date of this CTSA, Successful Respondent warrants that, as of the Effective Date and to the best of its knowledge and belief, there are no relevant facts or circumstances that could give rise to an organizational conflict of interest affecting this CTSA. Successful Respondent affirms that it has neither given, nor shall give, at any time hereafter, any economic opportunity, future employment, gift, loan, gratuity, special discount, trip, favor, or service to a public servant or any employee or representative of same, at any time during the procurement process or in connection with the procurement process except as allowed under relevant State and federal law.

## 19.7.3 Continuing Duty to Disclose

a) Successful Respondent agrees that, if after the Effective Date, Successful Respondent discovers or is made aware of an organizational conflict of interest, Successful Respondent will immediately and fully disclose such interest in writing to DIR. In addition, Successful Respondent must promptly disclose any relationship that might be perceived or represented as a conflict after its discovery by Successful Respondent or by DIR as a potential conflict. DIR reserves the right, in its sole and absolute discretion, to make a final determination regarding the existence of conflicts of interest, and Successful Respondent agrees to abide by DIR's decision.

Vendor Contract No.	
---------------------	--

b) The disclosure will include a description of the action(s) that Successful Respondent has taken or proposes to take to avoid or mitigate such organizational conflicts of interest.

## 19.7.4 Remedy

If DIR determines that an organizational conflict of interest exists, DIR may, at its sole discretion, terminate the CTSA. If DIR determines that Successful Respondent was aware of an organizational conflict of interest before the award of this CTSA and did not disclose the conflict in its Response to the RFO, such nondisclosure will be considered a Material Breach. Furthermore, the facts and circumstances related to such Material Breach may be submitted to the Texas Office of the Attorney General, Texas Ethics Commission, or appropriate State or federal law enforcement officials for further action.

### 19.7.5 Flow Down Obligation

Successful Respondent must include the provisions of CTSA Section 19.7 in all agent agreements and Subcontracts for work to be performed similar to the Services provided by Successful Respondent under this CTSA to preserve the State's rights against such agents and Subcontractors.

### 19.8 DIR Personnel Recruitment Prohibition

- a) Successful Respondent has not retained or promised to retain any agent, Subcontractor, person, or company, or utilized or promised to utilize an agent, or a Subcontractor that participated in DIR's development of specific criteria of this CTSA or who participated in the procurement that resulted in the selection of the Successful Respondent for this CTSA.
- b) Successful Respondent will not recruit or employ any DIR professional or technical personnel who have worked on Projects relating to the subject matter of this CTSA, or who have had any influence on decisions affecting the subject matter of this CTSA, for two (2) years following the expiration and/or termination of this CTSA.

#### 19.9 Anti-kickback Provision

Successful Respondent certifies that it will comply with the Anti-Kickback Act of 1986, 41 USC §§51-58 and Federal Acquisition Regulation 52.203-7.

#### 19.10 Debt or Back Taxes Owed to the State

In accordance with Section 403.055, Texas Government Code, Successful Respondent agrees that any payments due to Successful Respondent under this CTSA will be first applied toward any debt and/or back taxes Successful Respondent owes the State. Successful Respondent further agrees that payments will be so applied until such debts and back taxes are paid in full.

## 19.11 Outstanding Debts and Judgments

Successful Respondent certifies that it is not presently indebted to the State, and that Successful Respondent is not subject to an outstanding judgment in a suit by the State against Successful Respondent for collection of the balance. For purposes of this Section, an indebtedness is any amount sum of money that is due and owing to the State and is not currently under dispute. A false statement regarding Successful Respondent's status will be treated as a Material Breach.

Vendor	Contract No.		
vendor	COHITACI INO.		

#### 19.12 Antitrust

In submitting a Response, and in executing this CTSA, Successful Respondent certifies and agrees as follows:

- a) Neither Successful Respondent, nor agent, nor Subcontractor, nor the person represented by the Successful Respondent, nor any person acting for the represented person has:
  - 1) violated the antitrust laws codified by Sections 15.01, et seq., Texas Business & Commerce Code or the federal antitrust laws; or
  - 2) directly or indirectly communicated the Response associated with this CTSA to a competitor or other person engaged in the same line of business.
- b) Successful Respondent hereby assigns to DIR any and all claims for overcharges associated with this CTSA arising under the antitrust laws of the United States, 15 U.S.C. Section 1, et seq. (1973), as amended, and the antitrust laws of the State, Sections 15.01, et seq., Texas Business & Commerce Code.

## 19.13 Agency Executive

Under Section 669.003, Texas Government Code, relating to contracting with an executive of a State agency, Successful Respondent represents that no employee, consultant, officer, or director of Successful Respondent, or any other person involved with Successful Respondent, has, in the past four (4) years, served as an executive of DIR, or any other State agency, was involved with or has any interest in the RFO or any CTSA resulting from the RFO. If Successful Respondent employs or has used the services of a former executive head of DIR or other State agency, then Successful Respondent shall provide DIR the following information concerning such individual: Name of former executive, name of State agency, date of separation from State agency, position with Successful Respondent, and date of employment with Successful Respondent.

### 19.14 Deceptive Trade Practices; Unfair Business Practices

Successful Respondent represents and warrants that it has not been the subject of allegations of deceptive trade practices violations under Chapter 17, Texas Business & Commerce Code, or allegations of any unfair business practice in any administrative hearing or litigation and that Successful Respondent has not been found to be liable for such practices in such proceedings. Successful Respondent certifies that it has no officers who have served as officers of other entities who have been the subject allegations of deceptive trade practices violations or allegations of any unfair business practices in an administrative hearing or court suit and that such officers have not been found to be liable for such practices in such proceedings.

### 19.15 Cybersecurity Training

In accordance with Section 2054.5192, Texas Government Code, for any contract with a State agency or institution of higher education, if Successful Respondent, or a Subcontractor, officer, or Successful Respondent Personnel, will have access to a State computer system or database, then Successful Respondent shall ensure that such Successful Respondent Personnel, officer, or Subcontractor shall complete a cybersecurity training program certified under Section 2054.519, Texas Government Code, as selected by the State agency Customer. The cybersecurity training program must be completed by such Successful Respondent Personnel, officer, or Subcontractor during the Term of the CTSA and during any renewal period. Successful Respondent shall verify to the Customer state agency or institution of higher

Vendor	Contract No.	
--------	--------------	--

education completion of the program by each such Successful Respondent Personnel, officer, or Subcontractor.

# 19.16 Drug Free Workplace Policy

Successful Respondent shall comply with the applicable provisions of the Drug-Free Work Place Act of 1988 (41 U.S.C. §§8101-8106) and maintain a drug-free work environment; and the final rule, government-wide requirements for drug-free work place (Financial Assistance), issued by the Office of Management and Budget (2 C.F.R. Part 182) to implement the provisions of the Drug Free Work Place Act of 1988 is incorporated by reference and the Successful Respondent shall comply with the relevant provisions thereof, including any amendments to the final rule that may hereafter be issued.

#### 19.17 Preservation of Records

In accordance with Section 552.372 of the Texas Government Code, Respondent agrees to (1) preserve all contracting information related to the contract as provided by the records retention requirements applicable to the Agency for the duration of the contract, (2) promptly provide to the Agency any contracting information related to the contract that is in the custody or possession of the Respondent on request of the Agency, and (3) on termination or expiration of the contract, either provide at no cost to the Agency all contracting information related to the contract that is in the custody or possession of the Respondent or preserve the contracting information related to the contract as provided by the records retention requirements applicable to the Agency. Except as provided by Section 552.374(c) of the Texas Government Code, the requirements of Subchapter J, Chapter 552, Government Code, may apply to the contract and the Respondent agrees that the contract can be terminated if the Respondent knowingly or intentionally fails to comply with a requirement of that subchapter.

#### 20 INTELLECTUAL PROPERTY

## **20.1** Infringement Indemnification

- a) Successful Respondent will, at its expense, indemnify, and hold harmless DIR and the Customers, their respective employees, officers, directors, contractors, and agents from and against any losses, liabilities, damages, penalties, costs, and fees, including without limitation attorneys' fees and expenses, from any claim or action against DIR and/or Customers that is based on a claim of breach of the warranty set forth in SECTION 19.7.2 OF THE CTSA, DIR and/or Customer will promptly notify Successful Respondent in writing of the claim, provide Successful Respondent a copy of all information received by DIR and/or Customers with respect to the claim, and cooperate with Successful Respondent in defending or settling the claim. The defense will be coordinated by the Texas Office of the Attorney General for Texas State agencies and by Customer's legal counsel for non-State agency Customers.
- b) If any Service, or part thereof, that is the subject of an action described in Section 19.1(a), is held to constitute an infringement or misappropriation, or the use thereof is enjoined or restricted or if a proceeding appears to Successful Respondent to be likely to be brought with respect there to, Successful Respondent will, at its own expense, either:
  - 1) Procure for DIR and/or the affected Customer the right to continue using the Services; or
  - 2) Modify or replace the Services to comply with the specifications in the CTSA, if any, and to not violate any Intellectual Property rights.

Vendor	Contract No.	
--------	--------------	--

## 20.2 Exceptions

Successful Respondent is not responsible for any claimed breaches of the warranties set forth in Section 19.1 to the extent caused by:

- a) Modifications made to the Service in question by any party other than Successful Respondent or its agents or Subcontractors, or by DIR and/or Customer and/or their agents working at Successful Respondent's direction and in accordance with another vendors' specifications; or
- b) The combination, operation, or use of the Service with other items if Successful Respondent did not supply or approve for use with the item; or
- c) DIR's and/or Customer's failure to use any new or corrected versions of the Service made available or provided by Successful Respondent.

## 20.3 Rights in Data

- DIR or Customers will be and remain the owner of all Data made available by DIR or Customers to Successful Respondent or its agents, its Subcontractors or representatives pursuant to the CTSA. Successful Respondent and Successful Respondent Personnel will not use DIR or Customer Data for any purpose other than providing the Services, nor will any part of DIR or Customer Data be disclosed, sold, assigned, leased, or otherwise disposed of to the general public or to specific third parties (other than with respect to providing the Services) or commercially exploited by or on behalf of Successful Respondent, nor will any Successful Respondent Personnel other than those on a strictly need to know basis have access to DIR or Customer Data. Successful Respondent will not possess or assert any lien or other right against DIR or Customer Data. Without limiting the generality of this Section 20.3, Successful Respondent will only use personally identifiable information as strictly necessary to provide the Services and will disclose such information only to its employees who have a strict need to know such information, provided Successful Respondent obtains such employees' written agreement to keep such information confidential and to use it only as required in the performance of Services. Successful Respondent will comply at all times with all State and federal laws and regulations applicable to such personally identifiable information. Additionally, Successful Respondent may disclose such information to its Subcontractors who have a need to know in connection with the performance of Services, provided Successful Respondent obtains such Subcontractors' written agreement to keep such information confidential and to use it only as required in the performance of Services.
- b) DIR or Customer is and will remain the owner of all DIR or Customer Data pursuant to the CTSA. Except as otherwise provided in this CTSA, or as otherwise restricted by law, and excluding Successful Respondent Confidential Information, DIR and Customer may use the data provided by the Successful Respondent for any purpose relating to the operation and enhancement of TEX-AN 2021 Program. DIR or Customer will not possess or assert any lien or other right against the Successful Respondent's data.

### **20.4** Ownership of Pre-existing Materials

DIR or Customers and Successful Respondent will continue to own their respective proprietary technologies and information developed before entering into the CTSA. Any Software licensed through the Successful Respondent and sold to DIR or Customers as a part of a Service, will be licensed directly to DIR or Customers, as appropriate.

Vendor (	Contract No.	
----------	--------------	--

## **20.5** Third party Commercial Software

If applicable and necessary, all third party commercial Software used in performing the Services will be provided to DIR or Customers under a separate license agreement between DIR or Customers and the owner (or authorized licensor) of such software.

## **20.6** Pre-existing Materials for Custom Software Deliverables Related to Services

Neither Successful Respondent nor any of its agents, representatives, or Subcontractors will incorporate any pre-existing materials (including third party commercial software) into Services or use any pre-existing materials to produce Services if such pre-existing materials will be needed by DIR or Customer in order to use the Services unless:

- a) Such pre-existing materials and their owners are identified to DIR or Customer in writing prior to implementation or installation, and
- b) Such pre-existing materials are either readily commercially available products for which Successful Respondent or its Subcontractor, as the case may be, has obtained a license (in form and substance approved by DIR, in its sole and absolute discretion) in the name of DIR or Customer, or are materials that Successful Respondent or its Subcontractor, as the case may be, has the right to license to DIR or Customer and has agreed to license to DIR or Customer on terms and conditions approved by DIR, in its sole and absolute discretion, prior to using such preexisting materials to perform the Services. Successful Respondent represents, warrants, and covenants that it shall not, during the term of the CTSA or any time afterwards, assert any ownership rights, in any DIR-owned or Customer-owned Software, source code, and other materials.

# **20.7** Ownership of Work Product by DIR or Customer

The Services shall not be considered "work for hire" for DIR or Customers; however, the same shall be subject to licensing, on the same terms and provisions, as provided for herein, as reasonable, necessary, or required.

## **20.8** License to Intellectual Property

- a) If necessary to the performance of Services hereunder, upon notice from Successful Respondent, DIR and/or Customers shall enter into good faith negotiations to license to Successful Respondent any Intellectual Property owned by DIR and/or Customers.
- b) In the event the Successful Respondent makes or has made any modifications or configurations to any Software owned by DIR and/or a Customer under this CTSA, DIR and/or the Customer, as appropriate, shall own the modifications or configurations and the related scripts, processes, and documentation, and Successful Respondent hereby assigns all of its rights, titles and interests (including Intellectual Property rights) to such modifications or configurations and the related scripts, processes, and documentation to the appropriate party, either DIR or the Customer. Upon the request of DIR or the Customer, Successful Respondent will give a written assignment of the modifications or configurations and the related scripts, processes, and documentation to the requesting party. During the Term, Successful Respondent retains a terminable, non-transferable, non-exclusive and paid-up license to use the modifications in performing Services under this CTSA, and to use the modifications to provide Services to other Customers, including the right to sub-license or further license those modifications only to Customers.

Vendor Contract No.	
---------------------	--

c) The Successful Respondent shall own all modifications to the Successful Respondent owned Software made by the Successful Respondent and any modifications to the Successful Respondent Proprietary Software made on behalf of DIR and/or a Customer. DIR and/or the affected Customer hereby assign its rights, titles, and interests (including Intellectual Property rights) to such modifications to the Successful Respondent. Successful Respondent shall provide DIR and/or Customer with a nonexclusive, non-transferable, royalty-free limited right and license during the Term to use the modifications for purposes of receiving Services. Successful Respondent's license to DIR and/or the Customer to use Successful Respondent Proprietary Software shall extend to such modifications; provided, however, Successful Respondent's warranties and indemnities with respect to Successful Respondent Proprietary Software do not extend to any modifications made on behalf of DIR and/or a Customer by contractors other than Successful Respondent and its Subcontractors.

## 21 TRANSITION PLAN FOR EXPIRATION OF THE CTSA

Refer to Section 9, Termination Assistance, of the TEX-AN 2021 Statement of Work (SOW).

## **21.1** Scope

This transition plan for expiration of the CTSA addresses the orderly Transition to DIR, or the successor service provider, of all Services provided by Successful Respondent to DIR and/or Customers pursuant to a CSA through the CTSA, that has an expiration date beyond the Expiration Date of the CTSA (the Transition Effective Date), including planning activities which will be conducted prior to the Transition Effective Date. This transition plan does not apply to other contracts that Successful Respondent may hold with DIR outside the scope of this CTSA. Successful Respondent agrees that all references in this Article 21 to DIR, includes all employees of the agency and all designated contractors or consultants that may assist DIR in the Transition. DIR and Successful Respondent agree that the term "Transition" as used in this Article 21 means disposition from Successful Respondent to DIR, or the successor service provider, by any lawful method, including by sale and purchase, assumption, transfer, assignment, or other method.

### **21.2** General Statement for Transition

No later than twelve (12) months before the Transition Effective Date (commencement of Transition Period), Successful Respondent shall commence planning for the orderly Transition of all of the Services that have an expiration date beyond the Transition Effective Date under this CTSA to DIR or the successor service provider, in accordance with the terms of this Article, as the same may be amended from time to time. DIR and Successful Respondent desire to provide for the Transition of all things and matters comprising the Services, which have expiration dates beyond the Transition Effective Date. This Article shall not be construed narrowly, but broadly, to ensure the Transition of all Services with expiration dates beyond the Transition Effective Date occurs on or by that date. The parties do not intend for there to be residual Service-related matters that do not Transition on or by the Transition Effective Date.

### **21.3** Agreement for Transition Assistance Services

Transition Assistance Services, as specified in this Article 21, shall be provided from the commencement of the Transition Period and for up to twelve (12) months after the Transition Effective Date, unless further extended by the Parties. DIR will provide Successful Respondent with written notice of its intent to have Transition Assistance Services, in whole or in part, provided by Successful Respondent after the

Vendor Contract No.	
---------------------	--

Transition Effective Date, by giving at least sixty (60) calendar days advance notice. Such notice shall contain a list of the specific Transition Assistance Services required and the duration for each of the Transition Assistance Services sought. Successful Respondent shall review the notice and provide a written proposal response for the specific services sought, as more particularly described in Section 21.04(b)(5), within thirty (30) calendar days of receipt of DIR's notice. The parties agree to negotiate in good faith to conclude a written agreement for the required Transition Assistance Services no later than one-hundred twenty (120) calendar days prior to the Transition Effective Date.

#### 21.4 Transition Assistance Services

Transition Assistance Services shall include:

- a) Full cooperation and participation in the Transition planning process, as more particularly described in Section 21.6, which shall commence no later than the commencement of the Transition Period, at DIR's sole and absolute discretion, and shall involve a series of meetings among the representatives of DIR, Vendor, the successor service provider, if known, representatives of Successful Respondent's current State agency Customers and other parties whose participation would enhance the planning process, which process is more fully described in Phase Four (4) set forth in this Section 21.4; and
- b) Successful Respondent providing to DIR, or the successor Service Provider, all of the items and matters called out for in this Article 21, and other items or information reasonably necessary to complete the transition of the Services under the CTSA. Without limiting the foregoing, the Transition Assistance Services shall occur in five (5) phases:
  - Phase 1 Initial Reporting of Services Subject to Transition. On or by December 31st two (2) years prior to the Transition Effective Date, Successful Respondent shall provide a written report, in the form required by DIR, disclosing all Services, as of December 1st of the same year, that have expiration dates beyond the applicable Transition Effective Date under the CTSA. Without limiting the foregoing, Successful Respondent shall report on and include:
    - i. All Software that Successful Respondent uses to perform its Services with each Customer under each of the Services, including complete source and machine readable data, associated manuals, procedures, processes, documentation, descriptions, data files, specifications, configurations, and other such items as available in accordance with existing third party licensing agreements. For each item of Software listed, Successful Respondent shall provide a copy of the associated Software license held by Successful Respondent. Successful Respondent shall ensure the transferability of the associated licenses to a successor entity under conditions similar to that of this Transition with no fees or costs for Transition to DIR or the successor service provider, if any;
    - ii. All hardware and other Equipment, including all hardware leases for Equipment used to perform each Service, including leases which have a term beyond the Transition Effective Date. Successful Respondent shall provide a copy of the assignment/transfer and early termination clauses for each lease with a term beyond the Transition Effective Date and report on the status of transferability and any costs for transfer or early termination, if any. Successful Respondent shall report all known costs for transfer or termination of such leases, including the existing leases or as the same may be amended prior to the Transition Effective Date;

Vendor Contract No.	
---------------------	--

- iii. All maintenance agreements for Software, hardware, or other Equipment and assets, including copies of their assignment/transfer and early termination clauses, a status report on transferability to DIR or its designee and the costs for transfer or early termination, if any;
- iv. All third party supplier agreements, which have an effective date beyond the Transition Effective Date, including copies of their assignment/transfer and early termination clauses, a status report on transferability to DIR or its designee and the costs for transfer or early termination, if any;
- v. Copies of the CSAs with the Customers for Services.
- 2) Phase 2 Ongoing Reports on Services through the Transition Effective Date. Phase 2 is composed of five (5) activities, as listed below:
  - i. Follow Up Reporting. For each item reported in Phase I, Successful Respondent shall update DIR on a semi-annual basis up to the Transition Effective Date. The first such update after December 31st shall be due on or by June 30th of the following year.
  - ii. New TEX-AN 2021 CSA. For every CSA issued by Successful Respondent to a Customer between the initiation of the reporting period two (2) years out from the Transition Effective Date and the Transition Effective Date, Successful Respondent shall seek and receive DIR's approval prior to entering into any CSA. DIR shall be provided no less than five (5) Business Days in which to review the documentation for each proposed CSA, in substantially final form. In each instance Successful Respondent shall provide that the CSA contains: 1) a complete assignment or transfer to DIR, or its successor service provider, on or by the Transition Effective Date, preferably at no cost to DIR or its designee, and 2) a termination for convenience clause in favor of Successful Respondent and DIR in all such CSAs. Successful Respondent shall use commercially reasonable efforts to minimize the fee for assignment or transfer to DIR, or its successor service provider.
  - iii. Equipment Condition Report. Successful Respondent will develop a checklist and conduct checks/reviews of the installed Equipment and produce a report on or by six (6) months prior to the Transition Effective Date. Such report shall detail, at a minimum, the then-current condition of each piece of installed Equipment and the preventative maintenance needed within the next six (6) months after the report is issued. For all items, Successful Respondent shall provide details about existing maintenance contracts.
  - iv. Subcontractor Report. On or by January 1st two (2) years prior to the Transition Effective Date, and semi-annually thereafter up to the Transition Effective Date, Successful Respondent shall identify all then-current Subcontracts that provide Services under any CSA issued pursuant to the CTSA, in a form and level of detail approved by DIR, in its sole and absolute discretion.
  - v. Notwithstanding any other provision in this Article 21, Successful Respondent agrees to provide another updated report for all CSAs on or by August 31st, one (1) year prior to the Transition Effective Date, and six (6) months prior to the Transition Effective Date.
- 3) Phase 3 Completion of all documentation for Transition
  - i. DIR, or the successor service provider and Successful Respondent shall complete Transition associated documentation necessary to effect the Transition of all CSAs

Vendor Contract No.	
---------------------	--

under the CTSA in a timely manner, including, at a minimum, the Transition of all CSAs reported to DIR under Subsections (1) and (2) hereof. Successful Respondent shall facilitate interactions and necessary documentation with third party vendors, its Subcontractors, lessors, and other providers, at the request of DIR, or at the request of the successor service provider. DIR reserves the right, in its sole and absolute discretion, to require that DIR, the successor service provider, and Successful Respondent conduct a formal closing for the Transition, in order to ensure the completion of all necessary documentation.

- ii. DIR, or the successor service provider, shall cause each Transition document to provide the complete transfer of all right, title, and interest in and to the CSAs, including any related lease or third party contract related to the underlying Equipment, from Successful Respondent to DIR, or the successor service provider, effective as of the Transition Effective Date. Each Transition document shall also provide that from and after the Transition Effective Date, Successful Respondent shall have no obligation or liability to any third party vendor or lessor whatsoever related to the underlying Equipment. Successful Respondent shall ensure that the granting of such license, sublicense, assignment, and other relevant rights is not subject to third party approval or payment by DIR, Customer, or the successor service provider of license, assignment, or transfer fees or penalties. Successful Respondent shall have no responsibility to pay any third party vendor or lessor from and after the Transition Effective Date.
- 4) Phase 4 Planning Phase for Transition from the CTSA to an Agreement with a successor service provider. Successful Respondent shall fully and completely participate in all meetings called by DIR for the purpose of planning and implementing the transition of CSAs from the CTSA to the successor service provider. The goal of this planning phase is to provide an orderly transition of CSA operations under the CTSA to the operations with a successor service provider. Such planning activities shall commence no later than January 1st of the year of the Transition Effective Date, at DIR's direction and shall continue until all transition activities are complete. These Transition activities shall include, but not be limited to, the following:
  - i. Pre-planning no later than January 1st to March 31st of the year of the Transition Effective Date. These DIR meetings will include the Successful Respondent's current Customers, as determined by DIR, the successor service provider, Successful Respondent, and DIR. The scope of the pre-planning is to plan for the transfers of the current Customers' CSA operations, scheduling of all necessary activities, development of roles and responsibilities matrix related to Transition of the CSA, and planning for all the other steps of the Transition process, as listed below. The goal of this pre-planning phase is to produce a mutually acceptable Transition Plan document that defines roles and responsibilities for accomplishing each step in the Transition among the participating parties.
  - ii. Turnover of CSA operations to the successor service provider. In each step in the Transition planning and execution process, Successful Respondent is required to provide information as requested by DIR, the successor service provider and /or the current Successful Respondent's Customers in accordance with the agreed upon Transition Plan. Such information shall include, but not be limited to the following tangible items and matters of inherent knowledge:

- A. Assets Lists (hardware, Software, schematics, maintenance agreements, third party agreements)
- B. Procedures
- C. Policies
- D. Practices
- E. Protocols
- F. Customer biographies
- G. Organization charts
- H. Customer and Successful Respondent (including Subcontractors) decision makers; authorized Successful Respondent Personnel
- I. Institutional knowledge; unwritten rules; operational personnel; client background information; authorization lists;
- J. Operational details
- K. and
- L. Quick reference manuals.
- 5) Phase 5 Post-Transition Effective Date Services by Successful Respondent. At DIR's option, DIR will request and Successful Respondent shall respond with a proposal for Successful Respondent to provide assistance for continuing management and operation of the CSA as conducted under the CTSA for a period beyond Transition Effective Date up to one (1) year after the Transition Effective Date, but only upon the execution of an appropriate contract for such services, on terms and conditions as agreed by DIR, or the successor service provider, and Successful Respondent. In the event that DIR, or the successor service provider, requires the services of Successful Respondent during the Post-Transition Effective Date period, then DIR shall execute a separate agreement with Successful Respondent to provide services on a time and material basis.

# **21.5** Covenant of Cooperation

During the period of time from the commencement of the Transition, as defined, to the completion of Transition Assistance Services, Successful Respondent agrees to fully cooperate with DIR, and as appropriate, with the successor service provider, to fulfill the terms of the CTSA. DIR and Successful Respondent agree to fulfill the terms of this Article 21 in a manner that is the least disruptive to the Customers under the CSA. DIR and Successful Respondent agree to notify the other Party in the event either discovers a condition, contract, or other matter that would benefit or adversely affect performance of the CSA, if generally known. In the event it is determined that amendment to this Article 21 is in the best interest of the State, Successful Respondent agrees to negotiate amendments to this Article 21 in good faith with DIR. In all events, the Parties agree to cooperate in the negotiation and execution of such other and further documents as necessary to fulfill the terms of this Article 21.

### **21.6** Other Successful Respondent Agreements with Customers

DIR and Successful Respondent agree that this Article does not apply to any other contracts that Successful Respondent has, or may have in the future, with Customer that are separate transactions from the CSA authorized under the CTSA.

Vendor	Contract No.	
vendor	COMPACE NO.	

### 22 MISCELLANEOUS

### **22.1** Entire Agreement

This CTSA, including with any documents, attachments, and exhibits specifically referred to herein or attached hereto, each of which is incorporated herein for all purposes, constitutes the entire agreement with respect to the subject matter hereof between the Parties. On execution of this CTSA, all prior agreements and understandings between the Parties relating to the subject matter hereof shall be null and void. The Parties specifically acknowledge that there are no unwritten side agreements or oral agreements between them which alter, amend, modify, or supplement this CTSA as of the Effective Date.

### **22.2** Counterparts

This CTSA may be executed in one (1) or more counterparts, each of which shall be deemed an original and all of which together shall constitute one and the same single agreement. Electronic signatures shall be deemed original signatures.

### **22.3** Authority to Execute

IN WITNESS WHEREOF, the Parties have executed this TEX-AN 2021 Communications Technology Services Agreement in their capacities as stated below with authority to bind their organizations on the dates set forth by their signatures. This Agreement shall be effective from the date of the last signature thereto (the Effective Date).

### **Department of Information Resources**

Amanda Garofor

Amanda Crawford Executive Director

11/5/2021 | 5:26 PM CDT

Date

M.H.

Legal

Hughes Network Systems, LLC (Successful Respondent)

—pocusigned by: Mathuw Brown

Matthew Brown Vice President and Associate General Counsel

10/28/2021 | 12:59 PM PDT

Date

### Exhibit A To TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Vendor	Contract No.	
vendor	COMPACE NO.	

### EXHIBIT A. DEFINITIONS

### 1 Introduction

The purpose of *Exhibit A Definitions* is to define certain terms used for the CTSA. This only contains definitions for words or phrases to which DIR is ascribing a special meaning. Words and phrases not listed here are to be given their usual definition within the context of the sentence in which they are used.

### 2 DEFINITIONS AND ACRONYMS

Term	Definition	
ADA	Americans with Disabilities Act.	
ADC	Austin Data Center.	
Acceptance	Agreement from DIR and/or its Customer(s) that Service installations, moves,	
	or changes have been successfully completed.	
Adjustment	Correction to an Invoice resulting from a billing error or as a result of Billing	
	Dispute reconciliation.	
Administrative Fee	In general, the fee paid to DIR to defray the DIR costs of operating and	
	administering the TEX-AN 2021 Program. The fee is a percentage applied to	
	Successful Respondent's Service costs.	
Affiliate	(i) Any person or entity directly or indirectly owning, controlling or holding,	
	with power to vote, fifty percent (50%) or more of the outstanding voting	
	securities of Successful Respondent; (ii) any person or entity for which fifty	
	percent (50%) or more of the outstanding voting securities are directly or	
	indirectly owned, controlled or held, with power to vote by Successful	
	Respondent; or (iii) any person, or entity directly or indirectly controlling,	
	controlled by, or under common control with, Successful Respondent.	
Assignee	An individual or organization that receives an assignment of all or part of the	
	interests of one of the Party's to the CTSA, and that undertakes to assume the	
_	responsibilities of that original Party.	
Assurance	Set of processes used to maintain and guarantee the required performance of a	
A 4414	Service.	
Attachment Authorized End	A form or document appended to an exhibit to the CTSA.	
Users or Authorized	Individuals or organizations that are eligible to use the Services by statute, also	
User User	known as <b>Customers</b> .	
Availability	The ratio of the time when a Service is functional and available to the user for	
Availability	the performance of prescribed functions (e.g., to originate and terminate Calls)	
	to the total amount of time in the reporting interval, minus scheduled	
	maintenance Downtime. Availability is expressed as a decimal between 0 and	
	1 and is normally calculated for one (1) calendar month.	
Bill	A formal statement of charges. See also Invoice.	
Billing Dispute	A Billing Inquiry may lead to a written notice from the DIR or Customer to the	
Diming Dispute	Successful Respondent regarding an error in data or billing element requiring	
	an adjustment or correction to the Monthly Consolidated Invoice which may	
	include any applicable Credits.	
	merade any appreadic creatis.	

Term	Definition	
Billing Inquiry	Question or inquiry submitted by DIR or Customer to the Successful	
	Respondent regarding billing which, if not satisfactorily resolved by DIR and	
	the Successful Respondent, may result in a Billing Dispute.	
Billing Plan	A plan submitted with Successful Respondent's Response demonstrating	
	compliance with the RFO for billing matters. The final Billing Plan, as	
	approved by DIR per Exhibit D, Billing Plan.	
BMIC	Basic Maintenance of Internal Connections.	
<b>Business Day</b>	Monday through Friday, excluding State holidays, 7:30 a.m. to 5:00 p.m. local	
	time. State holidays will include all holidays with the status "All Agencies	
	closed" as designated by the State Auditor's Office. State holidays will not	
	include State optional holidays or holidays that require skeleton crews. If the	
	CTSA calls for performance on a day that is not a Business Day, then	
~	performance is intended to occur on the next Business Day.	
Cause	Has the meaning as provided in Section <u>17.3</u> of this CTSA.	
CFO	Chief Financial Office.	
Change(s)	Any planned modification to a Service Instance or to a Successful	
	Respondent's network. Changes may include corrections, enhancements,	
~	modifications, additions, and replacements.	
Commercially	As applied to a Service in a geographic area, a Service or related Service	
Available	feature provided legally by a Successful Respondent to one (1) or more	
	entities, which as independent entities from such Successful Respondent, use	
C	the Services for its legal commercial business purposes.	
Comptroller of Public Accounts	A State agency. The State's chief tax collector, accountant, revenue estimator	
(CPA)	and treasurer. CPA includes the Texas State Treasury.	
Confidential	Is defined in CTSA Section 15.1(a).	
Information	is defined in C15/4 Section 13.1(a).	
Consolidated State	Means the consolidated telecommunications system as defined by Section	
Network	2170.001(a)(2), Texas Government Code, as the network of	
	telecommunications services serving the State government. This Consolidated	
	State Network is composed of DIR, Customer, and Successful Respondent	
	Network segments as defined herein.	
Contract (or CTSA)	Means the TEX-AN 2021 Communications Technology Services Agreement,	
	or (CTSA) – The final version of this contractually binding agreement between	
	the Texas Department of Information Resources and the Successful	
	Respondent awarded pursuant to the RFO, including all exhibits and	
	attachments thereto.	
Contract	The date marking the end of the period commencing on the Effective Date and	
Anniversary Date	ending one (1) year later and for each year in the Term thereafter.	
Contract Holder(s)	Other Successful Respondents awarded a Contract pursuant to RFO DIR-CPO-	
C	TMP-552 and part of the TEX-AN 2021 Program.	
Corrective Action	Has the meaning assigned in Section <u>17.2</u> , Tailored Remedies.	
Plan	Commercial off the shalf	
Crodit(s)	Commercial off-the shelf.	
CTCD	Monetary assessments against amounts owed.	
CTCD	Certified Texas Contract Developer.	
CTCM	Certified Texas Contract Manager.	

Term	Definition	
CTS	Communication Technology Services.	
<b>Current Service</b>	The geographic area where the Successful Respondent is currently providing	
Area	Commercially Available services.	
Customer	Any eligible purchaser of Services. Customers are defined in accordance with	
	Section 2170.004, Texas Government Code.	
<b>Customer Circuit</b>	DIR internal tracking number based on the agency/division of the Customer	
Record Code (CKR)	and the type of Service ordered.	
<b>Customer Facilities</b>	Building and infrastructure owned or operated by Customer.	
<b>Customer Network</b>	Customer networks are built and maintained by the Customer. Customer	
	Networks may or may not connect to the DIR Network or a Successful	
	Respondent Network, depending upon whether the Customer needs access to	
	Services provided by the DIR Network or a Successful Respondent Network.	
	Network to network demarcation points are determined on a case by case basis	
	depending upon type of Service.	
<b>Customer Premises</b>	All communications terminal Equipment that is located on the Customer's	
<b>Equipment (CPE)</b>	premises.	
Data	Information available in physical and/or electronic form.	
<b>Data Destruction</b>	Handling, securing and disposing of data in both physical and electronic form,	
	in compliance with, federal and State Laws that include, but are not limited to:	
	1 Texas Administrative Code Chapter 202; Section 403.278, Texas	
	Government Code; Chapter 2175, Texas Government Code; the Internal	
	Revenue Code; the Fair and Accurate Credit Transactions Act; the Family	
	Educational Rights and Privacy Act; the Health Insurance Portability and	
	Accountability Act of 1996; and the Gramm-Leach-Bliley Act.	
<b>Deficiency</b> (ies)	Has the meaning as used in Section 17.2.2 of this CTSA.	
Deliverable	A report or item that must be completed and delivered under the terms of the	
	CTSA or a CSA. The measurable result or output of a process prepared,	
	developed, or procured by Successful Respondent as part of the Services under	
	the CTSA or a CSA.	
Department of	State agency that oversees and assists in managing the information resource	
Information (DID)	assets of the State pursuant to Chapters 2054, 2059, and 2170, Texas	
Resources (DIR)	Government Code, as the same may be amended from time to time.	
Descriptions	The description(s) of the awarded Services set forth in Exhibit B.	
Developing	Includes technologies Respondents currently have and those that are in work	
Technologies  DID Goodson at any	and soon to become available.	
DIR Contractor	Has the meaning as defined in Section <u>4.5</u> of this CTSA.	
DIR Cooperative	A streamlined cooperative purchasing program for state, local, public	
(Coop) Contracts	education, and other public entities within and outside the State of Texas,	
Program DIR Facilities	administered by the CPO of DIR.	
DIR Facilities DIR Network(s)	Building and infrastructure owned or operated by DIR.  The DIP Network provides connectivity for primarily State government.	
DIK NELWOFK(S)	The DIR Network provides connectivity for primarily State government agencies but also includes educational institutions, political subdivisions, and	
	assistance organizations. Customer and Contract Holder Networks connect to	
	the DIR Network for interconnectivity and for enterprise services such as	
	connectivity to the State data centers and internet service.	
DIR Rules	Rules and procedures promulgated by DIR in accordance with the Texas	
DIN Nuics	Administrative Procedures Act and Texas Administrative Code.	
	Administrative Frocedures Act and rexast Administrative Code.	

Term	Definition	
Direct Sales	Direct Sales Transaction Services – Refer to Section <u>4.1</u> of this CTSA Services	
Transaction	that are direct billed as determined by DIR.	
Services	·	
Disentanglement	The process of winding up all issues in the business relationship between DIR,	
	Customer, and Successful Respondent in the event of any termination of the	
	CTSA. See Section <u>17.7</u> , Payment and Other Provisions at CTSA Termination.	
Disentanglement	The plan developed and submitted by Successful Respondent and approved by	
Plan	DIR, in its sole and absolute discretion, to provide for the orderly and efficient	
	winding up of the issues of the business relationship between Successful	
	Respondent, Customer, and DIR in the event of any termination of the CTSA.	
Disputes	See Section <u>17.7</u> , <u>Payment and Other Provisions at CTSA Termination</u> .  A Billing Dispute is a disagreement over the accuracy of charges in an Invoice	
Disputes	either to DIR or a Customer. Other Disputes may be disagreements over	
	actions taken or not taken pursuant to the CTSA or a CSA, which are subject	
	to the provisions of Section 16.10 Dispute Resolution of the CTSA.	
Downtime	Period of time when the system is unavailable, in whole or in part. Downtime	
	may be scheduled or unscheduled.	
DWDM	Dense wave division multiplexing.	
Effective Date	The commencement date for the CTSA, which is the date on which the last	
	Party signs the CTSA.	
Electronic	An electronic fast-track system process for the continuous improvement of the	
Administrative	Rates to DIR and Prices available to the Customers under the CTSA, and other	
Update (EAU)	non-substantive administrative updates, without reliance on a formal amendment process.	
Electronic State	A web portal operated by the CPA, to which all State agencies must post	
Business Daily	procurement opportunities, in compliance with Section 2155.083, Texas	
Dusiness Dany	Government Code.	
Emerging	Any technology within scope of the RFO currently not available from a	
Technologies	Successful Respondent that may become available during the term of the	
	CTSA.	
EPLS	Excluded Persons List System.	
Equipment	Computer hardware, routers, telecommunication devices, peripherals, and	
	similar materials (and all modifications, replacements, upgrades,	
	enhancements, documentation, materials, and media related thereto) that are	
	used or provided in connection with this CTSA or Services related to this	
ESBD	CTSA.  Flactronic State Rusiness Daily	
Executive	Electronic State Business Daily.  A committee of DIR executives tasked with helping resolve issues that may	
Leadership Team	arise between DIR and Successful Respondent.	
Expiration Date	As defined in CTSA Section 3.1.	
Failure	A detected cessation of capability to perform a specified function or functions	
	within previously established limits, which is beyond adjustment by the	
	operator through means of controls normally accessible during routine	
	operations of the system.	
FCC	The Federal Communication Commission, with regulatory authority over the	
	telecommunications industry, including certain telecommunications carriers.	

Vendor Contract No.\_\_\_\_\_

Term	Definition	
Feature	Additional capability beyond the basic Service that may be selected at the	
	Customer's option. Features normally are separately priced, although some	
	Features have been defined to be Not Separately Priced.	
Force Majeure	An event experienced by a Party that is within the scope of the CTSA clause	
Event	found at Section 8.3.2.	
<b>Government Entity</b>	Has the meaning provided in Texas Government Code § 2170.004.	
or Entities		
Grade of Service	A number of network design variables used to provide a measure of adequacy	
	for a group of resources under specified conditions.	
HUB	Historically Underutilized Business.	
HSP	HUB Subcontracting Plan.	
HVAC	Heating, ventilation, and air conditioning.	
IC	Internal connections.	
ID	Identification.	
Independent	A third party provider of independent assessment and testing methodologies	
Verification and	for quality assurance or performance verification purposes.	
Validation		
contractors		
Initial Term	Has the meaning provided in Section 3.1 of the CTSA.	
Intellectual	Any trademarks, patents, inventions, copyrights, trade secrets, or domain	
Property	names created prior to, on, or after the Effective Date.	
Invoice	Depending on the context in which used, may mean the Monthly Consolidated	
	Invoice issued by Successful Respondent to DIR, or an Invoice issued to a	
	Direct Sales Transaction Service Customer for Direct Sales Transaction	
- A	Services. In the proper context, Invoice may refer to both types of Invoices.	
Invoice File	Summarized Invoice data that are delivered by the Successful Respondent to	
	DIR, or a Customer, in a machine-readable format.	
IT	Information technology.	
IVR	Interactive voice response.	
Key Performance	Measurable Service attribute that is critical to its proper functioning and	
Indicator (KPI)	delivery to the Customer.	
LAN	Local area network.	
Local Services	A type of Service as discussed in Section 11.2 e) of the SOW, which may be	
	directly sold to Customers by Successful Respondent as set forth in Article 4	
Managed Network	of the CTSA, if awarded to Successful Respondent under this CTSA.  Has the meaning set forth in RFO Section 2.1.a) including facility, power, and	
Services	HVAC solutions.	
Managed Services	Has the meaning set forth in RFO Section 2.1.	
Management	Has the meaning provided in Section 2.1.  Has the meaning provided in Section 2.3 and in the CTSA's Exhibit named	
Plan(s)	Plans.	
Marketing Plan	A plan submitted with Successful Respondent's Response to demonstrate	
True noting 1 lan	Successful Respondent's resources and activities to promote the CTSA and	
	gain Customers. The Marketing Plan, as approved by DIR, in its sole and	
	absolute discretion, shall become an attachment to the CTSA.	
Material Breach	Has the meaning set forth in Section 16.2 Tailored Remedies.	
MIBS	Managed internal broadband services.	
MPLS	Multiple protocol label switching.	
	I	

Term	Definition	
Monthly	A formal, single statement of charges provided by Successful Respondent to	
Consolidated	DIR, which includes the Invoice file, Detailed Billing File, Adjustment file,	
Invoice	and monthly informational memorandum for all Services provided by	
	Successful Respondent.	
<b>Monthly Recurring</b>	Regular fees repeatedly billed each month by Successful Respondents for	
Charge (MRC)	Services performed. See also Non-Recurring Charge (NRC).	
Move, Add, Change,	Term describing a variance to Services.	
or Delete (MACD)		
Network Access	Has the meaning set forth in RFO Section 2.1.a) and 2.1.b).	
Network Power	Has the meaning set forth in Section ??? of the SOW addressing Network	
Management	Power Management Services and Section ??? of the SOW addressing Network	
Services	Power Management Services Response Instructions.	
Network Security	The NSOC, which is owned by DIR, protects TEX-AN 2021 computer	
<b>Operations Center</b>	systems and information assets from unauthorized external intervention or	
(NSOC)	improper use, in accordance with Chapter 2059, Texas Government Code.	
Next Generation 911	Has the meaning set forth in Section ??? of the SOW addressing 911 Services	
(also NextGen 911)	to a 911 Administrative Entity and Section ??? addressing 911 Services to a	
	911 Administrative Entity Response Instructions.	
Non-Recurring	Extraordinary or unusual fees that are unlikely to occur again in the normal	
Charge (NRC)	course of performing Services. They include fees for facilities, Services, or	
	products that occurred one time or infrequently. Examples include expedite	
	fees, installations, or special construction.	
Not Separately	The price and or cost for a particular cost or fee element which includes	
Priced (NSP)	another cost or fee element.	
NSOC	Network and security operations center.	
Order	A request or authorization by DIR or Customer, as applicable, to acquire a	
	Service.	
Parties or Party	Has the meaning given in the Preamble to CTSA.	
PBX	Private branch exchange.	
Performance Data	Measured and recorded data that reflect resource utilization.	
Performance	Has the meaning given in the Performance Management Exhibit.	
Management		
Point of Presence	Successful Respondent-owned or controlled physical location (1) where	
(POP)	Successful Respondent-owned or controlled network facilities used to provide	
	Services are located and (2) where access to or from on-net locations is	
- ·	connected to the Successful Respondent's Network.	
Prices	Are the values for Direct Sales Transaction Services as reflected in Exhibit B	
	to the CTSA which Successful Respondent is allowed to directly bill to	
	Customers and are composed of the Successful Respondent's price plus the	
	Administrative Fee, and also includes the values billed to Customers by DIR	
D • 4	acting as the Successful Respondent's billing agent.	
Project	A transaction that requires special, custom, or unique treatment by the	
	Successful Respondent due to size, complexity, or importance of the Services	
Dublic Inferment	ordered.	
Public Information	As defined by the Texas Public Information Act, Texas Government Code,	
	Chapter 552.	

Term	Definition	
Public Safety	Has the meaning as describe in Section 4.1 of the SOW addressing 911	
	Services to a 911 Administrative Entity and in Section 11.9 of the SOW	
	addressing 911 Services to a 911 Administrative Entity Response Instructions.	
PUC	The Texas Public Utility Commission.	
Purchase Order	DIR's or Customer's fiscal form or format, which is used when making a	
	purchase of Services (e.g., formal written Purchase Order, Procurement Card,	
	Electronic Purchase Order, or another authorized instrument).	
Rates to DIR	The Successful Respondent's charges to DIR for Services, as established and	
	agreed to between DIR and the Successful Respondent in the CTSA. Rates to	
	DIR do not include the Administrative Fee.	
Renewal Term(s)	Has the meaning set forth in Section 3.2 of the CTSA.	
Request for Offer	Has the meaning given in the Preamble to the CTSA.	
(RFO)		
Respondent	A firm, company, entity or individual that responds to the solicitation. Unless	
	the Contract clearly indicates otherwise, all terms and conditions of the	
	Contract that refer to Respondent apply with equal force to Successful	
	Respondent.	
Response	Has the meaning given in the Preamble to the CTSA.	
RFO	Request for offer.	
RFRO	Request for revised offer.	
SAM	System for Award Management.	
SD WAN	Software defined wide area network.	
SDC	San Angelo Data Center.	
SDN	Software defined networking.	
Security Incident	An event which results in accidental or deliberate unauthorized access,	
	exposure, loss, disclosure, modification, disruption, or destruction of	
	information resources, including DIR or Customer data.	
Service Level	Agreement between DIR and Successful Respondent which establishes the	
Agreement(s)	minimum standard of performance for Services by the Successful Respondent	
(SLA(s))	under the CTSA. Compliance with SLAs is determined by measurement of	
	KPIs and AQLs. The SLA(s) also specifies the amount of Credit to which DIR	
	or the Customer is entitled if the Successful Respondent fails to meet the	
G • T 4	applicable AQL.	
Service Instance	Has the meaning as used in Section $\underline{2.4}$ of the CTSA's Performance	
Service Order	Management Exhibit.  Written notice from Successful Respondent that contains data elements	
Completion Notice	notifying DIR and/or the Customer that the Service for a given Order has been	
(SOCN)		
Service(s)	fully installed and is ready for Acceptance.  Means the TEX-AN 2021 Communications Technology Services (TEX-AN	
Set vice(s)	•	
SHB	2021 Services) as specified in the CTSA and awarded under the CTSA.  Sam Houston building	
SIM Platform	Security information management system hardware and Software that	
MAN I IGHUIIII	monitors the security of the DIR Network.	
Small Office/Home	A type of Service as specified in Section 4.1 of the CTSA, which may be	
Office (SO/HO)	directly sold to Customers by Successful Respondent as set forth in Article 4	
Internet	Relationship among DIR, Successful Respondent, other TEX-AN 2021	
Connectivity	Contract Holders, and Customers of the CTSA, if awarded to Successful	
Connectivity	Contract Fronces, and Customers of the C15/1, if awarded to successful	

Term	Definition	
	Respondent under this CTSA. Service available only to locations with twenty-	
	five (25) or less Internet users per location.	
Software	All Materials consisting of software programs and programming (and all	
	modifications, replacements, Upgrades, enhancements, documentation,	
	materials and media related thereto), including Antivirus Software,	
	Application Software, Development Tools, and System Software.	
State	The State of Texas.	
Statement of Work	Document used to clarify project requirements and document additional terms,	
(SOW)	conditions, and payment details for services rendered.	
Stop Marketing	Has the meaning set forth in Section 16.2 Tailored Remedies.	
Order		
STS	Share Technology Services.	
Subcontract	Any written agreement between Successful Respondent and other third party	
	to fulfill the requirements of this CTSA.	
Subcontractor	A party to a Subcontract which is not the Successful Respondent.	
Subcontractor	Has the meaning set forth in Section 6.3.5 Subcontractor's Compliance.	
Compliance		
Obligations		
Successful	The entity(ies) awarded a CTSA under this solicitation.	
Respondent		
Successful	A Successful Respondent Network is built and maintained by Successful	
Respondent	Respondent for supporting Customer and DIR Requirements. A Successful	
Network	Respondent Network provides Customer connectivity similar to the DIR	
	Network. Successful Respondent Networks also provide access between	
	Customer Networks and the enterprise services on the DIR Network.	
Successful	People, including employees, agents, and individuals serving as subcontractors	
Respondent	to the Successful Respondent.	
Personnel		
Successful	All software owned by Successful Respondent or its Affiliates and used in	
Respondent's	providing the Services.	
Proprietary		
Software		
Technology Based	Audio, video, and web-based conferencing services available as part of an	
Conferencing	integrated solution, or as a subscription-based solution, but does not include	
Services	the sale of hardware. Conferencing hardware is available through DIR	
TD 1	Cooperative Contracts.	
Telecommunication	Offerings include, but are not limited to, management of Customer owned, or	
Managed Services	Successful Respondent provided Equipment in support of management of call	
(or Telecom	processing architecture, call centers or contact center services, interactive	
Managed Services)	voice response (IVR)/auto-attendant services, managed private branch exchange (PBX) services, management of voice and data networks, network	
	optimization services, and hosted solutions on a subscriptions basis. Services are negotiated through an SOW process and a sample SOW document is	
	located as Appendix A.	
Telecommunications	The Services identified by DIR, or per applicable laws and regulations, as	
Service Priority	Services for Successful Respondents to initiate, restore, or otherwise provide	
(TSP)	on a priority basis to ensure the continued effective operation of State-defined	
	critical Services.	
	Cition Services.	

Term	Definition	
Term	The period of time during which the CTSA is in effect between DIR and the	
	Successful Respondent, as more particularly described in CTSA Article 3	
	<u>Term</u> .	
TEX-AN	The activities and contracts authorized by Chapter 2170, Texas Government	
	Code, administered by DIR on behalf of the State.	
TEX-AN 2021	The final version of this contractually binding agreement between DIR and the	
Communications	Successful Respondent, including all exhibits and attachments thereto.	
<b>Technology Services</b>		
Agreement, or		
(CTSA) TEX-AN Next	Moone the assument TEV AN Dragram as it exists to day	
Generation or TEX-	Means the current TEX-AN Program as it exists today.	
AN NG		
TEX-AN 2021	A contractual document containing the terms, conditions, and scope for	
Customer Services	Services to be deployed and operated under this CTSA, and executed by the	
Agreement (CSA)	Customer, Successful Respondent, and DIR. Attachments H-1 and H-2 contain	
	the template for the form of TEX-AN 2021 Customer Services Agreement.	
Texas.gov	Official State internet portal in accordance with Chapter 2054, Texas	
	Government Code.	
The Electronic	A method for purchasers (governments, institutions, consumers, etc.) to	
Product	evaluate the effect of a product on the environment. It assesses various	
Environmental	lifecycle environmental aspects of a device and ranks products as Gold, Silver,	
Assessment Tool	or Bronze based on a set of environmental performance criteria.	
(EPEAT)		
Transition	Process for the coordinated transfer of any Service from the outgoing contract	
	holder at the termination or expiration of this CTSA to an incoming Successful	
Transition	Respondent.  Are defined as set forth in Section 21.4, Transition Assistance Services.	
Assistance Services	Are defined as set forth in Section 21.4, Transition Assistance Services.	
Transition Effective	Has the meaning set forth in Section 21.1, Scope.	
Date	Thus the meaning set forth in section 21.1, scope.	
Transition Period	Has the meaning set forth in Section 21.2, General Statement for Transition.	
Transition Plan for	The plan for the period of time and activities for the transition of TEX-AN	
<b>Expiration of the</b>	2021 Services under this CTSA from the Successful Respondent to DIR or a	
CTSA (Transition	successor service provider designated by DIR at the end of the Term. See	
Plan)	Article 21 of the CTSA.	
VIS	Vendor Information System.	
Voice	Has the meaning described in Section 3.1, Voice Services of the SOW.	
VPAT	Voluntary Product Accessibility Template.	
WAN	Wide area network.	
WCAG	Web content accessibility guidelines	
Wired	In the context of this RFO, Wired refers to the transmission of data over a	
Windos	wire-based communication technology.	
Wireless	In the context of this RFO, Wireless is the electromagnetic transfer of	
	information between two or more points that are not connected by an electrical	
	conductor.	

— End of Exhibit A —

Vendor Contract No.\_\_\_\_

### Exhibit B To TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Vendor Contract No.	
---------------------	--

# EXHIBIT B. DESCRIPTIONS, RATES TO DIR, PRICES, AND RELATED TELECOMMUNICATIONS FEES AND SURCHARGES FOR AWARDED SERVICES

### 1 Introduction

The purpose of Exhibit B is to address the agreed-upon Descriptions, Rates to DIR, Prices, and related telecommunications fees and surcharges for Awarded Services.

### 2 CONTRACTUAL PRINCIPLES

- a) This Exhibit B addresses the following principles:
- b) The specifics for the Descriptions, Rates to DIR, Prices, and related telecommunications fees and surcharges for Awarded Services within scope of this CTSA are contained in this Exhibit B, Attachment B-1 through Attachment B-4. The Successful Respondent's Descriptions, Rates to DIR, Prices, and fees are consistent with the terms of the CTSA.
- c) The Marketing Plan is the Successful Respondent's projection of the activities it will engage in to maximize the economic value of the CTSA during the Term of the CTSA. The Marketing Plan includes a statement of operations for the awarded Services and the narrative discussion including all key assumptions, risks, and risk mitigation strategies used to generate Purchase Orders from Customers.

### 3 METHODOLOGY FOR UPDATING ASSOCIATED ATTACHMENTS

#### 3.1 Recommendations

Either DIR or Successful Respondent may submit to the other Party a recommendation for changes to Attachment C-1. Such recommendation will be in writing and will:

- a) Specifically identify the portion or portions of the Attachments to be changed,
- b) Include the specific Rates to DIR, Prices, or telecommunications fees and surcharges to be changed and/or Descriptions to be altered in the Attachments, and
- c) Identify the reasons for the proposal and anticipated revenue impact from the change or repercussions for Failure to adopt the change. A copy of the recommendation will be provided to the other Party. The Parties will use electronic communications to submit, review, and implement changes to the Attachments. Attachment changes shall be provided to Customers via the Successful Respondent's webpage and DIR's TEX-AN 2021 Program webpage.

### 3.2 Review and Input

a) DIR will determine whether input on the recommendation from other Customers is appropriate and will promptly circulate the recommendation to those identified. If DIR does request input from any Customer, DIR will require any input to be provided in writing and submitted within a

Vendor Contract No.	
---------------------	--

specified period of time, not to exceed thirty (30) Business Days from the date of the recommendation.

b) DIR and Successful Respondent will each receive a copy of the written input from the Customers, if applicable. The Party making the recommendation may propose adjustments to the recommendation to address any input of the Customers.

### 3.3 Approval

If DIR and Successful Respondent agree upon the recommendation, the Parties will execute a revision to Attachments, as applicable, which will be effective when signed by authorized representatives of both DIR and Successful Respondent. The revised Attachments will be posted in a location agreed to by Successful Respondent and DIR. Revisions may be in the form of electronic records of the new Descriptions, Rates to DIR, Prices, and/ or telecommunications fees and surcharges for changes to Attachment B-1. Successful Respondent shall update its webpage accessible to Customers immediately to have the revised portions of the Attachment B-1 available to Customers as soon as possible.

### 3.4 Appeal

If DIR and Successful Respondent cannot agree on a recommendation, the recommendation may be submitted to the Executive Leadership Team for resolution.

### 4 ATTACHMENTS TO EXHIBIT B

The following attachments are incorporated as part of Exhibit B:

Attachments to Exhibit B	Description and Contents
Attachment B-1 Descriptions, Rates to DIR, Prices, and related telecommunications fees and surcharges for Awarded Services	Attachment B-1 contains the Descriptions, Rates to DIR, Prices, and the telecommunications fees and surcharges for the Awarded Services.
Attachment B-2 Descriptions, Rates to DIR, Prices and related telecommunications fees and surcharges for Carrier Class Services; and only applicable to Successful Respondents that are awarded Carrier Class Services.	Attachment B-2 contains service descriptions, rates to DIR and customer prices and related telecommunications fees, taxes and surcharges related to Carrier Class Services and include the respective administrative fee (12% for services where DIR acts as billing agent, and 4% for direct sales services) for services described therein. Attachment B-2 is attached to the CTSA as referenced here.
Attachment B-3 Descriptions, Rates to DIR, Prices and related telecommunications fees and surcharges for Wireless Services; and only applicable to Successful Respondents that are awarded Wireless Services.	Attachment B-3 contains service descriptions, rates to DIR and customer prices and related telecommunications fees, taxes and surcharges related to Wireless Services and include the respective administrative fee of 2% for services described therein. Attachment B-3 is attached to the CTSA as referenced here.
Attachment B-4 Descriptions, Rates to DIR, Prices and related telecommunications fees and surcharges for Telecom Managed Services; and only applicable to Successful Respondents that are awarded Telecom Managed Services.	Attachment B-4 contains service descriptions, rates to DIR and customer prices and related telecommunications fees, taxes and surcharges related to Telecom Managed Services and include the respective administrative fee of .50% for services described therein. There is no pricing related to Attachment B-4, but the Sample Statement of Work Template for Telecom Managed Services is included as an attachment to the CTSA.

Vendor	Contract No.	
vendor	COHITACLINO.	

### Attachment B-1.

The parties will agree which services at what prices will be contained in the tables in this section.

### 1. Carier Class Services

Includes, but not limited to, Voice Service, local and long distance, VOIP, bundled voice services, managed PBX, and unified communications voice services, data services, including Ethernet, Internet, MPLS, broadband services, and Small Office/Home Office (SOHO) service, next generation 911 services, fiber based technologies, and network power management services. Carrier Class Services awarded under this contract are those specifically stated in Attachment B-2 to Exhibit B.

#### 2. Wireless Services

Includes, but not limited to wireless voice, data and bundled plans, bundled mobile device and network management services, intelligent community services, wireless business services, microwaves, satellite voice plans, wireless WAN, and wireless broadband. Wireless Services awarded under this contract are those specifically stated in Attachment B-3 to Exhibit B

### 3. Telecom Managed Services

Includes, but not limited to, managed or hosted voice and data network solutions, network optimization analysis and review, network project management, call attendant, IVR, and relate services as detailed by customers through the Statement of Work Template and bid process.

Pricing for services awarded under this contract can be found in Attachment B-2, and Attachment B-3.

— End of Exhibit B —

Vendor Contract No.\_\_\_\_

### TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Vendor Contract No.	
---------------------	--

### EXHIBIT C. PERFORMANCE MANAGEMENT

This Exhibit C is applicable to the Carrier Class Servies of this contract DIR-TELE-CTSA-007.

This Exhibit C is not applicable to the Wireless Services nor the Telecom Managed Services of this contract DIR-TELE-CTSA-007.

### 1 Introduction

The purpose of Exhibit C is to comprehensively address the agreed-upon Performance Management for delivery of awarded Services.

### 2 Performance Management Contractual Principles

Exhibit C Performance Management addresses the following principles:

### 2.1 Credit Assessments

DIR has chosen to address a majority of the risks related to Successful Respondent performance through the assessment of Credits against Rates to DIR and Prices for Services. The following provisions form the basis for assessing Credits as described in the CTSA:

- a) If the Successful Respondent does not fulfill its Services as specified under the CTSA, DIR and the Customers will be damaged;
- b) Establishing the precise amount or value of such damage may be difficult to quantify; and
- c) Except as otherwise provided in the CTSA, Credits will start to accumulate immediately following the end of the Service billing period in which an applicable SLA was not met.

### **2.2** Service Level Agreements (SLAs)

- a) Successful Respondent will comply with the performance criteria outlined in this Exhibit C.
- b) For each Service, Successful Respondent and DIR have agreed on acceptable SLAs and acceptable Credits associated with failure to deliver on each SLA, as set forth in Service Level Agreement (SLA) Plan. Successful Respondent will provide continuous improvement of the SLAs over the life of the CTSA. DIR and the Successful Respondent will review all of the SLAs on an annual basis.
- c) All proposed plans and mechanisms to be used for reporting shall be subject to approval by DIR, and all approved SLAs will be governed by the CTSA, as well as any special provisions agreed upon by DIR, Customer, and the Successful Respondent with respect to each SLA and/or CSA.

### 2.3 Reporting

a) Failure of the Successful Respondent to submit the Management Plans required in Exhibit D Plans and Reports required in Attachment C-1 may result in Credits in the amount of \$500 per day after a five (5) day cure period. The same Credits also apply to Management Plans that are changed due to any future updates or revision requests that may be required by DIR. A revised plan addressing changes recommended by DIR must be submitted to DIR within five (5) Business Days of receiving notice from DIR. Changes to Management Plans will be made in accordance

Vendor Contract No.	
---------------------	--

with Exhibit D Plans, Article 3 Methodology for Updating Associated Exhibit Attachments and Plans.

b) Due dates for the initial drafts and final drafts of the Management Plans are specified in Exhibit D Plans.

#### 2.4 Service Instance Performance

- a) The Successful Respondent will measure SLAs per Service as defined below, and report on performance as described in the CTSA.
- b) Per Service, as applicable:
  - 1) Mean Time to Repair
  - 2) Service Availability
  - 3) Latency
  - 4) Packet Delivery Rate
  - 5) Jitter

### 2.5 Chronic Outage Reports

Tailored remedies as described in the CTSA, may be used by DIR if the Successful Respondent fails to produce and submit all required reports as documented by the schedule of reports agreed to by the Parties pursuant to the Master Vendor Reporting Guide and stated in this Attachment C-1. Successful Respondent will not be liable for any failure or delay in performing its obligations under this Section if such failure or delay is due to Force Majeure Events. Additional reporting may be required by DIR in accordance with the CTSA.

### **2.6** Reporting on Performance Management

All plans and mechanisms to be used for reporting shall be subject to approval by DIR.

### 3 METHODOLOGY FOR UPDATING ASSOCIATED ATTACHMENT C-1

Attachment C-1 sets forth the actual reports related to Performance Management due on a monthly basis and as related to Successful Respondent business functions.

### 3.1 Recommendations

Either DIR or Successful Respondent may submit to the other Party a recommendation for changes to reports listed in Attachment C-1. Such recommendation will be in writing and will:

- a) Specifically identify the portion or portions of the Master Vendor Reporting Guide to be changed,
- b) Include the specific information to be changed and/or descriptions to be altered, and
- c) Identify the reasons for the proposal and anticipated revenue impact from the change or repercussions for failure to adopt the change. A copy of the recommendation will be provided to the other Party. The parties will use electronic communications to submit, review, and implement changes to Master Vendor Reporting Guide and Attachment C-1. Attachment C-1 changes shall be provided to Customers via DIR's contract landing webpage.

Vendor	Contract No.	

### 3.2 Review and Input

- a) DIR will determine whether input on the recommendation from other Customers is appropriate and will promptly circulate the recommendation to those identified. If DIR does request input from any Customer, DIR will require any input to be provided in writing and submitted within a specified period of time, not to exceed thirty (30) Business Days from the date of the recommendation.
- b) DIR and Successful Respondent will each receive a copy of the written input from the Customers, if applicable. The Party making the recommendation may propose adjustments to the recommendation to address any input of the Customers.

### 3.3 Approval

If DIR and Successful Respondent agree upon the recommendation, the Parties will execute a revision to Attachment C-1 or Master Vendor Reporting Guide, as applicable, which will be effective when signed by authorized representatives of both DIR and Successful Respondent. The revised Attachment C-1 will be posted in a location agreed to by Successful Respondent and DIR. Revisions may be in the form of electronic records of the new, or updated reports related to Attachment C-1 or the Master Vendor Reporting Guide.

### 3.4 Appeal

If the DIR and Successful Respondent cannot agree on a recommendation, the recommendation may be submitted to the Executive Leadership Team for resolution.

### 4 DOCUMENTS REFERENCED IN EXHIBIT C PERFORMANCE MANAGEMENT

### 4.1 Attachments to Exhibit C

Attachment C-1 is incorporated herein as part of Exhibit C Performance Management.

Vendor Contract No.\_\_\_\_\_

### Attachment C-1. Tables reflecting Performance Measurement Reports due on a monthly basis

The tables in Attachment C-1 set forth the actual Performance Measurement Reports as related to Successful Respondent business functions.

- (1) Change/Release Management Report.
- (2) After Action Reports,
- (3) Request for Change Report.
- (4) Trouble Ticket Aging Report
- (5) SLA Non-Compliance Report
- (6) Outbound CDR Report
- (7) Inbound CDR
- (9) Marketing Report
- (10) Invoice Summary Report
- (11) Status for Projects by Customer Report
- (12) HUB Subcontractor Report,
- (13) Direct Sales Transactions Reports
- (14) Monthly Recurring Charge (MRC) Report
- (15) Non-Recurring Charge (NRC) Report
- (16) Adjustments/Credits Report
- (17) Long Distance (LD) Usage

— End of Exhibit C —

Vendor Contract No.\_\_\_\_

### TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Exhibit D. Plans
D.1

Vendor	Contract No.	

### EXHIBIT D. PLANS

This Exhibit D is applicable to the Carrier Class Services of this contract DIR-TELE-CTSA-007.

With respect to the Wireless Services and Telecom Managed Services portion of this contract, DIR-TELE-CTSA-007, this Exhibit D only applies to the Marketing Plan, which is required of all Successful Respondents.

### 1 Introduction

The purpose of Exhibit D Plans is to comprehensively address agreed-upon CTSA Management Plans related to services where DIR acts as the Successful Respondent's billing agent, with the exception of the Marketing Plan which is required of all Successful Respondents.

### 2 REQUIREMENTS

- a) The Successful Respondent is responsible for Services delivered under the CTSA.
- b) The Successful Respondent shall demonstrate strong project management practices.
- c) The Successful Respondent shall create, maintain, implement, and update Management Plans in accordance with this Exhibit D.
- d) Initial and final drafts of the Management Plans are to be submitted to DIR for approval within the timeframes specified in Table 1 of Section 4.2, Plans and Timelines in this Exhibit D.
- e) The Successful Respondent will provide updated Management Plans annually or as requested by DIR.
- f) The Management Plans and Services to be provided by Successful Respondent are included in Attachment D-1 to Exhibit D:

### 3 METHODOLOGY FOR UPDATING ASSOCIATED PLANS

#### 3.1 Recommendations

Either DIR or Successful Respondent may submit to the other Party a recommendation for changes or additions to Plans. Such recommendation will be in writing and will:

- a) Specifically identify the portion or portions of the plan to be changed,
- b) Include the specific language to accomplish the proposed change, and
- c) Identify the reasons for the proposal and anticipated improvements from the change or repercussions for failure to adopt the change. A copy of the recommendation will be provided to the other Party.

### 3.2 Review and Input

a) DIR will determine whether input on the recommendation from any other Customers is appropriate and will promptly circulate the recommendation to those identified. If DIR does request input from any other Customers, DIR will require any input to be provided in writing and

Exhibit D. Plans D.2

Vendor Contract No.	
---------------------	--

submitted within a specified period of time, not to exceed sixty (60) Business Days from the date of the recommendation.

b) DIR and Successful Respondent will each receive a copy of the written input from the other Customers, if applicable. The Party making the recommendation may propose adjustments to the recommendation to address any input of the other Customers.

### 3.3 Approval

- a) If DIR and Successful Respondent agree upon the recommendation, the Parties will execute a revision to the Plan, as applicable, which will be effective when signed by authorized representatives of both DIR and Successful Respondent and the revised Plan will be posted in a location agreed to by Successful Respondent and DIR.
- b) An approved Plan revision resulting in a substantial change to CTSA operations may trigger a formal amendment process.

### 3.4 Appeal

If the DIR and Successful Respondent cannot agree on a recommendation, the recommendation may be submitted to the Executive Leadership Team for resolution.

### 4 DOCUMENTS REFERENCED IN EXHIBIT D PLANS

### 4.1 Attachments to Exhibit D

Once accepted by DIR, the final version of the Exhibit D attachments will be incorporated as part of Exhibit D Plans.

### **4.2** Exhibit D Plans

The following plans are associated with Exhibit D Plans:

4.2.1 Attachments D-1 through D-7 to be determined by the Parties and posted to the contract landing page on the DIR website.

Plan #	Management Plan Name
D1	Program Management Plan
D2	Marketing Plan
D3	Service Delivery & Order Process Plan
D4	Help Desk Plan
D5	Change Management Plan
D6	Service Level Agreement (SLA) Plan
D7	Billing Plan

— End of Exhibit D —

Exhibit D. Plans D.3

### TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Exhibit E. Insurance E.1

Vendor	Contract No.	
v chuon	COMMACLING.	

### EXHIBIT E. INSURANCE

### 1 Introduction

The purpose of Exhibit E Insurance is to address the Successful Respondent insurance requirements for the CTSA.

### 2 INSURANCE CONTRACTUAL PRINCIPLES

- a) Successful Respondent agrees to keep in full force and effect and maintain at its sole cost and expense the following policies of insurance with the specified minimum limits of liability during the term of this CTSA and shall provide proof of such insurance coverage to DIR within five (5) Business Days following the execution of the CTSA. Successful Respondent may not begin performance under the CTSA until such proof of insurance coverage is provided to, and approved by, DIR, in its sole and absolute discretion. All required insurance must be issued by companies whose A.M. Best Company Financial Strength Ratings are at least A financially rated and duly licensed, admitted, and authorized to do business in the State. DIR will be named as Additional Insured on all required coverage (except for Worker's Compensation coverage). Notwithstanding the foregoing, the parties agree that Successful Respondent's carrier(s) for non-Worker's Compensation and non-Business Automobile Liability coverages may be a company other than one admitted in the State, so long as the company has a minimum A.M. Best Company Financial Strength Rating of A- and complies with all other requirements stated herein. The minimum acceptable insurance provisions are as follows:
  - 1) Commercial General Liability: Commercial General Liability must include a combined single limit of \$500,000 per occurrence for coverage A, B, & C including products/completed operations, where appropriate, with a separate aggregate of \$500,000. The policy shall contain the following provisions:
    - i. Blanket contractual liability coverage for liability assumed under the CTSA;
    - ii. Independent Contractor coverage;
    - iii. State, DIR and Customer listed as an additional insured;
    - iv. Thirty (30)-day Notice of Termination in favor of DIR and/or Customer; and
    - v. Waiver of Transfer Right of Recovery Against Others in favor of DIR and/or Customer.
  - 2) Workers' Compensation Insurance: Workers' Compensation Insurance and Employers' Liability coverage must include limits consistent with statutory benefits outlined in the Texas Workers' Compensation Act (Art. 8308-1.01 et seq. Tex. Rev. Civ. Stat) and minimum policy limits for Employers' Liability of \$250,000 bodily injury per accident, \$500,000 bodily injury disease policy limit, and \$250,000 per disease per employee.
  - 3) Business Automobile Liability Insurance: Business Automobile Liability Insurance must cover all owned, non-owned, and hired vehicles with a minimum combined single limit of \$500,000 per occurrence for bodily injury and property damage. Alternative acceptable limits are \$250,000 bodily injury per person, \$500,000 bodily injury per occurrence and at least \$100,000 property damage liability per accident. The policy shall contain the following endorsements in favor of DIR and/or Customer:

Exhibit E. Insurance E.2

Vendor Contract No.

- i. Waiver of Subrogation;
- ii. 30-day Notice of Termination; and
- iii. Additional Insured.
- 4) Professional Liability (also known as Errors and Omissions Liability) and, as and to the extent becoming commercially available, Cyber Risk Insurance covering acts, errors and omissions arising out of Successful Respondent's operations or Services. The per claim and policy aggregate amount(s) shall be negotiated by the Parties.
- b) Similar coverage as described above shall be required from Successful Respondent in each instance in which it agrees to perform work related to Services at a Customer's premises. A certificate of insurance demonstrating compliance with these coverages and listing the Customer as the additional Insured shall be provided to the Customer prior to Successful Respondent commencing work at a Customer's premises.

### 3 METHODOLOGY FOR UPDATING ASSOCIATED EXHIBIT E ATTACHMENTS AND PLANS

There is no need to update Attachment E-1 Successful Respondent's Certificate of Insurance. Any changes to insurance requirements will be made at the exhibit level.

### 4 DOCUMENTS REFERENCED IN EXHIBIT E INSURANCE

### **4.1** Exhibit E Insurance Attachments

In Attachment E-1 Successful Respondent's Certificate of Insurance, Successful Respondent has included its Certificate of Insurance, which sets forth the Successful Respondent's insurance coverage.

Attachments to Exhibit E	Description	
Attachment E-1 Successful Respondent's Certificate of Insurance	Successful Respondent's Certificate of Insurance	

### 4.2 Attachment E-1. Successful Respondent's Certificate of Insurance

— End of Exhibit E —

Exhibit E. Insurance E.3

Vendor Contract No.\_\_\_\_

### TEX-AN 2021 Communications Technology Services Agreement

Between

The State of Texas, acting by and through the Texas Department of Information Resources

and

**Hughes Network Systems, LLC (Successful Respondent)** 

Vendor	Contract N	ĺo	
vendor	COHERCEN	O.	

## EXHIBIT F. FORM OF TEX-AN 2021 CUSTOMER SERVICES AGREEMENT

### 1 Introduction

The purpose of *Exhibit F Form of TEX-AN 2021 Customer Services Agreement* is to address the agreed-upon form of the CSA.

### 2 CONTRACTUAL PRINCIPLES

- a) Attachment F-1 TEX-AN 2021 Customer Service Agreement is the template for the Form of the CSA among DIR and Customer.
- b) Attachment F-1 TEX-AN 2021 Customer Service Agreement is a template which will be the basis of negotiations with Customers, and contains the terms and conditions between and among the Customer and DIR. The terms and conditions of a CSA may be negotiated to meet Customer business needs, so long as the CSA terms and conditions do not conflict with or weaken the terms of the CTSA.

### 3 METHODOLOGY FOR UPDATING ASSOCIATED EXHIBIT ATTACHMENT

#### 3.1 Recommendations

In addition to negotiations with Customers to reach an acceptable CSA, DIR or Customer may submit to the other Party a recommendation for changes or additions to Attachment F-1 TEX-AN 2021 Customer Service Agreement. Such recommendation will be in writing and will:

- a) Specifically identify the portion or portions of the attachment requested to be changed,
- b) Include the specific language to accomplish the proposed change, and

Identify the reasons for the proposal and anticipated improvements from the change or repercussions for failure to adopt the change.

### 4 DOCUMENTS REFERENCED IN EXHIBIT F FORM OF TEX-AN 2021 CUSTOMER SERVICES AGREEMENT

The following attachments are incorporated as part of Exhibit F Form of TEX-AN 2021 Customer Services Agreement:

Attachments to Exhibit F	Description and Contents
Attachment F-1 Form of TEX- AN 2021 Customer Services Agreement	Attachment F-1 Form of TEX-AN 2021 Customer Services Agreement provides the agreement to be executed by authorized representatives of the Customer and DIR to maximize standardization of the terms and conditions and provide DIR the opportunity for oversight of the contractual relationship between Customer and DIR.

Attachment B-2 to Exhibit B

Hughes Network Systems, LLC

Contract DIR-TELE-CTSA-007

#### **Pricing Notes:**

- 1. Pricing does not include any taxes or government applicable fees
- **2.** Pricing is based on the specified service term per site (Site Service Term).
- **3.** Customer acknowledges that an underlying carrier may discontinue service after installation. In such event, Hughes will provide customer notice and the opportunity to select an alternative service from the table above, if available. If no alternative service is available, Customer may terminate the service at affected sites with thirty (30) day notice without penalty.
- **4.** Customer acknowledges that services are based on availability and Hughes cannot guarantee that any given wireline/wireless service is available for a site, regardless of prequalification results. In the event that the prequalified service is not available at any site, Customer may select a different broadband service.
- **5.** There may be additional installation costs, such as additional inside wiring charges, jacks, filters, etc., which may be assessed by the applicable telecom or cable companies (ISP). A site survey will be performed by the ISP and Hughes will notify the Customer of buildout charges that may apply as other direct costs. Hughes will only proceed after receiving permission from the Customer.
- **6.** Pricing assumes that inside wiring between the location where the wireline circuit enters the building and the Customer office where the modem is located is already in place. Consequently, no inside wiring costs are included in the pricing listed above. If such inside wiring is required, additional charges will apply.
- 7. When Hughes places an order for a wireline circuit with an underlying carrier, the order is placed for a circuit with a particular service plan based on speed. Those speeds are based on what is anticipated to be the rate at which the modem will synchronize with the circuit. The speeds are "up to" speeds, as the modem may sync at a lower speed due to line conditions. Consequently, circuit speeds are not guaranteed. Additionally, for traffic carried within a secure tunnel across the circuit, the effective throughput will be reduced due to the packet and encryption overhead. The performance experienced by a user will also vary depending on potential areas of congestion within the general Internet, server loading, and application specific performance.
- **8.** Customer acknowledges that Hughes' suppliers require a minimum site service term, as identified in the table below, to receive the pricing set forth in this price schedule. Customer will be responsible for the following early termination charges not to exceed those listed below in the event that Customer terminates for convenience.

Circuit Type	Early Termination Charges				
ADSL, IDSL	Up to \$150 per terminated site (waived after site completes 12 months of service)				
Dedicated DSL	\$99 (waived after site completes 24 months of service)				
Cable (Charter/Ti me Warner/Sp ectrum)	100% of remaining charges (waived after site completes 12 months of service)				
Cable (Cox)	\$100 per terminated site (waived after site completes 36 months of service)				
Cable (Comcast/S uddenlink/ Mediacom)	100% of remaining service charges (waived after site completes 36 months of service)				
Cable (Optimum/ Rogers)	\$50 (waived after site completes 36 months of service)				
Cellular Wireless	\$200 per terminated site (waived after site completes 12 months of service)				
FiOS	\$320 per terminated site				
FTTI, Bell Canada Bell Fibe	\$50 per terminated site				
Granite Grid, Spectrum Fiber	100% of remainder of service charges				
U- Verse/IPBB	\$199 per terminated site (waived after site completes 36 months of service)				
Asymmetri c/Symmetr ic Ethernet	100% of remainder of service charges, plus Customer is responsible to return the underlying provider's equipment. Upon ordering decommissioning of an Asymmetric Ethernet site, either Hughes or the provider will send Customer a postage prepaid label and box to be used for				
Fixed Wireless	100% of remainder of service charges for first 12 months (if terminated in first 12 months), plus 45% of remaining service charges for months 13-36. In				

**9.** If a technician is dispatched to fix a reported problem at a site, and it is determined that the problem is not related to the broadband circuit, a \$250 site visit charge applies for each visit if multiple visits are required. Customer shall provide to Hughes a monthly report of dispatches authorized by Customer. If a Provider technician must go to site, there will be a Time and Material (T&M) charge.

- **10.** Time and Material (T&M) charges also apply to Hughes site visits, including but not limited to:
- · For out of scope items such as Customer damage to the equipment, force majure, or disconnected cables.
- The T&M charge is the sum of the "Hourly Rate" plus the "Travel Zone Charge" shown below. The Travel Zone Charge is charged for each direction of travel. Material is charged at cost plus 20%.

Time Work is Performed	Hourly Rate
Normal Hours (8:00 AM – 5:00 PM, Monday through Friday)	\$125/hour
After Hours Weekday (5:00 PM – 8:00 AM, Monday through Thursday)	\$150/hour
Weekend (5:00 PM Friday – 8:00 AM Monday)	\$150/hou r

Distance	Travel Zone Charge			
0 - 25	\$77			
miles	Φ//			
26 - 50	\$143			
miles	φ143			
51 - 100	\$286			
miles	Ψ260			
101+ miles	\$473			
101+ IIIIes	φ+73			

- **11**. Pricing does not include Backhauls and associated terminating equipment. It assumes that Customer will provide that.
- **12.** All OCONUS Satellite survey and installation are ICB, excluded from quoted price for each site. Site survey is required to provide accurate price for survey and installation.
- **13.** For VSAT sites (if any), site installation price is based on standard installation. For any site for which a complex installation is identified, a separate incremental complex installation quote will be provided to the Customer. This charge will be added on top of the standard installation charge.
- **14**. 4G xxMbps/xxGB service plans offer flat rate 4G service without overage charge. Throttling starts when plan threshold is exceeded. Service plan subject to coverage availability. Service has restriction on video/audio streaming, surveillance and Wi-Fi support.
- 15. For broadband Locations identified by Hughes as requiring a site survey, Hughes will perform a site survey to identify technically-suitable Locations for installing the Equipment and cables. Hughes works with its providers to negotiate construction cost waivers whenever possible, in consideration of a longer-term commitment (typically 36 months per site). If a site survey shows that no cable construction costs are required, and Customer elects to not proceed with the cable installation, HNS will charge the Customer \$250 per site survey performed. For Locations not receiving a cable construction waiver, if a site survey shows that cable construction costs are required, and Customer elects to not proceed with the cable installation, no site survey charge will apply.
- 16. Following charges apply to all services installed in an enclosed shopping mall or Multi Story Building.

Site Survey and Report	\$350/each	All Services in an enclosed shopping mall or Multi Story Building
Mall and multi-story building installation surcharge	\$300/each	All Services in an enclosed shopping mall or Multi Story Building

- 17. Customer acknowledges that HNS' underlying circuit providers have reserved the right to determine their service areas, and the right to maintain, reconfigure, adjust pricing or discontinue any service and/or product. In the event that HNS's suppliers make any such changes that prohibit HNS from continuing to provide circuits or to provide circuits at previous price levels, HNS will provide Customer a new quote for the affected circuit.
- **18.** Pricing is valid for 90 days

### PRICING WITH DIR ADMINISTRATIVE FEE (CUSTOMER PAYS THIS AMOUNT)

Geographical Area: Texas

### **VOIP SERVICES PRICING**

Services below line are billed directly by Vendor with a 4% admine fee.

Voice services listed here are only available in conjunction with SOHO services, and not available as a stand alone service.

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
HughesNet Voice Plans	\$31.15	Unlimited local and long-distance calling, In the domestic U.S. and Canada PLUS two optional calling plans for International calls to the most frequently called countries FREE premium features Caller ID, Call Waiting, Voicemail and more No phone number change required Keep your same phone number, or choose a new one** https://legal.hughesnet.com/BusinessVoice SubAgree.cfm	\$78.00	Analog telephone adaptor (ATA)	12 months
HughesNet Voice Plans	\$31.15	Unlimited local and long-distance calling, In the domestic U.S. and Canada PLUS two optional calling plans for International calls to the most frequently called countries FREE premium features Caller ID, Call Waiting, Voicemail and more No phone number change required Keep your same phone number, or choose a new one** https://legal.hughesnet.com/BusinessVoice SubAgree.cfm	\$0.00	Analog telephone adaptor (ATA)	24 months
Option; International 200	\$10.35	Includes 200 minutes of calling to 60 countries. Applicable countries can be found on legal.HughesNet.com			
Option; International Unlimited	\$23.87	Includes unlimited calling to 60 countries. Applicable countries can be found on legal.HughesNet.com			

### PRICING WITH DIR ADMINISTRATIVE FEE (CUSTOMER PAYS THIS AMOUNT)

**Geographical Area: Texas** 

### **INTERNET SERVICES PRICING**

Services below this line are administered and billed by DIR with a 12% admin fee

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
ADSL Provider1 (Up to 1.5M/640k)	\$57.12	ADSL Provider1 (Up to 1.5M/640k)	\$201.60	ADSL Provider1 (Up to 1.5M/640k)	24 months
ADSL Provider1 (Up to 1.5M/896k)	\$57.12	ADSL Provider1 (Up to 1.5M/896k)	\$201.60	ADSL Provider1 (Up to 1.5M/896k)	24 months
ADSL Provider1 (Up to 2M/1.5M)	\$78.40	ADSL Provider1 (Up to 2M/1.5M)	\$201.60	ADSL Provider1 (Up to 2M/1.5M)	24 months
ADSL Provider1 (Up to 3M/896k)	\$75.04	ADSL Provider1 (Up to 3M/896k)	\$201.60	ADSL Provider1 (Up to 3M/896k)	24 months
ADSL Provider1 (Up to 3M/2M)	\$78.40	ADSL Provider1 (Up to 3M/2M)	\$201.60	ADSL Provider1 (Up to 3M/2M)	24 months
ADSL Provider1 (Up to 3M/3M)	\$78.40	ADSL Provider1 (Up to 3M/3M)	\$201.60	ADSL Provider1 (Up to 3M/3M)	24 months
ADSL Provider1 (Up to 3M/5M)	\$78.40	ADSL Provider1 (Up to 3M/5M)	\$201.60	ADSL Provider1 (Up to 3M/5M)	24 months
ADSL Provider1 (Up to 5M/896k)	\$75.04	ADSL Provider1 (Up to 5M/896k)	\$201.60	ADSL Provider1 (Up to 5M/896k)	24 months
ADSL Provider1 (Up to 5M/1.5M)	\$78.40	ADSL Provider1 (Up to 5M/1.5M)	\$201.60	ADSL Provider1 (Up to 5M/1.5M)	24 months
ADSL Provider1 (Up to 5M/3M)	\$78.40	ADSL Provider1 (Up to 5M/3M)	\$201.60	ADSL Provider1 (Up to 5M/3M)	24 months
ADSL Provider1 (Up to 7M/896k)	\$75.04	ADSL Provider1 (Up to 7M/896k)	\$201.60	ADSL Provider1 (Up to 7M/896k)	24 months
ADSL Provider1 (Up to 10M/640k)	\$96.32	ADSL Provider1 (Up to 10M/640k)	\$201.60	ADSL Provider1 (Up to 10M/640k)	24 months
ADSL Provider1 (Up to 10M/1.5M)	\$100.80	ADSL Provider1 (Up to 10M/1.5M)	\$201.60	ADSL Provider1 (Up to 10M/1.5M)	24 months
ADSL Provider1 (Up to 10M/2M)	\$100.80	ADSL Provider1 (Up to 10M/2M)	\$201.60	ADSL Provider1 (Up to 10M/2M)	24 months
ADSL Provider1 (Up to 10M/3M)	\$100.80	ADSL Provider1 (Up to 10M/3M)	\$201.60	ADSL Provider1 (Up to 10M/3M)	24 months
ADSL Provider1 (Up to 12M/896k)	\$96.32	ADSL Provider1 (Up to 12M/896k)	\$201.60	ADSL Provider1 (Up to 12M/896k)	24 months
ADSL Provider1 (Up to 12M/2M)	\$100.80	ADSL Provider1 (Up to 12M/2M)	\$201.60	ADSL Provider1 (Up to 12M/2M)	24 months
ADSL Provider1 (Up to 12M/5M)	\$100.80	ADSL Provider1 (Up to 12M/5M)	\$201.60	ADSL Provider1 (Up to 12M/5M)	24 months
ADSL Provider1 (Up to 18M/896k)	\$126.56	ADSL Provider1 (Up to 18M/896k)	\$201.60	ADSL Provider1 (Up to 18M/896k)	24 months
ADSL Provider1 (Up to 18M/1.5M)	\$131.04	ADSL Provider1 (Up to 18M/1.5M)	\$201.60	ADSL Provider1 (Up to 18M/1.5M)	24 months
ADSL Provider1 (Up to 18M/2M)	\$131.04	ADSL Provider1 (Up to 18M/2M)	\$201.60	ADSL Provider1 (Up to 18M/2M)	24 months
ADSL Provider1 (Up to 18M/3M)	\$138.88	ADSL Provider1 (Up to 18M/3M)	\$201.60	ADSL Provider1 (Up to 18M/3M)	24 months
ADSL Provider2 (Up to 1.5M/500k)	\$59.36	ADSL Provider2 (Up to 1.5M/500k)	\$386.40	ADSL Provider2 (Up to 1.5M/500k)	24 months
ADSL Provider2 (Up to 1.5M/512k)	\$59.36	ADSL Provider2 (Up to 1.5M/512k)	\$386.40	ADSL Provider2 (Up to 1.5M/512k)	24 months
ADSL Provider2 (Up to 3M/500k)	\$59.36	ADSL Provider2 (Up to 3M/500k)	\$386.40	ADSL Provider2 (Up to 3M/500k)	24 months
ADSL Provider2 (Up to 3M/640k)	\$59.36	ADSL Provider2 (Up to 3M/640k)	\$386.40	ADSL Provider2 (Up to 3M/640k)	24 months
ADSL Provider2 (Up to 4M/1M)	\$59.36	ADSL Provider2 (Up to 4M/1M)	\$386.40	ADSL Provider2 (Up to 4M/1M)	24 months
ADSL Provider2 (Up to 5M/769k)	\$59.36	ADSL Provider2 (Up to 5M/769k)	\$386.40	ADSL Provider2 (Up to 5M/769k)	24 months
ADSL Provider2 (Up to 5M/896k)	\$59.36	ADSL Provider2 (Up to 5M/896k)	\$386.40	ADSL Provider2 (Up to 5M/896k)	24 months

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
ADSL Provider2 (Up to	\$59.36	ADSL Provider2 (Up to 5M/5M)	\$386.4	O ADSL Provider2 (Up to	24
5M/5M)	,	(2)	, , , , ,	5M/5M)	months
ADSL Provider2 (Up to 6M/500k)	\$59.36	ADSL Provider2 (Up to 6M/500k)	\$386.4	ADSL Provider2 (Up to 6M/500k)	24 months
ADSL Provider2 (Up to 7M/896k)	\$59.36	ADSL Provider2 (Up to 7M/896k)	\$386.4	ADSL Provider2 (Up to 7M/896k)	24 months
ADSL Provider2 (Up to 7M/5M)	\$59.36	ADSL Provider2 (Up to 7M/5M)	\$386.4	ADSL Provider2 (Up to 7M/5M)	24 months
ADSL Provider2 (Up to 10M/750k)	\$59.36	ADSL Provider2 (Up to 10M/750k)	\$386.4		24 months
ADSL Provider2 (Up to 10M/896k)	\$59.36	ADSL Provider2 (Up to 10M/896k)	\$386.4		24 months
ADSL Provider2 (Up to 10M/1M)	\$59.36	ADSL Provider2 (Up to 10M/1M)	\$386.4		24 months
ADSL Provider2 (Up to 10M/10M)	\$59.36	ADSL Provider2 (Up to 10M/10M)	\$386.4		24 months
ADSL Provider2 (Up to 12M/896k)	\$59.36	ADSL Provider2 (Up to 12M/896k)	\$386.4		24 months
ADSL Provider2 (Up to 12M/5M)	\$59.36	ADSL Provider2 (Up to 12M/5M)	\$386.4		24 months
ADSL Provider2 (Up to 15M/750k)	\$59.36	ADSL Provider2 (Up to 15M/750k)	\$386.4		24 months
ADSL Provider2 (Up to 20M/896k)	\$71.68	ADSL Provider2 (Up to 20M/896k)	\$386.4		24 months
ADSL Provider2 (Up to 20M/1.5M)	\$71.68	ADSL Provider2 (Up to 20M/1.5M)	\$386.4		24 months
ADSL Provider2 (Up to 20M/2M)	\$71.68	ADSL Provider2 (Up to 20M/2M)	\$386.4		24 months
ADSL Provider2 (Up to 20M/5M)	\$71.68	ADSL Provider2 (Up to 20M/5M)	\$386.4		24 months
ADSL Provider2 (Up to 30M/1.5M)	\$71.68	ADSL Provider2 (Up to 30M/1.5M)	\$386.4		24 months
VDSL Provider1 (Up to 4M/1M)	\$42.56	VDSL Provider1 (Up to 4M/1M)	\$201.6		24 months
VDSL Provider1 (Up to 5M/896k)	\$42.56	VDSL Provider1 (Up to 5M/896k)	\$201.6		24 months
VDSL Provider1 (Up to 10M/1M)	\$120.96	VDSL Provider1 (Up to 10M/1M)	\$201.6		24 months
VDSL Provider1 (Up to 20M/1.5M)	\$56.00	VDSL Provider1 (Up to 20M/1.5M)	\$632.8		24 months
VDSL Provider1 (Up to 20M/2M)	\$120.96	VDSL Provider1 (Up to 20M/2M)	\$201.6		24 months
VDSL Provider1 (Up to 25M/2M)	\$136.64	VDSL Provider1 (Up to 25M/2M)	\$201.6		24 months
VDSL Provider1 (Up to 30M/1.5M)	\$136.64	VDSL Provider1 (Up to 30M/1.5M)	\$201.6	·	24 months
VDSL Provider1 (Up to 40M/2M)	\$64.96	VDSL Provider1 (Up to 40M/2M)	\$593.6		24
VDSL Provider1 (Up to 40M/3M)	\$64.96	VDSL Provider1 (Up to 40M/3M)	\$593.6	·	months 24 months
VDSL Provider1 (Up to	\$152.32	VDSL Provider1 (Up to 40M/5M)	\$201.6	O VDSL Provider1 (Up to	24
VDSL Provider1 (Up to 40M/75M)	\$152.32	VDSL Provider1 (Up to 40M/75M)	\$201.6	` .	months 24
VDSL Provider1 (Up to	\$200.48	VDSL Provider1 (Up to 50M/10M)	\$201.6	` '	months 24
VDSL Provider1 (Up to	\$200.48	VDSL Provider1 (Up to 60M/12M)	\$201.6	` .	months 24
VDSL Provider1 (Up to	\$264.32	VDSL Provider1 (Up to 80M/8M)	\$201.6	` '	months 24
80M/8M)  VDSL Provider1 (Up to	\$264.32	VDSL Provider1 (Up to 100M/9M)	\$201.6	` .	months 24
VDSL Provider1 (Up to	\$296.80	VDSL Provider1 (Up to 200M/6M)	\$201.6	` '	months 24
VDSL Provider1 (Up to	\$296.80	VDSL Provider1 (Up to 300M/7M)	\$201.6	` .	months 24
300M/7M)				300M/7M)	months

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
VDSL Provider1 (Up to	\$344.96	VDSL Provider1 (Up to 400M/4M)	\$201.60	VDSL Provider1 (Up to	24
400M/4M) VDSL Provider1 (Up to	\$344.96	VDSL Provider1 (Up to 500M/5M)	\$201.60	400M/4M)  VDSL Provider1 (Up to	months 24
500M/5M)	Ψ0 1 1.00	, , , , , , , , , , , , , , , , , , ,		500M/5M)	months
VDSL Provider1 (Up to 1000M/3M)	\$408.80	VDSL Provider1 (Up to 1000M/3M)	\$201.60	VDSL Provider1 (Up to 1000M/3M)	24 months
VDSL Provider1 (Up to	\$126.56	VDSL Provider1 (Up to 20M/896k)	\$201.60	VDSL Provider1 (Up to	24
20M/896k)	C424.04	VDCL Dravidary (Lla ta 2004/004)	C004 C0	20M/896k)	months
VDSL Provider2 (Up to 20M/2M)	\$131.04	VDSL Provider2 (Up to 20M/2M)	\$201.60	VDSL Provider2 (Up to 20M/2M)	24 months
VDSL Provider2 (Up to	\$131.04	VDSL Provider2 (Up to 20M/5M)	\$201.60	VDSL Provider2 (Up to	24
20M/5M) VDSL Provider2 (Up to	\$138.88	VDSL Provider2 (Up to 20M/20M)	\$201.60	20M/5M) VDSL Provider2 (Up to	months 24
20M/20M)	Ψ100.00	<u> </u>	ψ201.00	20M/20M)	months
VDSL Provider2 (Up to 35M/50M)	\$156.80	VDSL Provider2 (Up to 35M/50M)	\$201.60	VDSL Provider2 (Up to 35M/50M)	24 months
VDSL Provider2 (Up to	\$156.80	VDSL Provider2 (Up to 35M/75M)	\$201.60	VDSL Provider2 (Up to	24
35M/75M)	£440.00	VDCL ProviderO (Up to OFM/450M)	#204.60	35M/75M)	months
VDSL Provider2 (Up to 35M/150M)	\$148.96	VDSL Provider2 (Up to 35M/150M)	\$201.60	VDSL Provider2 (Up to 35M/150M)	24 months
VDSL Provider2 (Up to	\$148.96	VDSL Provider2 (Up to 35M/300M)	\$201.60	VDSL Provider2 (Up to	24
35M/300M) VDSL Provider2 (Up to	\$148.96	VDSL Provider2 (Up to 35M/500M)	\$201.60	35M/300M)  VDSL Provider2 (Up to	months 24
35M/500M)	Ψ110.00	, , , , , , , , , , , , , , , , , , , ,	Ψ201.00	35M/500M)	months
VDSL Provider2 (Up to 40M/5M)	\$148.96	VDSL Provider2 (Up to 40M/5M)	\$201.60	VDSL Provider2 (Up to 40M/5M)	24 months
VDSL Provider2 (Up to	\$148.96	VDSL Provider2 (Up to 40M/25M)	\$201.60	VDSL Provider2 (Up to	24
40M/25M)	<b>#450.00</b>	\(\(\text{DQL}\) \(\text{Days in large (1 \text{large 4.0 \text{A}/4 \text{FQM})}\)	<b>#004.00</b>	40M/25M)	months
VDSL Provider2 (Up to 40M/150M)	\$156.80	VDSL Provider2 (Up to 40M/150M)	\$201.60	VDSL Provider2 (Up to 40M/150M)	24 months
VDSL Provider2 (Up to	\$383.04	VDSL Provider2 (Up to 55M/20M)	\$201.60	VDSL Provider2 (Up to	24
55M/20M) VDSL Provider2 (Up to	\$383.04	VDSL Provider2 (Up to 60M/30M)	\$201.60	55M/20M)  VDSL Provider2 (Up to	months 24
60M/30M)	ψ303.04	VBOL 1 TOVIGETZ (OP to CONVISCINI)	Ψ201.00	60M/30M)	months
VDSL Provider2 (Up to	\$469.28	VDSL Provider2 (Up to 75M/35M)	\$201.60	VDSL Provider2 (Up to	24
75M/35M) VDSL Provider2 (Up to	\$469.28	VDSL Provider2 (Up to 80M/40M)	\$201.60	75M/35M) VDSL Provider2 (Up to	months 24
80M/40M)		<u> </u>		80M/40M)	months
VDSL Provider2 (Up to 100M/12M)	\$361.76	VDSL Provider2 (Up to 100M/12M)	\$201.60	VDSL Provider2 (Up to 100M/12M)	24 months
VDSL Provider2 (Up to	\$501.76	VDSL Provider2 (Up to 100M/50M)	\$201.60	VDSL Provider2 (Up to	24
100M/50M) VDSL Provider3 (Up to	\$95.20	VDSL Provider3 (Up to 40M/2M)	\$644.00	100M/50M)  VDSL Provider3 (Up to	months 24
40M/2M)	φ93.20	VDSL Floviders (Op to 40ki/ziki)	φ044.00	40M/2M)	months
VDSL Provider3 (Up to	\$95.20	VDSL Provider3 (Up to 40M/5M)	\$644.00	VDSL Provider3 (Up to	24
40M/5M) VDSL Provider3 (Up to	\$95.20	VDSL Provider3 (Up to 40M/40M)	\$644.00	40M/5M)  VDSL Provider3 (Up to	months 24
40M/40M)		<u> </u>		40M/40M)	months
VDSL Provider3 (Up to 60M/5M)	\$143.36	VDSL Provider3 (Up to 60M/5M)	\$644.00	VDSL Provider3 (Up to 60M/5M)	24 months
VDSL Provider3 (Up to	\$143.36	VDSL Provider3 (Up to 80M/10M)	\$644.00	VDSL Provider3 (Up to	24
80M/10M) VDSL Provider3 (Up to	\$143.36	VDSL Provider3 (Up to 100M/10M)	\$644.00	80M/10M)  VDSL Provider3 (Up to	months 24
100M/10M)	φ143.30	VDSL Providers (Op to 10000/1000)	φ044.00	100M/10M)	months
VDSL Provider3 (Up to	\$143.36	VDSL Provider3 (Up to 100M/100M)	\$644.00	VDSL Provider3 (Up to	24
100M/100M) VDSL Provider3 (Up to	\$168.00	VDSL Provider3 (Up to 140M/20M)	\$644.00	100M/100M)  VDSL Provider3 (Up to	months 24
140M/20M)		<u> </u>		140M/20M)	months
VDSL Provider3 (Up to 200M/40M)	\$207.20	VDSL Provider3 (Up to 200M/40M)	\$644.00	VDSL Provider3 (Up to 200M/40M)	24 months
VDSL Provider3 (Up to	\$207.20	VDSL Provider3 (Up to 200M/200M)	\$644.00	VDSL Provider3 (Up to	24
200M/200M)	<b>COLL 00</b>	VDCI Provide 2 (United FOOLAGE)	004400	200M/200M)	months
VDSL Provider3 (Up to 500M/100M)	\$255.36	VDSL Provider3 (Up to 500M/100M)	\$644.00	VDSL Provider3 (Up to 500M/100M)	24 months
VDSL Provider3 (Up to	\$255.36	VDSL Provider3 (Up to 500M/500M)	\$644.00	VDSL Provider3 (Up to	24
500M/500M)				500M/500M)	months

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
VDSL Provider3 (Up to	\$319.20	VDSL Provider3 (Up to 1000M/1000M)	\$644.00	VDSL Provider3 (Up to	24
1000M/1000M)				1000M/1000M)	months
ADSL Provider3 (Up to 1M/128k)	\$43.68	ADSL Provider3 (Up to 1M/128k)	\$78.40	ADSL Provider3 (Up to 1M/128k)	24 months
ADSL Provider2 (Up to 3M/384k)	\$43.68	ADSL Provider2 (Up to 3M/384k)	\$78.40	ADSL Provider2 (Up to 3M/384k)	24 months
ADSL Provider3 (Up to 3M/2M)	\$76.16	ADSL Provider3 (Up to 3M/2M)	\$78.40	ADSL Provider3 (Up to 3M/2M)	24 months
ADSL Provider3 (Up to 6M/512k)	\$54.88	ADSL Provider3 (Up to 6M/512k)	\$78.40	ADSL Provider3 (Up to 6M/512k)	24 months
ADSL Provider3 (Up to 6M/768k)	\$96.32	ADSL Provider3 (Up to 6M/768k)	\$78.40	ADSL Provider3 (Up to 6M/768k)	24 months
ADSL Provider3 (Up to 7M/2M)	\$76.16	ADSL Provider3 (Up to 7M/2M)	\$78.40	ADSL Provider3 (Up to 7M/2M)	24 months
ADSL Provider3 (Up to 9M/1M)	\$54.88	ADSL Provider3 (Up to 9M/1M)	\$78.40	ADSL Provider3 (Up to 9M/1M)	24 months
ADSL Provider3 (Up to 12M/1M)	\$54.88	ADSL Provider3 (Up to 12M/1M)	\$78.40	ADSL Provider3 (Up to 12M/1M)	24 months
ADSL Provider3 (Up to 18M/1M)	\$80.64	ADSL Provider3 (Up to 18M/1M)	\$78.40	ADSL Provider3 (Up to 18M/1M)	24 months
ADSL Provider3 (Up to 25M/2M)	\$80.64	ADSL Provider3 (Up to 25M/2M)	\$78.40	ADSL Provider3 (Up to 25M/2M)	24 months
ADSL Provider3 (Up to 35M/2M)	\$88.48	ADSL Provider3 (Up to 35M/2M)	\$78.40	ADSL Provider3 (Up to 35M/2M)	24 months
ADSL Provider3 (Up to 40M/20M)	\$156.80	ADSL Provider3 (Up to 40M/20M)	\$78.40	ADSL Provider3 (Up to 40M/20M)	24 months
ADSL Provider3 (Up to 45M/3M)	\$96.32	ADSL Provider3 (Up to 45M/3M)	\$78.40	ADSL Provider3 (Up to 45M/3M)	24 months
ADSL Provider3 (Up to 70M/3M)	\$112.00	ADSL Provider3 (Up to 70M/3M)	\$78.40	ADSL Provider3 (Up to 70M/3M)	24 months
ADSL Provider3 (Up to 90M/5M)	\$160.16	ADSL Provider3 (Up to 90M/5M)	\$78.40	ADSL Provider3 (Up to 90M/5M)	24 months
ADSL Provider3 (Up to 115M/7M)	\$224.00	ADSL Provider3 (Up to 115M/7M)	\$78.40	ADSL Provider3 (Up to 115M/7M)	24 months
ADSL Provider4 (Up to 1.5M/384k)	\$88.48	ADSL Provider4 (Up to 1.5M/384k)	\$117.60	ADSL Provider4 (Up to 1.5M/384k)	24 months
ADSL Provider4 (Up to 3M/768k)	\$95.20	ADSL Provider4 (Up to 3M/768k)	\$117.60	ADSL Provider4 (Up to 3M/768k)	24 months
ADSL Provider4 (Up to 6M/768k)	\$100.80	ADSL Provider4 (Up to 6M/768k)	\$117.60	ADSL Provider4 (Up to 6M/768k)	24 months
ADSL Provider4 (Up to 8M/896k)	\$100.80	ADSL Provider4 (Up to 8M/896k)	\$117.60	ADSL Provider4 (Up to 8M/896k)	24 months
ADSL Provider4 (Up to 10M/896k)	\$106.40	ADSL Provider4 (Up to 10M/896k)	\$117.60	ADSL Provider4 (Up to 10M/896k)	24 months
ADSL Provider4 (Up to 12M/896k)	\$106.40	ADSL Provider4 (Up to 12M/896k)	\$117.60	ADSL Provider4 (Up to 12M/896k)	24 months
ADSL Provider4 (Up to 15M/1.5M)	\$80.64	ADSL Provider4 (Up to 15M/1.5M)	\$117.60	ADSL Provider4 (Up to 15M/1.5M)	24 months
ADSL Provider5 (Up to 3M/384k)	\$80.64	ADSL Provider5 (Up to 3M/384k)	\$156.80	ADSL Provider5 (Up to 3M/384k)	24 months
ADSL Provider5 (Up to 6M/512k)	\$80.64	ADSL Provider5 (Up to 6M/512k)	\$156.80	ADSL Provider5 (Up to 6M/512k)	24 months
ADSL Provider5 (Up to 6M/768k)	\$80.64	ADSL Provider5 (Up to 6M/768k)	\$156.80	ADSL Provider5 (Up to 6M/768k)	24 months
ADSL Provider5 (Up to 12M/1M)	\$80.64	ADSL Provider5 (Up to 12M/1M)	\$156.80	ADSL Provider5 (Up to 12M/1M)	24 months
ADSL Provider5 (Up to 18M/1M)	\$88.48	ADSL Provider5 (Up to 18M/1M)	\$156.80	ADSL Provider5 (Up to 18M/1M)	24 months
ADSL Provider6 (Up to 1.5M/0.128M)	\$59.36	ADSL Provider6 (Up to 1.5M/0.128M)	\$442.40	ADSL Provider6 (Up to 1.5M/0.128M)	24 months
ADSL Provider6 (Up to 1.5M/0.384M)	\$59.36	ADSL Provider6 (Up to 1.5M/0.384M)	\$442.40	ADSL Provider6 (Up to 1.5M/0.384M)	24 months
ADSL Provider6 (Up to 1.5M/768k)	\$69.44	ADSL Provider6 (Up to 1.5M/768k)	\$442.40	ADSL Provider6 (Up to 1.5M/768k)	24 months
ADSL Provider6 (Up to 3M/768k)	\$87.36	ADSL Provider6 (Up to 3M/768k)	\$442.40	ADSL Provider6 (Up to 3M/768k)	24 months

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
ADSL Provider6 (Up to	\$96.32	ADSL Provider6 (Up to 6M/768k)	\$442.40	ADSL Provider6 (Up to	24
6M/768k) ADSL Provider6 (Up to	¢07.26	ADCI Dravidore (Up to 7M/906k)	¢442.40	6M/768k)	months 24
7M/896k)	\$87.36	ADSL Provider6 (Up to 7M/896k)	\$442.40	ADSL Provider6 (Up to 7M/896k)	months
ADSL Provider6 (Up to	\$109.76	ADSL Provider6 (Up to 8M/1M)	\$397.60	ADSL Provider6 (Up to	24
8M/1M)				8M/1M)	months
ADSL Provider6 (Up to	\$77.28	ADSL Provider6 (Up to 10M/1M)	\$442.40	ADSL Provider6 (Up to 10M/1M)	24
10M/1M) ADSL Provider6 (Up to	\$64.96	ADSL Provider6 (Up to 15M/1M)	\$442.40	ADSL Provider6 (Up to	months 24
15M/1M)	4000	/ LDG_ : remaine (op to remainin)		15M/1M)	months
ADSL Provider7 (Up to	\$49.28	ADSL Provider7 (Up to	\$358.40	ADSL Provider7 (Up to	24
0.768M/0.128M)	¢50.26	0.768M/0.128M)	\$250.40	0.768M/0.128M)	months 24
ADSL Provider7 (Up to 1M/0.384M)	\$59.36	ADSL Provider7 (Up to 1M/0.384M)	\$358.40	ADSL Provider7 (Up to 1M/0.384M)	months
ADSL Provider7 (Up to	\$58.24	ADSL Provider7 (Up to 1.5M/384k)	\$358.40	ADSL Provider7 (Up to	24
1.5M/384k)				1.5M/384k)	months
ADSL Provider7 (Up to	\$72.80	ADSL Provider7 (Up to 3M/768k)	\$358.40	ADSL Provider7 (Up to	24
3M/768k) ADSL Provider7 (Up to	\$73.92	ADSL Provider7 (Up to 5M/768k)	\$358.40	3M/768k) ADSL Provider7 (Up to	months 24
5M/768k)	ψ, σ.σ.	Tibel Freducti (op to ontrock)	φοσο. 10	5M/768k)	months
ADSL Provider7 (Up to	\$132.16	ADSL Provider7 (Up to 7.1M/768k)	\$358.40	ADSL Provider7 (Up to	24
7.1M/768k)	Φ4545C	ADCI Duniday7 (Lin to 4584/484)	#250.40	7.1M/768k) ADSL Provider7 (Up to	months 24
ADSL Provider7 (Up to 15M/1M)	\$154.56	ADSL Provider7 (Up to 15M/1M)	\$358.40	15M/1M)	months
Fiber Provider1 (Up to	\$50.40	Fiber Provider1 (Up to 1.5M/384k)	\$162.40	Fiber Provider1 (Up to	24
1.5M/384k)		i i i i i i i i i i i i i i i i i i i		1.5M/384k)	months
Fiber Provider1 (Up to	\$50.40	Fiber Provider1 (Up to 3M/512k)	\$162.40	Fiber Provider1 (Up to	24
3M/512k) Fiber Provider1 (Up to	\$50.40	Fiber Provider1 (Up to 6M/768k)	\$162.40	3M/512k) Fiber Provider1 (Up to	months 24
6M/768k)	φου. 10	l lost i fortaci i (op to divi rock)	ψ102.10	6M/768k)	months
Fiber Provider1 (Up to	\$61.60	Fiber Provider1 (Up to 12M/1M)	\$162.40	Fiber Provider1 (Up to	24
12M/1M)	ФС4 CO	Fiber Drawiders (Up to 40M/4M)	£4.00.40	12M/1M)	months
Fiber Provider1 (Up to 18M/1M)	\$61.60	Fiber Provider1 (Up to 18M/1M)	\$162.40	Fiber Provider1 (Up to 18M/1M)	24 months
Fiber Provider1 (Up to	\$61.60	Fiber Provider1 (Up to 24M/3M)	\$162.40	Fiber Provider1 (Up to	24
24M/3M)				24M/3M)	months
Fiber Provider1 (Up to 25M/5M)	\$61.60	Fiber Provider1 (Up to 25M/5M)	\$162.40	Fiber Provider1 (Up to 25M/5M)	24
Fiber Provider1 (Up to	\$80.64	Fiber Provider1 (Up to 25M/25M)	\$162.40	Fiber Provider1 (Up to	months 24
25M/25M)	φσσ.σ τ	Tibor Frontaci (op to Zem/Zem)	ψ102.10	25M/25M)	months
Fiber Provider1 (Up to	\$96.32	Fiber Provider1 (Up to 45M/6M)	\$162.40	Fiber Provider1 (Up to	24
45M/6M) Fiber Provider1 (Up to	\$80.64	Fiber Provider1 (Up to 50M/10M)	\$162.40	45M/6M)	months 24
50M/10M)	\$60.64	Fiber Provider (Op to Solw/Tolw)	\$162.40	Fiber Provider1 (Up to 50M/10M)	months
Fiber Provider1 (Up to	\$80.64	Fiber Provider1 (Up to 50M/50M)	\$162.40	Fiber Provider1 (Up to	24
50M/50M)				50M/50M)	months
Fiber Provider1 (Up to 75M/8M)	\$129.92	Fiber Provider1 (Up to 75M/8M)	\$162.40	Fiber Provider1 (Up to 75M/8M)	24 months
Fiber Provider1 (Up to	\$80.64	Fiber Provider1 (Up to 100M/20M)	\$162.40	Fiber Provider1 (Up to	24
100M/20M)	,,,,,,	(ор и сониции,	7	100M/20M)	months
Fiber Provider1 (Up to	\$80.64	Fiber Provider1 (Up to 100M/100M)	\$162.40	Fiber Provider1 (Up to	24
100M/100M) Fiber Provider1 (Up to	\$144.48	Fiber Provider1 (Up to 200M/40M)	\$162.40	100M/100M) Fiber Provider1 (Up to	months 24
200M/40M)	Φ144.40	Fiber Frovider (Op to 2001//401/)	\$102.40	200M/40M)	months
Fiber Provider1 (Up to	\$144.48	Fiber Provider1 (Up to 200M/200M)	\$162.40	Fiber Provider1 (Up to	24
200M/200M)	•			200M/200M)	months
Fiber Provider1 (Up to 300M/75M)	\$144.48	Fiber Provider1 (Up to 300M/75M)	\$162.40	Fiber Provider1 (Up to 300M/75M)	24
Fiber Provider1 (Up to	\$144.48	Fiber Provider1 (Up to 300M/300M)	\$162.40	Fiber Provider1 (Up to	months 24
300M/300M)	, , , , , ,	(=	1	300M/300M)	months
Fiber Provider1 (Up to	\$192.64	Fiber Provider1 (Up to 500M/100M)	\$162.40	Fiber Provider1 (Up to	24
500M/100M) Fiber Provider1 (Up to	\$192.64	Fiber Provider1 (Up to 500M/500M)	¢162.40	500M/100M) Fiber Provider1 (Up to	months 24
500M/500M)	φ192.04	Tibel Flovidel I (Op to 500IV/500IVI)	\$162.40	500M/500M)	months
Fiber Provider1 (Up to	\$192.64	Fiber Provider1 (Up to 1000M/200M)	\$162.40	Fiber Provider1 (Up to	24
1000M/200M)				1000M/200M)	months

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Fiber Provider1 (Up to	\$192.64	Fiber Provider1 (Up to 1000M/1000M)		\$162.40	Fiber Provider1 (Up to	24
1000M/1000M)	007.00	F" D : 1 0 (11 1 0514(0514)		Φ040.00	1000M/1000M)	months
Fiber Provider2 (Up to 25M/25M)	\$67.20	Fiber Provider2 (Up to 25M/25M)		\$313.60	Fiber Provider2 (Up to 25M/25M)	24 months
Fiber Provider2 (Up to	\$77.28	Fiber Provider2 (Up to 50M/50M)		\$313.60	Fiber Provider2 (Up to	24
50M/50M)	*****	( )		***************************************	50M/50M)	months
Fiber Provider2 (Up to	\$94.08	Fiber Provider2 (Up to 75M/75M)		\$313.60	Fiber Provider2 (Up to	24
75M/75M)					75M/75M)	months
Fiber Provider2 (Up to	\$134.40	Fiber Provider2 (Up to 150M/150M)		\$313.60	Fiber Provider2 (Up to	24
150M/150M) Fiber Provider2 (Up to	\$235.20	Fiber Provider2 (Up to 300M/300M)		\$313.60	150M/150M) Fiber Provider2 (Up to	months 24
300M/300M)	Ψ200.20	l liber i reviderz (ep te econveccin)		φο το.σσ	300M/300M)	months
Fiber Provider2 (Up to	\$347.20	Fiber Provider2 (Up to 500M/500M)		\$313.60	Fiber Provider2 (Up to	24
500M/500M)					500M/500M)	months
Fiber Provider3 (Up to	\$94.08	Fiber Provider3 (Up to 75M/75M)		\$313.60	Fiber Provider3 (Up to	24
75M/75M) Fiber Provider3 (Up to	\$144.48	Fiber Provider3 (Up to 100M/100M)		\$313.60	75M/75M) Fiber Provider3 (Up to	months 24
100M/100M)	φ144.40	Tibel Floviders (op to Toolvii Toolvii)		φ313.00	100M/100M)	months
Fiber Provider3 (Up to	\$134.40	Fiber Provider3 (Up to 150M/150M)		\$313.60	Fiber Provider3 (Up to	24
150M/150M)					150M/150M)	months
Fiber Provider3 (Up to	\$235.20	Fiber Provider3 (Up to 200M/200M)		\$313.60	Fiber Provider3 (Up to	24
200M/200M) Fiber Provider3 (Up to	\$235.20	Fiber Provider3 (Up to 300M/300M)		\$313.60	200M/200M) Fiber Provider3 (Up to	months 24
300M/300M)	\$235.20	Fiber Froviders (Op to Social/Social)		φ313.00	300M/300M)	months
Fiber Provider3 (Up to	\$347.20	Fiber Provider3 (Up to 500M/500M)		\$313.60	Fiber Provider3 (Up to	24
500M/500M)	·	, ,		<u> </u>	500M/500M)	months
Fiber Provider5 (Up to	\$112.00	Fiber Provider5 (Up to 15M/5M)		\$240.80	Fiber Provider5 (Up to	24
15M/5M)	C4 40 00	Fibou Duovidous (Llasta 4504/4504)		<b>CO 40 00</b>	15M/5M)	months
Fiber Provider5 (Up to 15M/15M)	\$140.00	Fiber Provider5 (Up to 15M/15M)		\$240.80	Fiber Provider5 (Up to 15M/15M)	24 months
Fiber Provider5 (Up to	\$160.16	Fiber Provider5 (Up to 25M/10M)		\$240.80	Fiber Provider5 (Up to	24
25M/10M)	<b>V.00</b>	(ορ το συν.)		<b>V</b>	25M/10M)	months
Fiber Provider5 (Up to	\$184.80	Fiber Provider5 (Up to 25M/25M)		\$240.80	Fiber Provider5 (Up to	24
25M/25M)	0000.00	F" D : 1 5 (11 1 0514/4514)		Φ0.40.00	25M/25M)	months
Fiber Provider5 (Up to 35M/15M)	\$208.32	Fiber Provider5 (Up to 35M/15M)		\$240.80	Fiber Provider5 (Up to 35M/15M)	24 months
Fiber Provider5 (Up to	\$256.48	Fiber Provider5 (Up to 50M/20M)		\$240.80	Fiber Provider5 (Up to	24
50M/20M)	<b>V</b>	(ор из отмения)		<b>V</b> = 10100	50M/20M)	months
Fiber Provider5 (Up to	\$117.60	Fiber Provider5 (Up to 50M/50M)		\$240.80	Fiber Provider5 (Up to	24
50M/50M)	0445.00	F" D : 1 5 (11 : 7514(7514)	-	Φ0.40.00	50M/50M)	months
Fiber Provider5 (Up to 75M/75M)	\$145.60	Fiber Provider5 (Up to 75M/75M)		\$240.80	Fiber Provider5 (Up to 75M/75M)	24 months
Fiber Provider5 (Up to	\$440.16	Fiber Provider5 (Up to 300M/300M)		\$240.80	Fiber Provider5 (Up to	24
300M/300M)	ψ.10.10	l liber i revidere (ep te econveccin)		φ2 10.00	300M/300M)	months
Fiber Provider4 (Up to	\$88.48	Fiber Provider4 (Up to 75M/75M)		\$78.40	Fiber Provider4 (Up to	24
75M/75M)	0110.00	F" D : 1 4 (11 4 40014/40014)		<b>070.40</b>	75M/75M)	months
Fiber Provider4 (Up to 100M/100M)	\$112.00	Fiber Provider4 (Up to 100M/100M)		\$78.40	Fiber Provider4 (Up to 100M/100M)	24
Fiber Provider4 (Up to	\$144.48	Fiber Provider4 (Up to 200M/200M)		\$78.40	Fiber Provider4 (Up to	months 24
200M/200M)	<b>4</b>			ψ. σσ	200M/200M)	months
Fiber Provider4 (Up to	\$176.96	Fiber Provider4 (Up to 300M/300M)		\$78.40	Fiber Provider4 (Up to	24
300M/300M)					300M/300M)	months
Fiber Provider4 (Up to	\$208.32	Fiber Provider4 (Up to 500M/500M)		\$78.40	Fiber Provider4 (Up to	24
500M/500M) Fiber Provider4 (Up to	\$320.32	Fiber Provider4 (Up to 700M/700M)		\$78.40	500M/500M) Fiber Provider4 (Up to	months 24
700M/700M)	ψ020.02	Tibel Florider (Op to Footwiff Colv)		ψ10.40	700M/700M)	months
Fiber Provider4 (Up to	\$440.16	Fiber Provider4 (Up to 1000M/1000M)		\$78.40	Fiber Provider4 (Up to	24
1000M/1000M)					1000M/1000M)	months
Fiber Provider6 (Up to	\$81.76	Fiber Provider6 (Up to 10M/1M)		\$240.80	Fiber Provider6 (Up to	24
10M/1M) Fiber Provider6 (Up to	\$81.76	Fiber Provider6 (Up to 10M/10M)		\$240.80	10M/1M) Fiber Provider6 (Up to	months 24
10M/10M)	ψ01.70	Tibel Tiovide to (op to Tolvi/Tolvi)		ΨΖ40.00	10M/10M)	months
Fiber Provider6 (Up to	\$98.56	Fiber Provider6 (Up to 15M/1M)		\$240.80	Fiber Provider6 (Up to	24
15M/1M)					15M/1M)	months
Fiber Provider6 (Up to	\$98.56	Fiber Provider6 (Up to 15M/15M)		\$240.80	Fiber Provider6 (Up to	24
15M/15M)					15M/15M)	months

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Fiber Provider6 (Up to	\$114.24	Fiber Provider6 (Up to 25M/10M)	\$240.80	\ '	24
25M/10M) Fiber Provider6 (Up to	\$114.24	Fiber Provider6 (Up to 25M/25M)	\$240.80	25M/10M) Fiber Provider6 (Up to	months 24
25M/25M)	ψ114.24	Tibel Flovidero (Op to 25/0/25/0)	Ψ240.00	25M/25M)	months
Fiber Provider6 (Up to 50M/10M)	\$122.08	Fiber Provider6 (Up to 50M/10M)	\$240.80	Fiber Provider6 (Up to 50M/10M)	24 months
Fiber Provider6 (Up to 50M/50M)	\$129.92	Fiber Provider6 (Up to 50M/50M)	\$240.80	Fiber Provider6 (Up to 50M/50M)	24 months
Fiber Provider6 (Up to 150M/50M)	\$137.76	Fiber Provider6 (Up to 150M/50M)	\$240.80	Fiber Provider6 (Up to 150M/50M)	24 months
Fiber Provider6 (Up to 250M/250M)	\$274.40	Fiber Provider6 (Up to 250M/250M)	\$240.80		24 months
Fiber Provider7 (Up to 1M/1M)	\$143.36	Fiber Provider7 (Up to 1M/1M)	\$319.20		24 months
Fiber Provider7 (Up to 1.5M/1.5M)	\$185.92	Fiber Provider7 (Up to 1.5M/1.5M)	\$319.20		24 months
Fiber Provider7 (Up to 2M/2M)	\$210.56	Fiber Provider7 (Up to 2M/2M)	\$319.20		24 months
Fiber Provider7 (Up to 3M/3M)	\$232.96	Fiber Provider7 (Up to 3M/3M)	\$319.20		24 months
Fiber Provider7 (Up to 4M/4M)	\$308.00	Fiber Provider7 (Up to 4M/4M)	\$319.20		24 months
Fiber Provider7 (Up to 5M/5M)	\$344.96	Fiber Provider7 (Up to 5M/5M)	\$319.20		24 months
Fiber Provider7 (Up to 6M/6M)	\$360.64	Fiber Provider7 (Up to 6M/6M)	\$319.20		24 months
Fiber Provider7 (Up to 7M/7M)	\$420.00	Fiber Provider7 (Up to 7M/7M)	\$319.20		24 months
Fiber Provider7 (Up to 8M/8M)	\$479.36	Fiber Provider7 (Up to 8M/8M)	\$319.20		24 months
Fiber Provider7 (Up to 9M/9M)	\$496.16	Fiber Provider7 (Up to 9M/9M)	\$319.20		24 months
Fiber Provider7 (Up to 10M/10M)	\$512.96	Fiber Provider7 (Up to 10M/10M)	\$319.20		24 months
Fiber Provider7 (Up to 12M/12M)	\$618.24	Fiber Provider7 (Up to 12M/12M)	\$319.20		24 months
Fiber Provider7 (Up to 15M/15M)	\$736.96	Fiber Provider7 (Up to 15M/15M)	\$319.20		24 months
Fiber Provider7 (Up to 20M/20M)	\$883.68	Fiber Provider7 (Up to 20M/20M)	\$319.20		24 months
Fiber Provider7 (Up to 25M/25M)	\$1,148.00	Fiber Provider7 (Up to 25M/25M)	\$319.20		24 months
Fiber Provider7 (Up to 30M/30M)	\$1,324.96	Fiber Provider7 (Up to 30M/30M)	\$319.20		24 months
Fiber Provider7 (Up to 40M/40M)	\$1,619.52	Fiber Provider7 (Up to 40M/40M)	\$319.20		24 months
Fiber Provider7 (Up to 50M/50M)	\$1,840.16	Fiber Provider7 (Up to 50M/50M)	\$319.20		24 months
Fiber Provider7 (Up to 75M/75M)	\$2,503.20	Fiber Provider7 (Up to 75M/75M)	\$319.20		24 months
Fiber Provider7 (Up to 100M/100M)	\$2,592.80	Fiber Provider7 (Up to 100M/100M)	\$319.20		24
Fiber Provider8 (Up to 10M/0.768M)	\$128.80	Fiber Provider8 (Up to 10M/0.768M)	\$397.60		months 24
Fiber Provider8 (Up to	\$144.48	Fiber Provider8 (Up to 15M/1M)	\$397.60		months 24
Fiber Provider8 (Up to	\$144.48	Fiber Provider8 (Up to 20M/2M)	\$397.60	Fiber Provider8 (Up to	months 24
20M/2M) Fiber Provider8 (Up to	\$176.96	Fiber Provider8 (Up to 25M/2M)	\$397.60	` .	months 24
25M/2M) Fiber Provider8 (Up to	\$176.96	Fiber Provider8 (Up to 40M/5M)	\$397.60	` '	months 24
Fiber Provider8 (Up to	\$176.96	Fiber Provider8 (Up to 40M/20M)	\$397.60	` .	months 24
Fiber Provider8 (Up to	\$406.56	Fiber Provider8 (Up to 50M/50M)	\$397.60	` '	months 24
50M/50M)				50M/50M)	months

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Fiber Provider8 (Up to	\$288.96	Fiber Provider8 (Up to 60M/60M)	П	\$397.60	Fiber Provider8 (Up to	24
60M/60M) Fiber Provider8 (Up to	\$288.96	Fiber Provider8 (Up to 80M/80M)	Н	\$397.60	60M/60M) Fiber Provider8 (Up to	months 24
80M/80M)	4200.00	l i	Ш		80M/80M)	months
Fiber Provider8 (Up to 100M/100M)	\$288.96	Fiber Provider8 (Up to 100M/100M)		\$397.60	Fiber Provider8 (Up to 100M/100M)	24 months
Fiber Provider8 (Up to 200M/200M)	\$320.32	Fiber Provider8 (Up to 200M/200M)		\$397.60	Fiber Provider8 (Up to 200M/200M)	24 months
Fiber Provider8 (Up to 300M/300M)	\$368.48	Fiber Provider8 (Up to 300M/300M)		\$397.60	Fiber Provider8 (Up to 300M/300M)	24 months
Fiber Provider8 (Up to 400M/400M)	\$368.48	Fiber Provider8 (Up to 400M/400M)	Т	\$397.60	Fiber Provider8 (Up to 400M/400M)	24 months
Fiber Provider8 (Up to 500M/500M)	\$368.48	Fiber Provider8 (Up to 500M/500M)	П	\$397.60	Fiber Provider8 (Up to 500M/500M)	24 months
Fiber Provider8 (Up to 1000M/1000M)	\$448.00	Fiber Provider8 (Up to 1000M/1000M)	Н	\$397.60	Fiber Provider8 (Up to 1000M/1000M)	24
Fiber Provider9 (Up to	\$638.40	Fiber Provider9 (Up to 25M/25M)	Н	\$800.80	Fiber Provider9 (Up to	months 24
25M/25M) Fiber Provider9 (Up to	\$958.72	Fiber Provider9 (Up to 50M/50M)	Н	\$800.80	25M/25M) Fiber Provider9 (Up to	months 24
Fiber Provider9 (Up to	\$1,439.20	Fiber Provider9 (Up to 100M/100M)	Н	\$800.80	50M/50M) Fiber Provider9 (Up to	months 24
100M/100M) Fiber Provider9 (Up to	\$2,078.72	Fiber Provider9 (Up to 200M/200M)	Н	\$800.80	100M/100M) Fiber Provider9 (Up to	months 24
200M/200M) Fiber Provider9 (Up to	\$3,198.72	Fiber Provider9 (Up to 500M/500M)	Н	\$800.80	200M/200M) Fiber Provider9 (Up to	months 24
500M/500M) Fiber Provider9 (Up to	\$3,998.40	Fiber Provider9 (Up to 1000M/1000M)	Н	\$800.80	500M/500M) Fiber Provider9 (Up to	months 24
1000M/1000M) Fiber Provider10 (Up to	\$124.32	Fiber Provider10 (Up to 25M/25M)	Н	\$319.20	1000M/1000M) Fiber Provider10 (Up to	months 24
25M/25M)		``	Ш		25M/25M)	months
Fiber Provider10 (Up to 35M/35M)	\$210.56	Fiber Provider10 (Up to 35M/35M)		\$319.20	Fiber Provider10 (Up to 35M/35M)	24 months
Fiber Provider10 (Up to 200M/200M)	\$260.96	Fiber Provider10 (Up to 200M/200M)		\$156.80	Fiber Provider10 (Up to 200M/200M)	24 months
Fiber Provider11 (Up to 50M/50M)	\$87.36	Fiber Provider11 (Up to 50M/50M)		\$156.80	Fiber Provider11 (Up to 50M/50M)	24 months
Fiber Provider11 (Up to 100M/100M)	\$144.48	Fiber Provider11 (Up to 100M/100M)		\$156.80	Fiber Provider11 (Up to 100M/100M)	24 months
Fiber Provider11 (Up to 300M/300M)	\$253.12	Fiber Provider11 (Up to 300M/300M)		\$156.80	Fiber Provider11 (Up to 300M/300M)	24 months
Fiber Provider11 (Up to 500M/500M)	\$360.64	Fiber Provider11 (Up to 500M/500M)	П	\$156.80	Fiber Provider11 (Up to 500M/500M)	24 months
Fiber Provider11 (Up to 1000M/1000M)	\$432.32	Fiber Provider11 (Up to 1000M/1000M)	Т	\$156.80	Fiber Provider11 (Up to 1000M/1000M)	24 months
Cable Provider1 (Up to	\$120.96	Cable Provider1 (Up to 35M/5M)	Т	\$78.40	Cable Provider1 (Up to	24
35M/5M) Cable Provider1 (Up to	\$152.32	Cable Provider1 (Up to 100M/15M)	Н	\$78.40	35M/5M) Cable Provider1 (Up to	months 24
100M/15M) Cable Provider1 (Up to	\$184.80	Cable Provider1 (Up to 200M/20M)	Н	\$78.40	100M/15M) Cable Provider1 (Up to	months 24
200M/20M) Cable Provider1 (Up to	\$224.00	Cable Provider1 (Up to 300M/30M)	Н	\$78.40	200M/20M) Cable Provider1 (Up to	months 24
300M/30M) Cable Provider1 (Up to	\$304.64	Cable Provider1 (Up to 600M/35M)	Щ		300M/30M)  Cable Provider1 (Up to	months
600M/35M)	· ·		Ц	\$78.40	600M/35M)	months
Cable Provider1 (Up to 1000M/35M)	\$368.48	Cable Provider1 (Up to 1000M/35M)	Ц	\$78.40	Cable Provider1 (Up to 1000M/35M)	24 months
Cable Provider2 (up to 25M/5M)	\$96.32	Cable Provider2 (up to 25M/5M)		\$156.80	Cable Provider2 (up to 25M/5M)	24 months
Cable Provider2 (up to 50M/10M)	\$120.96	Cable Provider2 (up to 50M/10M)		\$156.80	Cable Provider2 (up to 50M/10M)	24 months
Cable Provider2 (up to 100M/20M)	\$192.64	Cable Provider2 (up to 100M/20M)		\$156.80	Cable Provider2 (up to 100M/20M)	24 months
Cable Provider2 (up to 200M/20M)	\$280.00	Cable Provider2 (up to 200M/20M)		\$156.80	Cable Provider2 (up to 200M/20M)	24 months
Cable Provider2 (up to 300M/30M)	\$344.96	Cable Provider2 (up to 300M/30M)		\$156.80	Cable Provider2 (up to 300M/30M)	24 months

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Cable Provider2 (Up to 500M/35M)	\$448.00	Cable Provider2 (Up to 500M/35M)		\$156.80	Cable Provider2 (Up to 500M/35M)	24 months
Cable Provider2 (Up to 1000M/35M)	\$648.48	Cable Provider2 (Up to 1000M/35M)	$\parallel$	\$156.80	Cable Provider2 (Up to 1000M/35M)	24 months
Cable Provider3 (Up to 60M/25M)	\$112.00	Cable Provider3 (Up to 60M/25M)		\$156.80	Cable Provider3 (Up to 60M/25M)	24 months
Cable Provider3 (Up to 150M/35M)	\$168.00	Cable Provider3 (Up to 150M/35M)	П	\$156.80	Cable Provider3 (Up to 150M/35M)	24 months
Cable Provider3 (Up to 250M/50M)	\$240.80	Cable Provider3 (Up to 250M/50M)	I	\$156.80	Cable Provider3 (Up to 250M/50M)	24 months
Cable Provider3 (Up to 450M/50M)	\$288.96	Cable Provider3 (Up to 450M/50M)	П	\$156.80	Cable Provider3 (Up to 450M/50M)	24 months
Cable Provider4 (Up to 200M/10M)	\$112.00	Cable Provider4 (Up to 200M/10M)		\$281.12	Cable Provider4 (Up to 200M/10M)	24 months
Cable Provider4 (Up to 600M/35M)	\$192.64	Cable Provider4 (Up to 600M/35M)	П	\$156.80	Cable Provider4 (Up to 600M/35M)	24 months
Cable Provider4 (Up to 940M/35M)	\$400.96	Cable Provider4 (Up to 940M/35M)		\$156.80	Cable Provider4 (Up to 940M/35M)	24 months
Cable Provider5 (Up to 10M/1M)	\$147.84	Cable Provider5 (Up to 10M/1M)	П	\$156.80	Cable Provider5 (Up to 10M/1M)	24 months
Cable Provider5 (Up to 10M/2M)	\$316.96	Cable Provider5 (Up to 10M/2M)		\$156.80	Cable Provider5 (Up to 10M/2M)	24 months
Cable Provider5 (Up to 12M/2M)	\$316.96	Cable Provider5 (Up to 12M/2M)	T	\$156.80	Cable Provider5 (Up to 12M/2M)	24 months
Cable Provider5 (Up to 50M/5M)	\$128.80	Cable Provider5 (Up to 50M/5M)		\$156.80	Cable Provider5 (Up to 50M/5M)	24 months
Cable Provider5 (Up to 50M/8M)	\$128.80	Cable Provider5 (Up to 50M/8M)	$\prod$	\$156.80	Cable Provider5 (Up to 50M/8M)	24 months
Cable Provider5 (Up to 100M/10M)	\$168.00	Cable Provider5 (Up to 100M/10M)		\$224.00	Cable Provider5 (Up to 100M/10M)	24 months
Quote Based Services Level	\$80.64	Quote Based Services Level 1(Cable/Fiber/DSL/Ethernet)	$\sqcap$	\$313.60	Quote Based Services Level	24 months
1(Cable/Fiber/DSL/Ether net)					1(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$120.96	Quote Based Services Level 2(Cable/Fiber/DSL/Ethernet)		\$313.60	Quote Based Services Level	24 months
2(Cable/Fiber/DSL/Ether net)		, , , , , , , , , , , , , , , , , , , ,			2(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$160.16	Quote Based Services Level 3(Cable/Fiber/DSL/Ethernet)	$\sqcap$	\$313.60	Quote Based Services Level	24 months
3(Cable/Fiber/DSL/Ether net)		,			3(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$200.48	Quote Based Services Level 4(Cable/Fiber/DSL/Ethernet)		\$313.60	Quote Based Services Level	24 months
4(Cable/Fiber/DSL/Ether net)		(3333737373737			4(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$240.80	Quote Based Services Level 5(Cable/Fiber/DSL/Ethernet)	T	\$313.60	Quote Based Services Level	24 months
5(Cable/Fiber/DSL/Ether net)		,,			5(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$280.00	Quote Based Services Level 6(Cable/Fiber/DSL/Ethernet)		\$313.60	Quote Based Services Level	24 months
6(Cable/Fiber/DSL/Ether net)		S(Casion Ison SCI Entornot)			6(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$320.32	Quote Based Services Level 7(Cable/Fiber/DSL/Ethernet)	П	\$313.60	Quote Based Services Level	24 months
7(Cable/Fiber/DSL/Ether net)		- (Sasion Ison Soll Entormot)			7(Cable/Fiber/DSL/Ethernet	
Quote Based Services Level	\$360.64	Quote Based Services Level 8(Cable/Fiber/DSL/Ethernet)		\$313.60	Quote Based Services Level	24 months
8(Cable/Fiber/DSL/Ether net)		S(Subject 100/100LEtticHict)			8(Cable/Fiber/DSL/Ethernet	Montrio
Quote Based Services Level	\$400.96	Quote Based Services Level 9(Cable/Fiber/DSL/Ethernet)	T	\$313.60	Quote Based Services Level	24 months
9(Cable/Fiber/DSL/Ether net)		Stable in Bondon Enternet			9(Cable/Fiber/DSL/Ethernet	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Quote Based Services Level 10(Cable/Fiber/DSL/Ethe met)	\$440.16	Quote Based Services Level 10(Cable/Fiber/DSL/Ethernet)	\$313.60	Quote Based Services Level 10(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 11(Cable/Fiber/DSL/Ethe rnet)	\$480.48	Quote Based Services Level 11(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 11(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 12(Cable/Fiber/DSL/Ethe rnet)	\$520.80	Quote Based Services Level 12(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 12(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 13(Cable/Fiber/DSL/Ethe rnet)	\$560.00	Quote Based Services Level 13(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 13(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 14(Cable/Fiber/DSL/Ethe rnet)	\$640.64	Quote Based Services Level 14(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 14(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 15(Cable/Fiber/DSL/Ethe rnet)	\$720.16	Quote Based Services Level 15(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 15(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 16(Cable/Fiber/DSL/Ethe rnet)	\$800.80	Quote Based Services Level 16(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 16(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 17(Cable/Fiber/DSL/Ethe rnet)	\$880.32	Quote Based Services Level 17(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 17(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 18(Cable/Fiber/DSL/Ethe rnet)	\$960.96	Quote Based Services Level 18(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 18(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 19(Cable/Fiber/DSL/Ethe rnet)	\$1,040.48	Quote Based Services Level 19(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 19(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 20(Cable/Fiber/DSL/Ethe rnet)	\$1,120.00	Quote Based Services Level 20(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 20(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 21(Cable/Fiber/DSL/Ethe rnet)	\$1,200.64	Quote Based Services Level 21(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 21(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 22(Cable/Fiber/DSL/Ethe rnet)	\$1,280.16	Quote Based Services Level 22(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 22(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 23(Cable/Fiber/DSL/Ethe rnet)	\$1,360.80	Quote Based Services Level 23(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 23(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 24(Cable/Fiber/DSL/Ethe rnet)	\$1,440.32	Quote Based Services Level 24(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 24(Cable/Fiber/DSL/Ethern et)	24 months
Quote Based Services Level 25(Cable/Fiber/DSL/Ethe rnet)	\$1,520.96	Quote Based Services Level 25(Cable/Fiber/DSL/Ethernet)	\$800.80	Quote Based Services Level 25(Cable/Fiber/DSL/Ethern et)	24 months

26(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 27(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 27(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 27(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 32(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 32(Cable/Fiber/DSL/Ethernet)	onths onths onths onths
26(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 27(Cable/Fiber/DSL/Ethernett)  27(Cable/Fiber/DSL/Ethernett)  27(Cable/Fiber/DSL/Ethernett)  27(Cable/Fiber/DSL/Ethernett)  27(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernett)  28(Cable/Fiber/DSL/Ethernett)  28(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernett)  28(Cable/Fiber/DSL/Ethernett)  8800.80  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernett)  29(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  29(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  29(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  4800.80  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  4800.80  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)  31(Cable/Fiber/DSL/Ethernett)	onths onths onths onths
Time   Countries	onths onths onths onths
Quote Based Services   \$1,680.00   Quote Based Services Level   27(Cable/Fiber/DSL/Ethernet)   \$800.80   Quote Based Services   Quote Based Services   24   Roo   27(Cable/Fiber/DSL/Ethernet)   \$800.80   Quote Based Services   Quote Based Services   24   Roo   Quote Based Services   Quote Based Services   24   Level   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   28(Cable/Fiber/DSL/Ethernet)   29(Cable/Fiber/DSL/Ethernet)   29(Cable/Fiber/DSL/E	onths onths onths onths
Level 27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  24  24  24  25(Cable/Fiber/DSL/Ethernet)  25(Cable/Fiber/DSL/Ethernet)  26(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  28(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  20(Cable/Fiber/DSL/Ethernet)  20(Cable/Fiber/DSL	onths onths onths onths
27(Cable/Fiber/DSL/Ethernet)  27(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level	onths onths onths
Quote Based Services Level 28(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 28(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethermet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethermet)	onths onths onths
Level 28(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  29(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  30(Cable/Fiber/DSL/Ethernet)  31(Cable/Fiber/DSL/Ethernet)	onths onths onths
28(Cable/Fiber/DSL/Ethern rmet)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 29(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 32(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 32(Cable/Fiber/DSL/Ethern et)	onths
Time   Company	onths onths onths
Level 29(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level  32(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level  \$800.80  Quote Based Services Level  Authorized  Authorize	onths onths onths
29(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 30(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethern et)  31(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level 32(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level  \$800.80  Quote Based Services Level 31(Cable/Fiber/DSL/Ethern et)  Quote Based Services Level  \$800.80	onths
Timet   Countries   Countrie	onths
Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)\$1,920.80Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)\$800.80Quote Based Services Level 30(Cable/Fiber/DSL/Ethernet)24 moi 30(Cable/Fiber/DSL/Ethernet)Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)\$2,000.32Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)\$800.80Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)24 Moi 31(Cable/Fiber/DSL/Ethernet)Quote Based Services Level\$2,080.96Quote Based Services Level 32(Cable/Fiber/DSL/Ethernet)\$800.80Quote Based Services Level24 Level	onths
Level 30(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 400.80  Quote Based Services Level 5800.80  Quote Based Services 24  Level 700.80  Quote Based Services 24  Level 8800.80	onths
rmet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level  Quote Based Services Level  \$800.80 Quote Based Services Level	onths
Quote Based Services     \$2,000.32     Quote Based Services Level     \$800.80     Quote Based Services     24       Level     31(Cable/Fiber/DSL/Ether rnet)     31(Cable/Fiber/DSL/Ethernet)     31(Cable/Fiber/DSL/Ethernet)     31(Cable/Fiber/DSL/Ethernet)       Quote Based Services     \$2,080.96     Quote Based Services Level     \$800.80     Quote Based Services     24       Level     32(Cable/Fiber/DSL/Ethernet)     \$800.80     Quote Based Services     24       Level     moi	onths
Level 31(Cable/Fiber/DSL/Ethernet)  Cable/Fiber/DSL/Ethernet)  Level 31(Cable/Fiber/DSL/Ethernet)  Quote Based Services Level 42(Cable/Fiber/DSL/Ethernet)  Cable/Fiber/DSL/Ethernet)  Level 31(Cable/Fiber/DSL/Ethernet)  Cable/Fiber/DSL/Ethernet)  Level 43(Cable/Fiber/DSL/Ethernet)  Cable/Fiber/DSL/Ethernet)	onths
31(Cable/Fiber/DSL/Ethern et)  Quote Based Services  \$2,080.96  Quote Based Services Level  \$800.80  Quote Based Services	
rnet) et)  Quote Based Services \$2,080.96 Quote Based Services Level \$800.80 Quote Based Services 24  Level \$32(Cable/Fiber/DSL/Ethernet) Level more	
Level 32(Cable/Fiber/DSL/Ethernet) Level mod	
	onths
32(Cable/Fiber/DSL/Ethern 32(Cable/Fiber/DSL	
rnet) et) Quote Based Services \$2,160.48 Quote Based Services Level \$800.80 Quote Based Services 24	
	onths
33(Cable/Fiber/DSL/Ethe 33(Cable/Fiber/DSL/Ethern	
met) et)	
Quote Based Services\$2,240.00Quote Based Services Level\$800.80Quote Based Services24Level34(Cable/Fiber/DSL/Ethernet)LevelLevelmol	onths
34(Cable/Fiber/DSL/Ether)	711113
rnet) et)	
Quote Based Services \$2,320.64 Quote Based Services Level \$800.80 Quote Based Services 24	
	onths
35(Cable/Fiber/DSL/Ethernet) 35(Cable/Fiber/DSL/Ethernet)	
Quote Based Services \$2,400.16 Quote Based Services Level \$800.80 Quote Based Services 24	
Level 36(Cable/Fiber/DSL/Ethernet) Level mod	onths
36(Cable/Fiber/DSL/Ethern 36(Cable/Fiber/DSL/Ethern	
rnet) et) Quote Based Services \$2,480.80 Quote Based Services Level \$800.80 Quote Based Services 24	
	onths
37(Cable/Fiber/DSL/Ethe 37(Cable/Fiber/DSL/Ethern	
rnet) et)	
Quote Based Services \$2,560.32 Quote Based Services Level \$800.80 Quote Based Services 24	
Level 38(Cable/Fiber/DSL/Ethernet) Level mod 38(Cable/Fiber/DSL/Ethernet) 38(Cable/Fiber/DSL/Ethernet)	onths
rnet)	
4G/LTE 1GB (Pooled \$29.12 4G/LTE 1GB (Pooled Plan) withNBD \$874.72 Fixed cellular modem with 12	
	onths
equipment maintenance Enterprise installation  4G/LTE 2GB (Pooled Plan)withNBD \$874.72 Fixed cellular modem with 12	
	onths
maintenance Enterprise installation	
4G/LTE 3GB (Pooled \$47.04 4G/LTE 3GB (Pooled Plan) withNBD \$874.72 Fixed cellular modem with 12	
	onths
equipment maintenance Enterprise installation  4G/LTE 4GB (Pooled Plan) withNBD \$874.72 Fixed cellular modem with 12	
	onths
equipment maintenance Enterprise installation	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
4G/LTE 5GB (Pooled	\$64.96	4G/LTE 5GB (Pooled Plan) withNBD	П	\$874.72	Fixed cellular modem with	12
Plan) withNBD		equipment maintenance	П		built-in antenna, Standard	months
equipment maintenance			Ц		Enterprise installation	
4G/LTE 10GB (Pooled	\$109.76	4G/LTE 10GB (Pooled Plan) withNBD	П	\$874.72	Fixed cellular modem with	12
Plan) withNBD		equipment maintenance	П		built-in antenna, Standard	months
equipment maintenance	\$454.5C	AC/LTE 45CD (Dealed Dies) with NDD	Н	<b>CO74 70</b>	Enterprise installation	10
4G/LTE 15GB (Pooled Plan) with NBD	\$154.56	4G/LTE 15GB (Pooled Plan) withNBD equipment maintenance	П	\$874.72	Fixed cellular modem with	12
equipment maintenance		equipment maintenance	П		built-in antenna, Standard Enterprise installation	months
4G/LTE 20GB (Pooled	\$199.36	4G/LTE 20GB (Pooled Plan) withNBD	Н	\$874.72	Fixed cellular modem with	12
Plan) withNBD	Ψ100.00	equipment maintenance	П	φο, 2	built-in antenna, Standard	months
equipment maintenance		- 1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-	П		Enterprise installation	
4G/LTE 25GB (Pooled	\$244.16	4G/LTE 25GB (Pooled Plan) withNBD	П	\$874.72	Fixed cellular modem with	12
Plan) withNBD		equipment maintenance	П		built-in antenna, Standard	months
equipment maintenance			Ц		Enterprise installation	
4G data overage (per	\$0.00		П	\$16.80	4G data overage (per GB)	
GB)	<b>#</b> 000 04	F: IME I (II I FMOM)	Ц	<b>04.070.00</b>	F: 100; 1 (1)	10
Fixed Wireless (Up to	\$360.64	Fixed Wireless (Up to 5M/2M)	П	\$1,976.80	Fixed Wireless (Up to	12
5M/2M) Fixed Wireless (Up to	\$376.32	Fixed Wireless (Up to 10M/2M)	Н	\$1,976.80	5M/2M) Fixed Wireless (Up to	months 12
10M/2M)	φ3/0.32	Tixed Wileless (Op to Tolvi/2lvi)	П	φ1,970.00	10M/2M)	months
Fixed Wireless (Up to	\$408.80	Fixed Wireless (Up to 15M/3M)	Н	\$1,976.80	Fixed Wireless (Up to	12
15M/3M)	ψ 100.00	I mad Tribood (op to Tellifelli)	П	ψ1,070.00	15M/3M)	months
Fixed Wireless (Up to	\$472.64	Fixed Wireless (Up to 20M/5M)	П	\$1,976.80	Fixed Wireless (Up to	12
20M/5M)			Ш		20M/5M)	months
Fixed Wireless (Up to	\$552.16	Fixed Wireless (Up to 35M/7M)	П	\$1,976.80	Fixed Wireless (Up to	12
35M/7M)			Ц		35M/7M)	months
Fixed Wireless (Up to	\$632.80	Fixed Wireless (Up to 50M/10M)	П	\$1,976.80	Fixed Wireless (Up to	12
50M/10M) Access Continuity - 2GB	\$44.79	Jupiter 25/3 Mbps with 2GB Anytime	Н		50M/10M)	months
Access Continuity - 20D	φ44.79	Usage	П			
	\$0.00	Usage	П	\$0.00		
Business 35	\$74.24	Jupiter 25/3 Mbps with 25GB Business	Н			
240000 00	Ų <u> </u>	Usage and 10GB Anytime Usage	П			
		, , , , , , , , , , , , , , , , , , , ,	Ш			
Business 50	\$106.06	Jupiter 25/3 Mbps with 25GB Business	П			
		Usage and 25GB Anytime Usage	П			
D : 75	<b>0.150.14</b>	1 11 05/0141 11 0500 0	Н			
Business 75	\$159.11	Jupiter 25/3 Mbps with 25GB Business	П			
		Usage and 50GB Anytime Usage	П			
Business 100	\$212.14	Jupiter 25/3 Mbps with 25GB Business	Н			
240000 100	Ψ=.=	Usage and 75GB Anytime Usage	П			
		, , , , , , , , , , , , , , , , , , , ,				
Business 150	\$291.42	Jupiter 25/3 Mbps with 50GB Business	П			
		Usage and 100GB Anytime Usage	П			
D : 050	<b>#</b> 400 55	1 '' 05/0141 '' 5000 0	Ц			
Business 250	\$498.55	Jupiter 25/3 Mbps with 50GB Business	П			
		Usage and 200GB Anytime Usage	П			
Business 500	\$1,028.88	Jupiter 25/3 Mbps with 50GB Business	П			
	, ,,	Usage and 450GB Anytime Usage	П			
			Ш			
Token for HTS Plan per GB				\$3.36	Token for HTS Plan per GB	
1 Static IP	\$34.45	1 Static IP (HughesNet VSAT	H			
		Services)	Ш			
5 Static Ips	\$57.41	5 Static Ips (HughesNet Vsat Services)				
1.0		1.0	Ц			
1 Static IP	\$16.80	1 Static IP - Non-VSATservices	Ц			
5 Static Ips	\$50.40	5 Static Ips - Non-VSATservices				
	_		-			

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
(Optional) Field Service - Next Business Day	\$22.68	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, Monday through Friday, holidays excepted, beginning the next business day after a call is received.				
(Optional) Field Service - Next Celendar Day	\$26.07	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year, beginning the next business day after a call is received.				
(Optional) Field Service - Same Day 8AM - 5PM	\$29.18	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year.				
(Optional) Field Service - Same Day 24 x 7	\$31.96	Field Service – Coverage Hours are 24 hours per day, 365 days per year.	П			
Remote Field Service, Hourly Charge				\$139.08	Hourly labor rate for field service technician for work during standard business hours. Only applies if customer does not have a maintenance plan.	
Remote Field Service, Hourly Charge				\$167.85	Hourly labor rate for field service technician for work not during standard business hours. Only applies if customer does not have a maintenance plan.	
Remote Field Service, Trip Charge				\$177.45	Trip charge for field service technician to visit customer site. Only applies if customer does not have a maintenance plan.	
Installation				\$335.72	Incremental charge for installation requested with less than (15) business days notice.	
Installation			П	\$287.76	All service plan agreements are 12 month term.	
Installation				\$335.72	Site survey to determine the time and materials to install HughesNet equipment. Quotation provided from results of survey.	
Installation				\$863.27	Standard Fixed HughesNet equipment Installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Installation				\$457.03	Standard Fixed HughesNet equipment De-installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Installation			T	\$812.02	Standard Fixed VSAT Re- installation .98m/1.2m	
Installation				\$1,269.52	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems and within same city or up to 100 miles between the existing and new site in the lower 48 states.	

Plan Name/Title	MRC	MRC Description	N	IRC	NRC Description	Term
Installation			\$1,5	523.42	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems where by the existing and new HughesNet equipment sites are more than 100 miles apart in the lower 48 states.	
Installation			\$36	64.49	Incremental charge for customer requested/required pole mount with height up to 10 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Installation			\$53	37.15	Incremental charge for customer requested/required pole mount with height 11 – 15 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Installation			\$3:	35.72	Charge to repoint/repolorize the antenna at the same customer location	
Installation			\$33	35.72	same point of entry and cable will be used	
Installation			\$38	83.68	new point of entry and cable is required	
Installation			\$3:	35.72	Upgrade indoor electronics or outdoor electronics - VSAT. Does not include hardware	
Installation			\$38	83.68	Upgrade indoor electronics and outdoor electronics - VSAT. Does not include hardware	
Installation			\$24	87.76	This charge only applies in cases where the installation has been cancelled by Customer within seven (7) days of the scheduled installation.	
Installation			\$	1.20	Additional Cat 5 data cable installed > 100'	
Service Plan Change (Upgrade/Downgrade) Fee			\$33	35.72	Charge to change from one service plan to another. Requires technician visit to customer's location to decommission the old service plan and re-commission the new service plan.	
Miscellaneous support services			\$16	67.85	Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Miscellaneous support services				\$167.85	Hourly charge. Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Training				\$4,767.54	1 Day of On-site Gen5/Flyaway Training Labor.	
Equipment				\$610.40	Cradlepoint IBR650 + Laird Quad band Exterior Ant (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$1,338.40	Omni Directional Panorama LTE Antenna with installation (can only be sold as part of an overall/comprehensive data solution)	
Installation				\$201.60	Provisioning fee per wireline circuit	
Installation			Ш	\$201.60	Installation support for BYOB deployment	
Installation				\$397.60	Standard Wireline Install - 1 CPE and 1 wireline	
Installation			Ш	\$638.40	Standard Wireline Install - Single CPE and 2 wireline	
Installation				\$722.40	Standard Wireline Install - Dual CPE and 2 wireline	
Installation				\$319.20	Fixed 4G modem install: Standalone Visit or Primary Transport	
Installation				\$190.40	Fixed 4G modem install: Concurrent Visit	
Installation			П	\$280.00	Exterior 4G Antenna Install: Standalone Visit	
Installation				\$117.60	Exterior 4G Antenna Install: Concurrent Visit	
Installation				\$520.80	Deinstall/Reinstall H/W at same location	
Installation				\$397.60	Mall Installation additional cost per site	
Installation				\$302.40	Upgrade Installation - Onsite for one device or circuit	
Installation				\$201.60	Administration Fee for Contruction Build Out Costs	
Installation				\$117.60	Installation arrival prior to 8:00 AM	
Installation				\$1,148.00	Temp Wireless Install/Deinstall	
Installation			П	\$240.80	Temp Wireless Expedited Install less than 7 days - in addition to standard price	
Equipment				\$1,376.77	Equipment (.9 m Antenna, Radio and HT2010W Modem) and Enterprise Installation requires 12 month commitment (can only be sold as part of an overall/comprehensive data solution)	12 months

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment				\$670.88	Equipment (.9 m Antenna, Radio and HT2010W Modem) and Enterprise Installation requires 24 month commitment (can only be sold as part of an overall/comprehensive data solution)	24 months
Flayaway VSAT				\$20,718.88	Gen5/Flyaway, includes HT2010 or HT2010W or equivalent, .98m antenna, 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data solution)	
Auto-deployable VSAT				\$38,080.00	Gen5/Auto-Deployable, includes HT2010 or HT2010W or equivalent, .98m antenna (Ka-98H), 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data solution)	
Static IP Address per device	\$16.80	Static IP Address per device	П			
device			П			
VPN SERVICES PRI	CING					
Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
VPN capacity at Hughes NOC	\$5.59	Hughes NOC Internet/VPN Access (per Mbps)				

OTHER DATA SERVICES PRICING									
Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term			
Equipment				\$414.40	Managed Branch CPE - FortiGate 40F with FortiOS Native Mode (can only be sold as part of an overall/comprehensive data solution)				
Equipment				\$828.80	Managed Branch CPE - FortiGate 40F-3G4G with FortiOS Native Mode (can only be sold as part of an overall/comprehensive data solution)				

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment				\$582.40	Managed Branch CPE - FortiGate 60F with FortiOS Native Mode (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$20,468.00	FG-1000D-USG - 2 x 10GE SFP+ slots, 16 x GE SFP Slots, 16 x GE RJ45ports, 2 x GE RJ45 Management ports, 2 x 120GB SSDonboard storage, dual AC power supplies (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$604.80	FortiSwitch 124E-POE - requires MRC (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$330.40	FortiSwitch 124E-non POE - requires MRC (can only be sold as part of an overall/comprehensive data solution)	
Installation			П	\$117.60	Switch Install: Concurrent Visit	
Installation			П	\$123.20	Router Self Installation Support	
Managed Services	\$33.60	Managed Services Fee for Primary Transport only (service can only be sold as a part of an overall/comprehensive data solution)				
Managed Services	\$23.52	Circuits Only management incl NAP mgt, provisioning, bill consolidation and Tier 3 HD (service can only be sold as a part of an overall/comprehensive data solution)				
Managed Services	\$11.20	Managed Services Fee for Secondary Transport (service can only be sold as a part of an overall/comprehensive data solution)				
Support	\$5.60	FortiCare 8x5 FortiSwitch 124E - POE	П			
Support	\$2.24	FortiCare 8x5 FortiSwitch 124E - non POE				
Managed Security	\$7.84	Managed Security - FG-40F - Firewall (service can only be sold as a part of an overall/comprehensive data solution)				
Managed Security	\$22.40	Managed Security - FG-40F - Content Filtering (service can only be sold as a part of an overall/comprehensive data solution)				
Managed Security	\$28.00	Managed Security - FG-40F - UTP (service can only be sold as a part of an overall/comprehensive data solution)				
Managed Security	\$44.80	Managed Security - FG-40F - SIEM (service can only be sold as a part of an overall/comprehensive data solution)				

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Managed Security	\$61.60	Managed Security - FG-40F - SIEM with Incident Response (service can				
		only be sold as a part of an overall/comprehensive data solution)				
Managed Security	\$7.83	Managed Security - HRFG-60F -				
	<b>V</b> 55	Firewall (service can only be sold as a				
		part of an overall/comprehensive data				
Managed Security	\$22.39	solution)  Managed Security - HRFG-60F -				
Managed Security	Ψ22.59	Content Filtering (service can only be				
		sold as a part of an				
	007.00	overall/comprehensive data solution)				
Managed Security	\$27.99	Managed Security - HRFG-60F - UTP (service can only be sold as a part of				
		an overall/comprehensive data				
		solution)				
Managed Security	\$44.79	Managed Security - HRFG-60F - SIEM				
		(service can only be sold as a part of an overall/comprehensive data				
		solution)				
Managed Security	\$61.59	Managed Security - HRFG-60F - SIEM				
		with Incident Response (service can only be sold as a part of an				
		overall/comprehensive data solution)				
Managed Security	\$10.07	Managed Security - FG-40F-3G4G -				
		Firewall (service can only be sold as a				
		part of an overall/comprehensive data solution)				
Managed Security	\$39.19	Managed Security - FG-40F-3G4G -				
,		Content Filtering (service can only be				
		sold as a part of an				
Managed Security	\$44.79	overall/comprehensive data solution)  Managed Security - FG-40F-3G4G -				
Managed Occurry	Ψ+4.73	UTP (service can only be sold as a				
		part of an overall/comprehensive data				
Managad Casurity	\$61.59	solution)  Managed Security - FG-40F-3G4G -				
Managed Security	ф61.59	SIEM (service can only be sold as a				
		part of an overall/comprehensive data				
		solution)	_			
Managed Security	\$78.39	Managed Security - FG-40F-3G4G - SIEM with Incident Response (service				
		can only be sold as a part of an				
		overall/comprehensive data solution)				
On-site support	\$11.20	FortiGate 40F NBD				
On-site support	\$22.40	FortiGate 40F-3G4G NBD				
On-site support	\$11.20	FortiGate 60F NBD				
On-site support	\$14.56	VC510 NBD				
On-site support	\$64.96	VC3400 NBD				
Tier3-Prem-Single	\$2.24	Tier 3, Premium Support, Single	-			
Tiore i Tom Cingle	Ψ2.2 ·	Transport				
Tier1-Single	\$2.24	Tier 3 + Tier 1 Help Desk, Single				
Tier1-Prem-Single	\$4.48	Transport Tier 3 + Tier 1, Premium Support,				
norr-rem-single	φ4.40	Single Transport				
Tier1-Plat-Single	\$6.72	Tier 3 + Tier 1, Platinum Support,				
Tion Durant 4011441	00.00	Single Transport				
Tier3-Prem-4GHAN	\$3.36	Tier 3, Premium Support, 4G HAN Transport				
Tier1-4GHAN	\$4.48	Tier 3 + Tier 1 Help Desk, 4G HAN				
		Transport				
Tier1-Prem-4GHAN	\$7.84	Tier 3 + Tier 1, Premium Support, 4G				

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Tier1-Plat-4GHAN	\$12.32	Tier 3 + Tier 1, Platinum Support, 4G HAN Transport			
Equipment			\$543.20	VMware SD-WAN Edge 510 Appliance, Deployment: Hosted Orchestrator for 3 years (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$5,409.60	VMware SD-WAN Edge 3400 Appliance, Deployment: Hosted Orchestrator for 3 years (can only be sold as part of an overall/comprehensive data solution)	
VC010M-PRE-HO-HG- L34S1-36P	\$38.08	VMware SD-WAN 10 Mbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC030M-PRE-HO-HG- L34S1-36P	\$66.08	VMware SD-WAN 30 Mbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC050M-PRE-HO-HG- L34S1-36P	\$101.92	VMware SD-WAN 50 Mbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC100M-PRE-HO-HG- L34S1-36P	\$136.64	VMware SD-WAN 100 Mbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC200M-PRE-HO-HG- L34S1-36P	\$236.32	VMware SD-WAN 200 Mbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC001G-PRE-HO-HG- L34S1-36P	\$564.48	VMware SD-WAN 1 Gbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC002G-PRE-HO-HG- L34S1-36P	\$712.32	VMware SD-WAN 2 Gbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC005G-PRE-HO-HG- L34S1-36P	\$1,137.92	VMware SD-WAN 5 Gbps Premium Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
VC010G-PRE-HO-HG- L34S1-36P	\$1,767.36	VMware SD-WAN 10 Gbps Premium Edition Service Subscription for 3 years, Prepaid, Hosted Orchestrator with Hosted Gateway, Basic Support Backline (L3-4)			
MX68-HW			\$968.80	Meraki MX68 Router/Security Appliance (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
LIC-MX68-ENT-1YR	\$31.36	Meraki MX68 Enterprise License and	П			
	<b>A .</b>	Support, 1YR	Ц			
LIC-MX68-SEC-1YR	\$61.60	Meraki MX68 Advanced Security License and Support, 1YR	П			
LIC-MX68-SDW-1Y	\$106.40	Meraki MX68 Secure SD-WAN Plus	П			
		License and Support, 1YR	Ш			
MX450-HW				\$19,460.00	Meraki MX450 Router/Security Appliance (can only be sold as part of an overall/comprehensive data solution)	
LIC-MX450-ENT-1YR	\$812.00	Meraki MX450 Enterprise License and Support, 1YR				
LIC-MX450-SEC-1YR	\$1,622.88	Meraki MX450 Advanced Security License and Support, 1YR				
LIC-MX450-SDW-1Y	\$2,797.76	Meraki MX450 Secure SD-WAN Plus License and Support, 1YR	П			
MS120-24P-HW		Electise and Support, TTN		\$2,811.20	Meraki MS120-24P 1G L2 Cld -Mngd 24x GigE 370W PoE Switch (can only be sold as part of an overall/comprehensive data solution)	
LIC-MS120-24P-1YR	\$12.32	Meraki MS120-24P Enterprise License and Support, 1 Year	П			
FIXED SATELLITE	SERVICES I					
Plan Name/Title	MRC	MRC Description	П	NRC	NRC Description	Term
128 kbps transmit / 256	\$282.24	HX/HT or equivalent Circuit on Galaxy	Н	\$2,432.64	HX/HT or Equivalent VSAT	
kbps receive	,	16 satellite using HX/HT VSAT platform. Oversubscription is limited to 25:1. Download threshold is set at 220 Mbytes.		<b>*-</b> 7 · <b></b> ·	modem, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
256 kbps transmit / 512 kbps receive	\$433.44	HX/HT Circuit on Galaxy 16 satellite using HX/HT VSAT platform.  Oversubscription is limiited to 25:1.  Download threshold is set at 440  Mbytes.		\$2,432.64	HX/HT VSAT modem, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
256 kbps transmit / 1024 kbps receive	\$588.00	HX/HT Circuit on Galaxy 16 satellite using HX/HT VSAT platform.  Oversubscription is limiited to 25:1.  Download threshold is set at 880  Mbytes.		\$2,432.64	HX/HT VSAT modem, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
512 kbps transmit / 1544 kbps receive	\$891.52	HX/HT Circuit on Galaxy 16 satellite using HX/HT VSAT platform. Oversubscription is limiited to 25:1. Download threshold is set at 1340 Mbytes.		\$2,432.64	HX/HT VSAT modem, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
1024 kbps transmit / 5120 kbps receive	\$2,263.52	HX/HT Circuit on Galaxy 16 satellite using HX/HT VSAT platform.  Oversubscription is limiited to 25:1.  Download threshold is set at 4420  Mbytes.		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
128 kbps transmit	\$1,842.40	HX/HT dedicated inbound channel (VSAT to Hughes NOC) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
256 kbps transmit	\$3,684.80	HX/HT dedicated inbound channel (VSAT to Hughes NOC) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
512 kbps transmit	\$7,369.60	HX/HT dedicated inbound channel (VSAT to Hughes NOC) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard	
1024 kbps transmit	\$14,739.20	this channel.  HX/HT dedicated inbound channel (VSAT to Hughes NOC) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.	İ	\$2,432.64	Hughes installation.  HX/HT VSAT modem  VSAT, including indoor unit,  2 watt radio, .98 meter  antenna and standard  Hughes installation.	
128 kbps transmit	\$964.32	HX/HT dedicated outbound channel (Hughes NOC to VSAT) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.	Ì	\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
256 kbps transmit	\$1,928.64	HX/HT dedicated outbound channel (Hughes NOC to VSAT) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
512 kbps transmit	\$3,856.16	HX/HT dedicated outbound channel (Hughes NOC to VSAT) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,432.64	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
1024 kbps transmit	\$7,712.32	HX/HT dedicated outbound channel (Hughes NOC to VSAT) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,635.36	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
2048 kbps transmit	\$15,425.76	HX/HT dedicated outbound channel (Hughes NOC to VSAT) on Galaxy 16 satellite. Customer decides how many satellite terminals are to be assigned to this channel.		\$2,635.36	HX/HT VSAT modem VSAT, including indoor unit, 2 watt radio, .98 meter antenna and standard Hughes installation.	
Installation		uns channel.	Ĭ	\$492.80	incremental charge to install a 1.2 meter antenna instead of .98 meter antenna on the HN9500.	
Deicing				\$890.40	incremental charge for electrical antenna deicing (.98 meter antenna) at time of initial installation.	
Deicing				\$924.00	incremental charge for electrical antenna deicing (1.2 meter antenna) at time of initial installation.	
1 Mbps	\$5.59	Terrestrial based Internet capacity from the Hughes NOC. This is typically used to establish VPN connections between the Hughes NOC and the Customer data center.				
Network Infrastructure	\$1,428.00	Network Infrastructure - This includes a redundant pair of IP gateways and proactive network monitoring. Required on a Per Network basis for networks that use the Hughes NOC.				
Enterprise Service Plans for Hardware: HT						

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Enterprise 100	\$38.58	Enterprise 100 service suited for credit card and point of sale transactions. Requires HT VSAT modem and minimum of .98 m antenna. Maximum speeds: 64 kbps up/256 kbps down.Fair Access Policies apply. This service requires a Private IP Gateway implementation and Vision Access for customer provided Tier 1/Tier 2 support. This service plan requires 12 month commitment.			
Enterprise 200	\$44.68	Enterprise 200 service suited for credit card, polling PC remote control and Intranet. Requires HT VSAT modem and minimum of .98 m antenna.  Maximum speeds: 128 kbps up/j1024 kbps down. Fair Access Policies apply. This service requires a Private IP Gateway implementation and Vision Access for customer provided Tier 1/Tier 2 support. This service plan requires 12 month commitment			
Enterprise 300	\$66.00	Enterprise 300 service suited for credit card, polling PC remote control and Intranet. Requires HT VSAT modem and minimum of .98 m antenna.  Maximum speeds: 256 kbps up/1544 kbps down. Fair Access Policies apply. This service requires a Private IP Gateway implementation and Vision Access for customer provided Tier 1/Tier 2 support. This service plan requires 12 month commitment.			
Enterprise 800	\$203.11	Enterprise 800 service. Requires HT VSAT modem and minimum of a .98 m antenna. Maximum speeds: 512 kbps up/1544 kbps down. Fair Access Policies apply. This service requires a Private IP Gateway implementation and Vision Access for customer provided Tier 1/Tier 2 support. Requires a minimum order of 40 sites. This service plan requires 12 month commitment.			
Enterprise 900	\$406.24	Enterprise 900 service. Requires HT VSAT modem and minimum of a 1.2 m antenna. Typical throughput rates: 1024 kbps up/204 8kbps down. Fair Access Policies apply. This service requires a Private IP Gateway implementation and Vision Access for customer provided Tier 1/Tier 2 support. Requires a minimum order of 40 sites. This service plan requires 12 month commitment.			

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment				\$2,540.51	HT VSAT Modem, .98m antenna, 2 watt rfu, standard installation. Customer equipment for use with HughesNet broadband IP Internet Access (IA) or Enterprise plans. Includes standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$2,752.66	HT VSAT Modem 1.2m antenna, 2 watt rfu, standard installation. Customer equipment for use with HughesNet broadband IP Internet Access (IA) or Enterprise plans. Includes standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$3,711.58	HT VSAT Modem, 1.8m antenna, 4 watt rfu, standard installation. Customer equipment for use with HughesNet broadband IP Internet Access (IA) or Enterprise plans. Includes standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Remote Maintenance			П		Solution	
Field Service Next Business Day	\$22.68	Field Service - Coverage Hours are 8:00 a.m. to 5:00 p.m., local time, Monday through Friday, holidays excepted, beginning the next business day after a call is received.				
Field Service Next Day	\$26.07	Field Service - Coverage Hours are 8:00 a.m. to 5:00 p.m., local time, 365 days per year, beginning the next day after a call is received.				
Field Service 9 x 5	\$26.69	Field Service - Coverage Hours are 8:00 a.m. to 5:00 p.m., local time, Monday through Friday, holidays excepted. Next business day coverage.				
Field Service 12 x 6	\$27.96	Field Service - Coverage Hours are 8:00 a.m. to 8:00 p.m., local time, Monday through Saturday				
Field Service 12 x 7	\$29.18	Field Service - Coverage Hours are 8:00 a.m. to 8:00 p.m., 365 days per year				
Field Service 18 x 7	\$30.44	Field Service - Coverage Hours are 6:00 a.m. to 12:00 midnight, 365 days per year				
Field Service 24 x 7	\$31.96	Field Service - Coverage Hours are 24 hours per day, 365 days per year				
Early Termination Charge	\$287.76	Early Termination charge applies to service cancelled prior to (12) months of service.				

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Vehicle-mounted						
Redeployable Solutions			4			
Equipment				\$39,200.00	Gen5/Auto-Deployable, includes HT2010 or HT2010W or equivalent, .98m antenna (Ka-98H), 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data	
					solution)	
Equipment				\$28,076.73 \$34,712.66	HN9460 with 0.98m auto-deployable antenna system with controller and 30ft cable, 2W BUC, standard shipping, installation, travel expenses and program management for use with HughesNet Ku Enterprise service plans. Two year warranty. (can only be sold as part of an overall/comprehensive data solution) HX90 with 1.2m auto-deployable antenna system	
					with controller and 30ft cable, 2W BUC, standard shipping, installation, travel expenses and program management for use with Hughes HX service plans. Two year warranty. (can only be sold as part of an overall/comprehensive data solution)	
Fly-away Redeployable				\$0.00		
Packages Equipment				\$59,519.68	HX200 flyaway package including 1.2m autodeployable antenna system with controller and 30ft cable, tripod mount, four cases, 6W BUC, program management, standard shipping and professional setup for use with Hughes HX service plans. Two year warranty. (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment				\$20,718.88	Gen5/Flyaway, includes HT2010 or HT2010W or equivalent, .98m antenna, 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data solution)	
Redeployable Support			Н			
тесеріоуавіс бирроп	\$4,390.40	Engineering Support beyond the modem (includes RFT spares refurb, semi annual onsite preventive maitenace, and 7x24 telephone support) - per month				
Maintenace				\$1,400.00	Emergency On-Site Maintenance - per day	
On-site support				\$1,624.00	On-Site Operations Engineer for non- operational issues - per day	
Labor (Move, Add, Change Services)						
Site survey				\$335.72	Site survey to determine the time and materials to install HughesNet equipment. Quotation provided from results of survey.	
Standard Fixed VSAT Installation : 98/120				\$863.27	Standard Fixed HughesNet equipment Installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Standard Fixed VSAT Installation : 180				\$1,472.64	Standard Fixed HughesNet equipment Installation for sites requiring 1.8 meter antenna systems in the lower 48 states.	
Standard Fixed VSAT De-installation : 98/120				\$457.03	Standard Fixed HughesNet equipment De-installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Standard Fixed VSAT De-installation : 180				\$761.60	Standard Fixed HughesNet equipment De-installation for sites requiring 1.8 meter antenna systems in the lower 48 states.	
Re-install .98m/1.2m antenna				\$812.00	Standard Fixed VSAT Re- installation .98m/1.2m	
Re-install 1.8m antenna				\$1,371.06	Standard Fixed VSAT Re- installation 1.8m	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Standard Fixed VSAT Site Relocation/Move - Local : 98/120				\$1,269.52	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems and within same city or up to 100 miles between the existing and new site in the lower 48 states.	
Standard Fixed VSAT Site Relocation/Move - Local : 180				\$2,132.79	Standard Fixed HughesNet equipment site move for 1.8 meter antenna systems and within same city or up to 100 miles between the existing and new site in the lower 48 states.	
Standard Fixed VSAT Site Relocation/Move - Non-local : 98/120				\$1,523.42	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems where by the existing and new HughesNet equipment sites are more than 100 miles apart in the lower 48 states.	
Standard Fixed VSAT Site Relocation/Move - Non-local : 180				\$2,539.04	Standard Fixed HughesNet equipment site move for 1.8 meter antenna systems where by the existing and new HughesNet equipment sites are more than 100 miles apart in the lower 48 states.	
Incremental charge for customer requested/required pole mount up to 10 feet for VSAT: 0.98m or 1.2 m				\$364.49	Incremental charge for customer requested/required pole mount with height up to 10 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Incremental charge for customer requested/required pole mount with height 11-15 feet for VSAT: 0.98 m or 1.2 m				\$537.15	Incremental charge for customer requested/required pole mount with height 11 – 15 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Re-pointing/Re-cross- polarization of VSAT antenna				\$335.72	Charge to repoint/repolorize the antenna at the same customer location	
Move indoor electronics - VSAT Move indoor electronics -				\$335.72	same point of entry and cable will be used	
VSAT Upgrade indoor				\$383.68 \$335.72	new point of entry and cable is required  Does not include hardware	
electronics or outdoor electronics - VSAT			$\perp \parallel$			
Upgrade indoor electronics and outdoor electronics - VSAT				\$383.68	Does not include hardware	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Aborted or Cancelled Installation: VSAT and DSL				\$287.76	This charge only applies in cases where the installation has been cancelled by Customer within seven (7) days of the scheduled installation.	
Expedited Installation Charge: DSL and VSAT				\$335.72	Incremental charge for installations requested with less than 15 business days of notice.	
Dedicated twisted pair line install extra cost > 100'				\$1.20	Additional Cat 5 data cable installed > 100'	
Miscellaneous support services				\$167.85	Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Miscellaneous support services				\$167.85	Hourly charge. Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Service Plan Change (Upgrade/Downgrade) Fee				\$335.72	Charge to change from one service plan to another. Requires technician visit to customer's location to decommission the old service plan and re-commission the new service plan.	
			Ц			
Space Segment and Capacity Management			Ш			
Space Segment and Capacity Management	\$5,166.56	Transponder capacity, 1 year, Ku band, 1.0MHz, based on the greater of Bandwidth or Power Equivalent Bandwidth (PEB)				
Space Segment and Capacity Management	\$5,010.88	Transponder capacity, 2 year, Ku band, 1.0MHz, based on the greater of Bandwidth or Power Equivalent Bandwidth (PEB)				
Space Segment and Capacity Management	\$4,888.80	Transponder capacity, 3 year, Ku band, 1.0MHz, based on the greater of Bandwidth or Power Equivalent Bandwidth (PEB)				
Space Segment and Capacity Management	\$4,790.24	Transponder capacity, 4 year, Ku band, 1.0MHz, based on the greater of Bandwidth or Power Equivalent Bandwidth (PEB)				
Baseband Equipment			П			
Baseband Equipment (1039685-1060 or equivalent)				\$309,220.80	Hughes Baseband NOC; Rack, Assy, HX-45U: 1039685-1060, L Band rev6 (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Baseband Equipment (1501821-8004 or equivalent)				\$12,233.76	Spare Configurable Demodulator Subsystem (CDS), ASSY, CDS FRU 10MSPS WIN2008 (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0504 or equivalent)				\$12,407.36	Redundant IP Gateway Servers (IPGW); ASSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0504 or equivalent)				\$12,407.36	Redundant IP Gateway Servers (IPGW); ASSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0504 or equivalent)				\$14,000.00	Redundant VPN Gateway Servers (VPGW), ASSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0504 or equivalent)				\$6,203.68	Virtual Automated Dial Backup Server (VADB); SSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1036077-0004 or equivalent)				\$101,935.68	Auto Cross Pole System (2 G8 Server and 2 Spectrum Analyzers Narrow Res-BW) (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment			П	\$183,680.00	On-Site Installation	
Baseband Equipment (1504140-0504 or equivalent)				\$14,000.00	Redundant VPN Gateway Servers (VPGW), ASSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0504 or equivalent)				\$6,203.68	Spare IP Gateway Server (IPGW), ASSY, SVR G8 DL360E, HN IPGW (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0502 or equivalent)				\$8,450.40	Spare Satellite Gateway Modulator (SGM) (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Baseband Equipment (1501821-8004 or equivalent)			\$13,440.00	Spare Configurable Demodulator Subsystem (CDS), ASSY, CDS FRU 10MSPS WIN2008 (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0501 or equivalent)			\$10,536.96	Spare Dynamic Network Control Cluster (DNCC), ASSY, SVR G8, DL360E, HN DNCC (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1500229-5001 or equivalent)			\$1,983.52	Spare Timing Unit (TU), CHASSIS ASSY, HX DUAL TIMING IRU (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1033731-6011 or equivalent)			\$8,064.00	Spare Timing Generator (TG2), ASSY, TG2 w/o 100 MHz R5/6 (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9502098-0001 or equivalent)			\$494.69	Spare Power Distribution Unit (PDU), POWER DIST UNIT, INTL, 16 outlets, 85/264VAC 30A (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504132-0151 or equivalent)			\$297.77	Extra Power Supply Units for Spare Servers (PSU), Power Supply, 460W, Hot Plug, 94% (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9501330-0002 or equivalent)			\$163.52	Spare PDU cable, PDU CABLE 9' (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9501330-0002 or equivalent)			\$297.77	Extra PSU for Production Servers, Power Supply, 460W, Hot Plug, 94% (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504132-0151 or equivalent)			\$494.69	Extra PDU cable, PDU CABLE 9' (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9502098-0001 or equivalent)			\$813.73	Extra Ethernet Cables (Green, 6ft), CABLE, CAT5E, 4PR, RJ45, GREEN, 6FT (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Baseband Equipment (EM7 or equivalent)	\$9,800.00	ScienceLocgic EM7 (includes all in one server, platform license, 50 device licenses, and onsite deployment) - annual cost		\$36,769.49	ScienceLocgic EM7 (includes all in one server, platform license, 50 device licenses, and onsite deployment) - annual cost	
Baseband Equipment (NKN- 13XK25VYH8T514C2 or equivalent)				\$30,486.86	Niksun, 2830 Series NetVCR appliance (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504140-0013 or equivalent)				\$34,090.15	NMS (Vision) Server (HP Proliant DL360e Gen 8 ) (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1036077-0004 or equivalent)				\$6,203.68	ACP Server (HP Proliant DL360e Gen 8) (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504650-0003 or equivalent)				\$10,420.21	CISCO 3850 48 port switch (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9012249-0004 or equivalent)				\$29,932.93	Spectrum Analyzer (Keysight E4402B) (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (9504917-0001 or equivalent)				\$6,097.06	L-band Splitter (Active Splitter,8-way,ADJ Gain,18V LNB) From ETL (can only be sold as part of an overall/comprehensive data solution)	
Baseband Equipment (1504132-0131 or equivalent)				\$310.59	Secondary Hard Drive for NMS server (can only be sold as part of an overall/comprehensive data solution)	
Baseband Maintenance	\$10,080.00	Engineering Support for a dedicated HN or HX System from the modem to the ethernet interface (includes Baseband refurb, emergency onsite maitenance, and 7x24 telephone support) - per month				
RFT Equipment			П			
Equipment				\$2,110.00	Dell Optiplex 7020 workstation (Compass) (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$12,079.75	Dow-Key Microwave Matrix MP-Series RF Switch (Compass) (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$2,564.80	Miteq, UPC Channel Card (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Equipment			\$14,529.76	Comtech UT-4514, Up- Converter Primary (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$14,101.92	Comtech UT-4514, Up- Converter Redundant (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$940.80	S.M. Microwave, Ku Band Signal Combiner (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$55,128.64	CPI VZU-6997AD, 750W TWTA HPA (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$9,657.76	Dummy load and CPI 01032330-01 redundancy switch module for CPI VZU- 6997AD, 750W TWTA HPA (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$4,272.80	HPA 1:1 Controller-Spare for the CPI VZU-6997AD, 750W TWTA HPA (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$14,529.76	Comtech DT-4511, Down- Converter Primary (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$14,101.92	Comtech DT-4511, Down- Converter Redundant (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$7,692.16	SSC 3430L, Beacon Tracking Receiver (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$4,272.80	LNA 1:1 Controller (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$9,925.44	Down Converter, Miteq DNB1-12.1TR (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$4,009.56	Down converter Switch, Cables, rack mount (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment				\$25,903.36	RFT Design and Engineering for RFT Equipment excluding antenna)	
Equipment			П	\$48,052.48	On-Site Installation (Excludes Antenna)	
Equipment				\$5,488.00	ADH NETCON Waveguide dehydrator (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$7,692.16	Beacon Receiver, SSC 3430-KU-E (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$78,394.50	Compass M&C with 11 input switch (can only be sold as part of an overall/comprehensive data solution)	
RFT Maintenance			Ш			
Maintenance	\$3,785.60	Compass Support (Annual Cost) includes customer support via email and telephone, bug fixes, software upgrades, and access to any available driver from the Compass library.				
RFT Maintenance		,	П			
Remotes			П			
Equipment				\$1,548.96	HX200 - IDU Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$616.00	HX50L - IDU Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$616.00	HN7700 - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$616.00	HN9460 - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
Equipment			9	\$560.00	Ku Band 2 Watt Outdoor Radio Unit - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment			9	\$436.80	1.2m Ku Band Antenna - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment			4	\$896.00	Equipment Only. Customer equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment			9	§168.00	Standard Mount 1.2m Ku Band Antenna - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment			9	\$448.00	Standard Mount 1.8m Ku Band Antenna - Equipment Only. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment				\$1.12	Extended IFL Cables (>300') - RG-6 Quad Shield w/ solid copper center conduit (dual line) 368 feet max. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Equipment			\$2.80	Extended IFL Cables (>300') - Plenum Rated RG-6 w/ solid copper center conductor (dual line) 368 feet max. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$2.24	Extended IFL Cables (>300') - RG-11 w/ solid copper center conductor (dual line) 593 feet max. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data	
Equipment			\$4.48	solution) Extended IFL Cables (>300') - Plenum Rated RG- 11 w/ solid copper center conductor (dual line) 593 feet max. Customer equipment for use with a Hughes Ku Access Plan. Does not include standard installation. (can only be sold as part of an overall/comprehensive data	
Demand Services and Time & Materials Repairs				solution)	
repair			\$86.24	Time & Materials Repair - DISTANCE ZONE CHARGE 0 - 25 Miles - per occurrence	
repair			\$160.16	Time & Materials Repair - DISTANCE ZONE CHARGE 25 - 50 Miles - per occurrence	
repair			\$320.32	Time & Materials Repair - DISTANCE ZONE CHARGE 51 - 100 Miles - per occurrence	
repair			\$529.76	Time & Materials Repair - DISTANCE ZONE CHARGE 101+ Miles - per occurrence	
repair			\$280.00	Time & Materials Repair - Re-Point / Re-Range site visit - per occurrence	
Training			\$0.00		
training			\$1,568.00	HN or HX System Introduction (2 days) at Hughes facility - per student (Travel Not Included)	

Plan Name/Title	MRC	MRC Description		NRC	NRC Description	Term
training				\$784.00	HN or HX Satellite Router Installation and Operations (1 days) at Hughes facility - per student (Travel Not Included)	
training				\$2,352.00	HN or HX Network Operations (3 days + weekend) at Hughes facility - per student (Travel Not Included)	
training				\$3,136.00	HN or HX Configure and Design (4 days) at Hughes facility - per student (Travel Not Included)	
training				\$2,016.00	On-Site RFT Training (3 days) - per class	
training				\$3,696.00	Extended Training - per day	
training				\$2,240.00	Annual Hughes User Group Conference (Customers will receive two complete conference registrations per year for every 800 remote sites) (Per Person Charge)	
training				\$4,767.54	1 Day of On-site Gen5/Flyaway Training Labor.	
Disaster Recovery			П			
Disaster Recovery				\$11,200.00	Disaster Recovery Initial Design and Engineering	
Disaster Recovery	\$12,320.00	Disaster Facilities Reservation	П			
Disaster Recovery				\$11,200.00	Disaster Declaration Implementation	
Disaster Recovery	\$5,600.00	Disaster Facilities Operation - per day	П			
Disaster Recovery	\$448.00	Disaster Recovery Testing, in excess of eight hours per six months - per hour				
Additional Items	\$0.00		П			
Business 35	\$74.24	Jupiter 25/3 Mbps with 25GB Business Usage and 10GB Anytime Usage				
Business 50	\$106.06	Jupiter 25/3 Mbps with 25GB Business Usage and 25GB Anytime Usage	П			
Business 75	\$159.11	Jupiter 25/3 Mbps with 25GB Business Usage and 50GB Anytime Usage				
Business 100	\$212.14	Jupiter 25/3 Mbps with 25GB Business Usage and 75GB Anytime Usage	П			
Business 150	\$291.42	Jupiter 25/3 Mbps with 50GB Business Usage and 100GB Anytime Usage				
Business 250	\$507.29	Jupiter 25/3 Mbps with 50GB Business Usage and 200GB Anytime Usage				
Business 500	\$1,028.88	Jupiter 25/3 Mbps with 50GB Business Usage and 450GB Anytime Usage				
Token for HTS Plan per GB	\$3.36					
1 Static IP	\$34.45	1 Static IP (HughesNet VSAT Services)				
5 Static Ips	\$57.41	5 Static Ips (HughesNet Vsat Services)	$\prod$			

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Equipment			\$1,376.77	Equipment (.9 m Antenna, Radio and HT2010W Modem or equivalent) and Enterprise Installation (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$670.88	Equipment (.9 m Antenna, Radio and HT2010W Modem or equivalent) and Enterprise Installation (can only be sold as part of an overall/comprehensive data solution)	
Equipment			\$20,718.88	Gen5/Flyaway, includes HT2010 or HT2010W or equivalent, .98m antenna, 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data	
Equipment			\$38,080.00	Gen5/Auto-Deployable, includes HT2010 or HT2010W or equivalent, .98m antenna (Ka-98H), 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans. (can only be sold as part of an overall/comprehensive data	
(Optional) Field Service - Next Business Day	\$22.68	(Optional) Field Service - Next Business Day		(Solution)	
(Optional) Field Service - Next Celendar Day	\$26.07	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, Monday through Friday, holidays excepted, beginning the next business day after a call is received.			
(Optional) Field Service - Same Day 8AM - 5PM	\$29.18	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year, beginning the next business day after a call is received.			
(Optional) Field Service - Same Day 24 x 7	\$31.96	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year.			
Remote Field Service, Hourly Charge			\$139.08	Field Service – Coverage Hours are 24 hours per day, 365 days per year.	
Remote Field Service, Hourly Charge			\$167.85	Hourly labor rate for field service technician for work during standard business hours. Only applies if customer does not have a maintenance plan.	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Remote Field Service, Trip Charge			\$177.45	Hourly labor rate for field service technician for work not during standard business hours. Only applies if customer does not have a maintenance plan.	
Expedited Installation Charge			\$335.72	Expedited Installation Charge	
Early Termination Charge (prior to 12 months)			\$287.76	Incremental charge for installation requested with less than (15) business days notice.	
Site survey			\$335.72	Site survey to determine the time and materials to install HughesNet equipment. Quotation provided from results of survey.	
Standard Fixed VSAT Installation : 98/120			\$863.27	Standard Fixed HughesNet equipment Installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Standard Fixed VSAT De-installation : 98/120			\$457.03	Standard Fixed HughesNet equipment De-installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Re-install .98m/1.2m antenna			\$812.02	Standard Fixed VSAT Re- installation .98m/1.2m	
Standard Fixed VSAT Site Relocation/Move - Local : 98/120			\$1,269.52	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems and within same city or up to 100 miles between the existing and new site in the lower 48 states.	
Standard Fixed VSAT Site Relocation/Move - Non-local : 98/120			\$1,523.42	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems where by the existing and new HughesNet equipment sites are more than 100 miles apart in the lower 48 states.	
Incremental charge for customer requested/required pole mount up to 10 feet for VSAT: 0.98m or 1.2 m			\$364.49	Incremental charge for customer requested/required pole mount with height up to 10 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Incremental charge for customer requested/required pole mount with height 11-15 feet for VSAT: 0.98 m or 1.2 m			\$537.15	Incremental charge for customer requested/required pole mount with height 11 – 15 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Re-pointing/Re-cross- polarization of VSAT antenna			\$335.72	Charge to repoint/repolorize the antenna at the same customer location	
Move indoor electronics - VSAT			\$335.72	same point of entry and cable will be used	
Move indoor electronics - VSAT			\$383.68	new point of entry and cable is required	
Upgrade indoor electronics or outdoor electronics - VSAT			\$335.72	Does not include hardware	
Upgrade indoor electronics and outdoor electronics - VSAT			\$383.68	Does not include hardware	
Aborted or Cancelled Installation: VSAT			\$287.76	This charge only applies in cases where the installation has been cancelled by Customer within seven (7) days of the scheduled installation.	
Dedicated twisted pair line install extra cost > 100'			\$1.20	Additional Cat 5 data cable installed > 100'	
Service Plan Change (Upgrade/Downgrade) Fee			\$335.72	Charge to change from one service plan to another. Requires technician visit to customer's location to decommission the old service plan and re-commission the new service plan.	
Miscellaneous support services			\$167.85	Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Miscellaneous support services			\$167.85	Hourly charge. Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Training			\$4,767.54	1 Day of On-site Gen5/Flyaway Training Labor.	

## PRICING WITH DIR ADMINISTRATIVE FEE (CUSTOMER PAYS THIS AMOUNT)

**Geographical Area: Texas** 

## **BUSINESS INTERNET SERVICES PRICING**

Services below this line are directly billed by Vendor with a 4% Admin Fee.

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Business 35	\$68.94	Jupiter 25/3 Mbps with 25GB Business Usage and 10GB Anytime Usage			
Business 50	\$98.49	Jupiter 25/3 Mbps with 25GB Business Usage and 25GB Anytime Usage			
Business 75	\$147.74	Jupiter 25/3 Mbps with 25GB Business Usage and 50GB Anytime Usage			
Business 100	\$196.99	Jupiter 25/3 Mbps with 25GB Business Usage and 75GB Anytime Usage			
Business 150	\$270.61	Jupiter 25/3 Mbps with 50GB Business Usage and 100GB Anytime Usage			
Business 250	\$471.06	Jupiter 25/3 Mbps with 50GB Business Usage and 200GB Anytime Usage			
Business 500	\$955.39	Jupiter 25/3 Mbps with 50GB Business Usage and 450GB Anytime Usage			
Token for HTS Plan per GB	\$3.12				
1 Static IP	\$31.99	1 Static IP (HughesNet VSAT Services)			
5 Static Ips	\$53.31	5 Static Ips (HughesNet Vsat Services)			
Equipment			\$1,278.43	Equipment (.9 m Antenna, Radio and HT2010W Modem or equivalent) and Enterprise Installation	12 months
Equipment			\$622.96	Equipment (.9 m Antenna, Radio and HT2010W Modem or equivalent) and Enterprise Installation	24 months
Equipment			\$19,238.96	Gen5/Flyaway, includes HT2010 or HT2010W or equivalent, .98m antenna, 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans.	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Equipment			\$35,360.00	Gen5/Auto-Deployable, includes HT2010 or HT2010W or equivalent, .98m antenna (Ka-98H), 2 watt, and Accessories, Hardware Only, shipped. Does not include installation, travel expenses and program management for use with HughesNet Gen5 for Government service plans.	
(Optional) Field Service - Next Business Day	\$21.06	(Optional) Field Service - Next Business Day			
(Optional) Field Service - Next Celendar Day	\$24.21	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, Monday through Friday, holidays excepted, beginning the next business day after a call is received.			
(Optional) Field Service - Same Day 8AM - 5PM	\$27.09	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year, beginning the next business day after a call is received.			
(Optional) Field Service - Same Day 24 x 7	\$29.68	Field Service – Coverage Hours are 8:00AM to 5:00PM local time, 365 days per year.			
Remote Field Service, Hourly Charge			\$129.15	Field Service – Coverage Hours are 24 hours per day, 365 days per year.	
Remote Field Service, Hourly Charge			\$155.86	Hourly labor rate for field service technician for work during standard business hours. Only applies if customer does not have a maintenance plan.	
Remote Field Service, Trip Charge			\$164.78	Hourly labor rate for field service technician for work not during standard business hours. Only applies if customer does not have a maintenance plan.	
Expedited Installation Charge			\$311.74	Expedited Installation Charge	
Early Termination Charge (prior to 12 months)			\$267.21	Incremental charge for installation requested with less than (15) business days notice.	
Site survey			\$311.74	Site survey to determine the time and materials to install HughesNet equipment. Quotation provided from results of survey.	

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Standard Fixed VSAT Installation : 98/120			\$801.61	Standard Fixed HughesNet equipment Installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Standard Fixed VSAT De-installation : 98/120			\$424.38	Standard Fixed HughesNet equipment De-installation for sites requiring .98 and 1.2 meter antenna systems in the lower 48 states.	
Re-install .98m/1.2m antenna			\$754.02	Standard Fixed VSAT Re- installation .98m/1.2m	
Standard Fixed VSAT Site Relocation/Move - Local : 98/120			\$1,178.84	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems and within same city or up to 100 miles between the existing and new site in the lower 48 states.	
Standard Fixed VSAT Site Relocation/Move - Non-local : 98/120			\$1,414.61	Standard Fixed HughesNet equipment site move for .98 and 1.2 meter antenna systems where by the existing and new HughesNet equipment sites are more than 100 miles apart in the lower 48 states.	
Incremental charge for customer requested/required pole mount up to 10 feet for VSAT: 0.98m or 1.2 m			\$338.46	Incremental charge for customer requested/required pole mount with height up to 10 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Incremental charge for customer requested/required pole mount with height 11-15 feet for VSAT: 0.98 m or 1.2 m			\$498.78	Incremental charge for customer requested/required pole mount with height 11 – 15 feet. Cable run from antenna to indoor unit up to 150 feet with no trenching or excavation.	
Re-pointing/Re-cross- polarization of VSAT antenna			\$311.74	Charge to repoint/repolorize the antenna at the same customer location	
Move indoor electronics - VSAT			\$311.74	same point of entry and cable will be used	
Move indoor electronics - VSAT			\$356.27	new point of entry and cable is required	
Upgrade indoor electronics or outdoor electronics -			\$311.74	Does not include hardware	
VSAT VSAT					

Plan Name/Title	MRC	MRC Description	NRC	NRC Description	Term
Upgrade indoor electronics and outdoor electronics - VSAT			\$356.27	Does not include hardware	
Aborted or Cancelled Installation: VSAT			\$267.21	This charge only applies in cases where the installation has been cancelled by Customer within seven (7) days of the scheduled installation.	
Dedicated twisted pair line install extra cost > 100'			\$1.11	Additional Cat 5 data cable installed > 100'	
Service Plan Change (Upgrade/Downgrade ) Fee			\$311.74	Charge to change from one service plan to another. Requires technician visit to customer's location to decommission the old service plan and re-commission the new service plan.	
Miscellaneous support services			\$155.86	Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Miscellaneous support services			\$155.86	Hourly charge. Including but not limited to obtain permits and approvals, provision of specially required additional or special documentation, attendance at meeting before local planning or zoning boards, etc.	
Training			\$4,427.00	1 Day of On-site Gen5/Flyaway Training Labor.	

	Item	Amount (percentage or fixed)	Statutory Citation or Authority to assess	DIR Exemption Review (DIR use only)
	All	USF is calaculated according to FCC guidelines	https://www.fcc.gov/general/universal-service	
ı				

Attachment B-3 to Exhibit B

Hughes Network Systems, LLC

Contract DIR-TELE-CTSA-007

## PRICING WITH DIR ADMINISTRATIVE FEE (CUSTOMER PAYS THIS AMOUNT)

**Geographical Area: Texas** 

## **BROADBAND GLOBAL AREA NETWORK SERVICES PRICING**

Services beow this line are billed directly by Vendor with a 2% Admin Fee

			DIR Contract			
Plan Name/Title	MRC	MRC Description	Price	NRC	NRC Description	Term
Single Site Telephone				\$1.00	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time					Plan- Air Time Monthly Usage	
Monthly Usage (MB)					(MB)	
Voice to fixed					Voice to fixed	
Single Site Telephone				\$1.31	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time				· ·	Plan- Air Time Monthly Usage	
Monthly Usage (MB)					(MB)	
Voice to cellular					Voice to cellular	
Single Site Telephone				\$0.87	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time				ψο.στ	Plan- Air Time Monthly Usage	12 1110111110
Monthly Usage (MB)					(MB)	
Voice BGAN to					Voice BGAN to	
BGAN/FBB/SBB/GSPS					BGAN/FBB/SBB/GSPS	
Single Site Telephone				\$0.84	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time				φ0.04		12 1110111115
					Plan- Air Time Monthly Usage	
Monthly Usage (MB)					(MB)	
Voicemail				CC C4	Voicemail	40
Single Site Telephone				\$6.61	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time					Plan- Air Time Monthly Usage	
Monthly Usage (MB)					(MB)	
ISDN/Fax/Audio 3.1/MtM					ISDN/Fax/Audio 3.1/MtM	
O'a ala O'a Talanhana				00.54	Oire de Oire Telenhene Drieine	40
Single Site Telephone				\$0.54	Single Site Telephone Pricing	12 months
Pricing Plan- Air Time					Plan- Air Time Monthly Usage	
Monthly Usage (MB)					(MB)	
SMS Charges Public Static IP	\$16.32	Public Static IP			SMS Charges	12 months
Public Dynamic IP	\$13.26 	Public Dynamic IP				12 months
VPN Set Up Fee				\$841.50	VPN Set Up Fee	12 months
VPN Fee	\$76.50	VPN Fee				12 months
RTM Full Fee	\$9.18	RTM Full Fee				12 months
SIM Activation Fee				\$100.98	SIM Activation Fee	12 months
Single Site Stepped Data	\$85.68	Single Site Stepped Data Pricing				12 months
Pricing Plan- Airtime		Plan- Airtime Monthly Usage (MB)				
Monthly Usage (MB)		Monthly Base Fee				
Monthly Base Fee						
,						
Single Site Stepped Data	\$338.64	Single Site Stepped Data Pricing				12 months
Pricing Plan- Airtime		Plan- Airtime Monthly Usage (MB)				
Monthly Usage (MB)		1MB - 80MB				
1MB - 80MB						
Single Site Stepped Data	\$580.38	Single Site Stepped Data Pricing				12 months
	ψ500.50					12 110111118
Pricing Plan- Airtime		Plan- Airtime Monthly Usage (MB) 80MB - 230MB				
Monthly Usage (MB)		OUIVID - ZOUIVID				
80MB - 230MB						
Single Site Stepped Data	\$965.94	Single Site Stepped Data Pricing				12 months
Pricing Plan- Airtime		Plan- Airtime Monthly Usage (MB)				
Monthly Usage (MB)		230MB - 480MB				
230MB - 480MB						
· · - · · - · · · -		I	1	1	I	1

Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 480MB - 980MB	\$1,449.42	Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 480MB - 980MB			12 months
Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 980MB - 4980MB	\$2,897.82	Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 980MB - 4980MB			12 months
Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 4980MB - 9,980MB	\$3,863.76	Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 4980MB - 9,980MB			12 months
Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 9,980MB - 30,000MB	\$5,795.64	Single Site Stepped Data Pricing Plan- Airtime Monthly Usage (MB) 9,980MB - 30,000MB			12 months
9211 Portable BGAN HDR Terminal			\$4,078.98	9211 Portable BGAN HDR Terminal - 9211 Terminal (can only be sold as part of an overall/comprehensive data solution)	
9211 Portable BGAN HDR Terminal			\$610.98	9211 Portable BGAN HDR Terminal - 9211 Extended Warranty, add'l 18 months	
9211 Portable BGAN HDR Terminal			\$814.98	9211 Portable BGAN HDR Terminal - 9211 Extended Warranty, add'l 42 months	
9202M Portable BGAN Terminal			\$2,803.98	9202M Portable BGAN Terminal - 9202M Terminal (can only be sold as part of an overall/comprehensive data solution)	
9202M Portable BGAN Terminal			\$508.98	9202M Portable BGAN Terminal - 9202M Extended Warranty, add'l 18 months	
9202M Portable BGAN Terminal			\$712.98	9202M Portable BGAN Terminal - 9202M Extended Warranty, add'l 42 months	
Portable Terminal Accessories			\$219.30	Portable Terminal Accessories - Spare Battery Pack (can only be sold as part of an overall/comprehensive data solution)	
Portable Terminal Accessories			\$40.80	Portable Terminal Accessories - AC/DC adapter (can only be sold as part of an overall/comprehensive data solution)	
Portable Terminal Accessories			\$4.08	Portable Terminal Accessories - US AC Power Cord (can only be sold as part of an overall/comprehensive data solution)	
Portable Terminal Accessories			\$147.90	Portable Terminal Accessories - DC/DC Vehicular Adapter (can only be sold as part of an overall/comprehensive data solution)	
Portable Terminal Accessories			\$30.60	Portable Terminal Accessories - 30m Extension Cord - DC power (can only be sold as part of an overall/comprehensive data solution)	

Portable Terminal	7 1	\$5,098.98 Portable Terminal Accessories -	
Accessories		\$5,098.98 Portable Terminal Accessories - C10 Antenna with mag mounts	
Accessories		(can only be sold as part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$5,149.98 Portable Terminal Accessories -	
Accessories		C10 Antenna Kit (with RF	
7.0000001100		cables and mag mounts) (can	
		only be sold as part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$223.38 Portable Terminal Accessories -	
Accessories		C10 Antenna mag mounts (can	
		only be sold as part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$2,344.98 Portable Terminal Accessories -	
Accessories		C11 Antenna (mag mounts not	
		included) (can only be sold as	
		part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$2,599.98 Portable Terminal Accessories -	
Accessories		C11 Antenna Kit (with RF	
		cables and mag mounts) (can	
		only be sold as part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$202.98 Portable Terminal Acc (can only	
Accessories		be sold as part of an	
7 10000001100		overall/comprehensive data	
		solution)essories - C11 Antenna	
		Mag Mounts (can only be sold	
		as part of an	
		overall/comprehensive data	
		solution)	
Portable Terminal		\$25.50 Portable Terminal Accessories -	
Accessories		RF pigtail for external C10 or	
		C11 antenna (can only be sold	
		as part of an	
		overall/comprehensive data	
		solution)	
9410 All-In-One Mobile		\$4,282.98 9410 All-In-One Mobile BGAN	
BGAN Terminal		Terminal - 9410W Mobile BGAN	
		Terminal, Wi-Fi & Ethernet (can	
		only be sold as part of an	
		overall/comprehensive data	
		solution)	
9410 All-In-One Mobile		\$457.98 9410 All-In-One Mobile BGAN	
BGAN Terminal		Terminal - 9410 Extended	
		Warranty, add'l 18 months	
9410 All-In-One Mobile		\$916.98 9410 All-In-One Mobile BGAN	
BGAN Terminal		Terminal - 9410 Extended	
		Warranty, add'l 42 months	
9410 Terminal		\$54.06 9410 Terminal Accessories -	
Accessories		Power & Data Cable, blunt wire,	
		5m (can only be sold as part of	
		an overall/comprehensive data	
		solution)	
9410 Terminal		\$77.52 9410 Terminal Accessories -	
Accessories		Power & Data Cable, blunt wire,	
		10m (can only be sold as part of	
		an overall/comprehensive data	
		solution)	

	1	
9410 Terminal	\$89.76 9410 Terminal Accessories -	
Accessories	Power & Data Cable, cig lighter	
	plus & RJ45 cosket, 5m (can	
	only be sold as part of an	
	overall/comprehensive data	
	solution)	
9410 Terminal	\$8.16 9410 Terminal Accessories -	
Accessories	RJ45 Wiring block (can only be	
Accessories		
	sold as part of an	
	overall/comprehensive data	
	solution)	
9410 Terminal	\$15.30 9410 Terminal Accessories -	
Accessories	Mating Power & Data	
	Connector, bare (can only be	
	sold as part of an	
	overall/comprehensive data	
	solution)	
9410 Terminal	\$1,428.00 9410 Terminal Accessories -	
Accessories	Power & Dara Cable, 100m (can	
	only be sold as part of an	
	overall/comprehensive data	
0440 T	solution)	
9410 Terminal	\$202.98 9410 Terminal Accessories -	
Accessories	9410 Antrenna Mag Mounts	
	(can only be sold as part of an	
	overall/comprehensive data	
9450-C11 Mobile BGAN	\$4,690.98 9450-C11 Mobile BGAN	
Terminal	Terminal - 9450TW-C11 Wi-Fi,	
	PoE Ethernet, RJ11 phone/fax	
	(can only be sold as part of an	
	overall/comprehensive data	
	solution)	
9450-C11 Mobile BGAN	\$4,792.98 9450-C11 Mobile BGAN	
Terminal	Terminal - 9450E-C11 PoE	
Terrimai		
	Ethernet, RJ11 phone/fax and	
	ISDN (can only be sold as part	
	of an overall/comprehensive	
	data solution)	
9450-C11 Mobile BGAN	\$4,180.98 9450-C11 Mobile BGAN	
Terminal	Terminal - 9450LW-C11 Lite w/	
	Wi-Fi & Ethernet (can only be	
	sold as part of an	
	overall/comprehensive data	
	solution)	
9450-C11 Mobile BGAN	\$4,078.98   9450-C11 Mobile BGAN	
Terminal	Terminal - 9450L-C11 Lite w/	
	Ethernet only (can only be sold	
	as part of an	
	overall/comprehensive data	
2.122.21.11	solution)	
9450-C11 Mobile BGAN	\$508.98 9450-C11 Mobile BGAN	
Terminal	Terminal - C11 Extended	
	Warranty, add'l 18 months	
9450-C11 Mobile BGAN	\$1,018.98 9450-C11 Mobile BGAN	
Terminal	Terminal - C11 Extended	
	Warranty, add'l 42 months	
0450 C40 Mahila DCAN		
9450-C10 Mobile BGAN	\$7,648.98 9450-C10 Mobile BGAN	
Terminal	Terminal - 9450TW-C10 Wi-Fi,	
	PoE Ethernet, RJ11 phone/fax	
	(can only be sold as part of an	
	overall/comprehensive data	
	solution)	

	_			
9450-C10 Mobile BGAN	T	\$7,750.98	9450-C10 Mobile BGAN	
Terminal			Terminal - 9450E-C10 PoE	
			Ethernet, RJ11 phone/fax and	
			ISDN (can only be sold as part	
			of an overall/comprehensive	
			data solution)	
9450-C10 Mobile BGAN		\$7,342.98	9450-C10 Mobile BGAN	
Terminal			Terminal - 9450LW-C10 Lite w/	
			Wi-Fi & Ethernet (can only be	
			sold as part of an	
			overall/comprehensive data	
			solution)	
9450-C10 Mobile BGAN		\$7,138.98	9450-C10 Mobile BGAN	
Terminal			Terminal - 9450L-C10 Lite w/	
			Ethernet only (can only be sold	
			as part of an	
			overall/comprehensive data	
			solution)	
9450-C10 Mobile BGAN		\$1,018.98	9450-C10 Mobile BGAN	
Terminal			Terminal - C10 Extended	
			Warranty, add'l 18 months	
OAFO CAO Mahila DOAN		\$0,000.00	9450-C10 Mobile BGAN	
9450-C10 Mobile BGAN		\$2,038.98	I	
Terminal		I	Terminal - C10 Extended	
	<u> </u>		Warranty, add'l 42 months	
9450 Terminal		\$51.00	9450 Terminal Accessories -	
Accessories		4550	9450 Power cable, bare wire	
Accessories				
			(can only be sold as part of an	
			overall/comprehensive data	
			solution)	
9450 Terminal		\$51.00	9450 Terminal Accessories -	
Accessories		,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	9450 Cigarette plug power cord	
Accessories				
			(can only be sold as part of an	
			overall/comprehensive data	
			solution)	
9450 Terminal		\$25.50	9450 Terminal Accessories - Wi-	
Accessories			Fi Antenna (can only be sold as	
7.0000001100			part of an	
			l'	
			overall/comprehensive data	
			solution)	
9450 Terminal		\$124.44	9450 Terminal Accessories -	
Accessories			9450 RF Cable, Coax 10m, TNC	
			TNC (can only be sold as part	
			of an overall/comprehensive	
			data solution)	
9450 Terminal		\$5,098.98	9450 Terminal Accessories -	
Accessories			C10 Antenna with mag mounts	
			(can only be sold as part of an	
			overall/comprehensive data	
			solution)	
9450 Terminal		\$223.38	9450 Terminal Accessories -	
Accessories			C10 Antenna mag mounts (can	
			only be sold as part of an	
l				
l			overall/comprehensive data	
			solution)	
9450 Terminal		\$2,344.98	9450 Terminal Accessories -	
Accessories			C11 Antenna (mag mounts not	
			included) (can only be sold as	
			part of an	
			l'	
			overall/comprehensive data	
			solution)	
9450 Terminal		\$202.98	solution)   9450 Terminal Accessories -	
		\$202.98	9450 Terminal Accessories -	
9450 Terminal Accessories		\$202.98	9450 Terminal Accessories - C11 Antenna Mag Mounts (can	
		\$202.98	9450 Terminal Accessories - C11 Antenna Mag Mounts (can only be sold as part of an	
		\$202.98	9450 Terminal Accessories - C11 Antenna Mag Mounts (can	

9502 External Antenna (Two-Piece) M2M BGAN Terminal	\$1,018.98	9502 External Antenna (Two- Piece) M2M BGAN Terminal - 9502 External Antenna (Two-
Terrillia		piece) Terminal (can only be sold as part of an overall/comprehensive data solution)
9502 External Antenna (Two-Piece) M2M BGAN Terminal	\$1,222.98	9502 External Antenna (Two-Piece) M2M BGAN Terminal - 9502 External Antenna (Two-piece) Terminal C1/D2 Compliant (can only be sold as part of an overall/comprehensive data solution)
9502 External Antenna (Two-Piece) M2M BGAN Terminal	\$49.98	9502 External Antenna (Two- Piece) M2M BGAN Terminal - Extended Warranty, add'l 6 months
9502 External Antenna (Two-Piece) M2M BGAN Terminal	\$202.98	9502 External Antenna (Two- Piece) M2M BGAN Terminal - Extended Warranty, add'l 24 months
9502 External Antenna (Two-Piece) M2M BGAN Terminal	\$304.98	9502 External Antenna (Two- Piece) M2M BGAN Terminal - Extended Warranty, add'l 48 months
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$1,528.98	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - 9502 Integrated Antenna (One- piece) Terminal (can only be sold as part of an overall/comprehensive data solution)
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$1,579.98	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - 9502 Integrated Antenna (One- piece) Terminal C1/D2 Compliant (can only be sold as part of an overall/comprehensive data solution)
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$131.58	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - Spare Parts Kit, 9502 One-piece (can only be sold as part of an overall/comprehensive data solution)
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$49.98	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - Extended Warranty, add'l 6 months
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$202.98	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - Extended Warranty, add'l 24 months
9502 Integrated Antenna (One-Piece) M2M BGAN Terminal	\$304.98	9502 Integrated Antenna (One- Piece) M2M BGAN Terminal - Extended Warranty, add'l 48 months
9502 Accessories	\$916.98	9502 Accessories - Indoor Unit (IDU) (can only be sold as part of an overall/comprehensive data solution)

9502 Accessories	\$1,222.98	9502 Accessories - Indoor Unit
		(IDU) C1/D2 (can only be sold
		as part of an
		overall/comprehensive data
9502 Accessories	\$365.16	solution) 9502 Accessories - 9502
9502 Accessories	\$303.10	External Antenna Assembly
		(Antenna & RF cable) (can only
		be sold as part of an
		overall/comprehensive data
		solution)
9502 Accessories	\$12.24	9502 Accessories - IDU Strap
		(can only be sold as part of an
		overall/comprehensive data
		solution)
9502 Accessories	\$86.70	9502 Accessories - Basic fixed
		mount kit (can only be sold as
		part of an
		overall/comprehensive data
		solution)
9502 Accessories	\$35.70	9502 Accessories - Azimuth
		Elevation Bracket 1.6"/4cm pole
		(can only be sold as part of an
		overall/comprehensive data
9502 Accessories	\$35.70	solution) 9502 Accessories - Azimuth
9502 Accessories	\$33.70	Elevation bracket 2"/5cm pole
		(can only be sold as part of an
		overall/comprehensive data
		solution)
Low Profile BGAN	\$3,058.98	Low Profile BGAN Terminal -
Terminal	73,333.53	Low Profile BGAN Terminal Kit
		(can only be sold as part of an
		overall/comprehensive data
		solution)
Low Profile BGAN Terminal	\$202.98	Low Profile BGAN Terminal -
		Extended Warranty, Additional 6
		Months
Low Profile BGAN	\$610.98	Low Profile BGAN Terminal -
Terminal		Extended Warranty, Additional
Law Braffla BOAN	<b>***</b>	24 Months
Low Profile BGAN	\$814.98	Low Profile BGAN Terminal -
Terminal		Extended Warranty, Additional
LPB Accessories	\$1,732.98	48 Months  LPB Accessories - Antenna,
LI D'ACCESSOIIES	ψ1,732.90	LPB (can only be sold as part
		of an overall/comprehensive
		data solution)
LPB Accessories	\$71.40	LPB Accessories - LPB RF
El B / (cocssones		Cable Straight TNC (can only
		be sold as part of an
		overall/comprehensive data
		solution)
LPB Accessories	\$71.40	LPB Accessories - LPB RF
		Cable Right Angle TNC (can
		only be sold as part of an
		overall/comprehensive data
	A/=0 ==	solution)
LPB Accessories	\$178.50	LPB Accessories - LPB 6 Pin
		Ethernet Cable (can only be
		sold as part of an
		overall/comprehensive data
LPB Accessories	\$153.00	solution)  LPB Accessories - LPB 3 Pin
LI D ACCESSURES	\$100.00	Power Cable (can only be sold
		as part of an
		overall/comprehensive data
		solution)
		00.00.011)

9201 Portable Class 1 BGAN Terminal			\$1,018.98	9201 Portable Class 1 BGAN Terminal - 9201 Terminal - DISCONTINUED, no battery (can only be sold as part of an overall/comprehensive data solution)	
9201 Portable Class 1 BGAN Terminal			\$916.98	9201 Portable Class 1 BGAN Terminal - Ext. Antenna (includes 10M cable) (can only be sold as part of an overall/comprehensive data solution)	
9201 Portable Class 1 BGAN Terminal			\$331.50	9201 Portable Class 1 BGAN Terminal - 2-4 Wire ISDN Terminal Adapter (can only be sold as part of an overall/comprehensive data solution)	
WIRELESS BROAD	BAND SER MRC		NRC	NRC Description	Term
Plan Name/Title		MRC Description		·	
4G/LTE 1GB (Pooled Plan) with NBD Maintennace	\$26.52	4G/LTE 1GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 2GB (Pooled Plan)with NBD Maintennace	\$34.68	4G/LTE 2GB (Pooled Plan)with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 3GB (Pooled Plan) with NBD Maintennace	\$42.84	4G/LTE 3GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 4GB (Pooled Plan) with NBD Maintennace	\$51.00	4G/LTE 4GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 5GB (Pooled Plan) with NBD Maintennace	\$59.16	4G/LTE 5GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 10GB (Pooled Plan) with NBD Maintennace	\$99.96	4G/LTE 10GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 15GB (Pooled Plan) with NBD Maintennace	\$140.76	4G/LTE 15GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 20GB (Pooled Plan) with NBD Maintennace	\$181.56	AG/LTE 20GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months
4G/LTE 25GB (Pooled Plan) with NBD Maintennace	\$222.36	4G/LTE 25GB (Pooled Plan) with NBD Maintennace	\$796.62	Fixed cellular modem with built- in antenna, Standard Enterprise installation	12 months